

Loyola University Chicago

Dissertations

**Theses and Dissertations** 

2016

# Neural Correlates of Political Attitudes: Emotion and Ideology in the Brain

Dane Gorman Wendell Loyola University Chicago

Follow this and additional works at: https://ecommons.luc.edu/luc\_diss

Part of the Political Science Commons

# **Recommended Citation**

Wendell, Dane Gorman, "Neural Correlates of Political Attitudes: Emotion and Ideology in the Brain" (2016). *Dissertations*. 2299. https://ecommons.luc.edu/luc\_diss/2299

This Dissertation is brought to you for free and open access by the Theses and Dissertations at Loyola eCommons. It has been accepted for inclusion in Dissertations by an authorized administrator of Loyola eCommons. For more information, please contact ecommons@luc.edu.



This work is licensed under a Creative Commons Attribution-Noncommercial-No Derivative Works 3.0 License. Copyright © 2016 Dane Gorman Wendell

LOYOLA UNIVERSITY CHICAGO

**NEURAL CORRELATES OF POLITICAL ATTITUDES:** 

EMOTION AND IDEOLOGY IN THE BRAIN

A DISSERTATION SUBMITTED TO

THE FACULTY OF THE GRADUATE SCHOOL

IN CANDIDACY FOR THE DEGREE OF

**DOCTOR OF PHILOSOPHY** 

**PROGRAM IN POLITICAL SCIENCE** 

BY

DANE GORMAN WENDELL

CHICAGO, ILLINOIS

AUGUST 2016

Copyright by Dane Gorman Wendell, 2016 All rights reserved.

# TABLE OF CONTENTS

LIST OF TABLES	iv
LIST OF FIGURES	v
ABSTRACT	vi
CHAPTER ONE: INTRODUCTION	1
CHAPTER TWO: LITERATURE REVIEW	11
CHAPTER THREE: THEORY	50
CHAPTER FOUR: SURVEY DATA AND DISPOSITION ANALYSIS	66
CHAPTER FIVE: HEMISPHERIC ASYMMETRY ANALYSIS	101
CHAPTER SIX: GO/NO-GO BEHAVIORAL INHIBITION ANALYSIS	133
CHAPTER SEVEN: DISCUSSION AND CONCLUSION	178
APPENDIX A: SURVEY QUESTIONNAIRE	197
WORKS CITED	206
VITA	240

# LIST OF TABLES

Table 1. Descriptive Statistics of Survey Measures.	77
Table 2. Correlation Matrix of Important Variables.	82
Table 3. Cronbach Alpha Scores of Scale Measures.	86
Table 4. Disposition Hypotheses.	116
Table 5. Frontal Asymmetry Score Interpretation.	123
Table 6. PANAS-X Scale Compositions.	124
Table 7. Descriptive Statistics of Lab Participants.	126
Table 8. Correlation Matrix of Emotional Measures.	128
Table 9. Go/No-Go and ACC Hypotheses.	164
Table 10. Correlation Matrix of Go/No-Go and ERP Component Results.	170
Table 11. Summary of Go/No-Go Studies.	177

# LIST OF FIGURES

Figure 1: Survey Data Compared to National Sample	78
Figure 2. Distribution of Political Ideology	80
Figure 3. Distribution of BIS/BAS Scales	81
Figure 4. Correlation Matrix Visualization	83
Figure 5. Distribution of Individual Wilson-Patterson Items	88
Figure 6. Factor Analysis Scatterplot	93
Figure 7. Factor Analysis Barplot	95
Figure 8. Behavioral Inhibition and Two Measures of Ideology.	96
Figure 9. Behavioral Activation and Two Measures of Ideology	98
Figure 10. The 10-20 System and Hemispheric Asymmetry	121
Figure 11. The Anterior Cingulate Cortex	142
Figure 12. The Error-Related Negativity Component	155
Figure 13. Common ERP Components	156
Figure 14. An Individual Trial of Go/No-Go Task	166
Figure 15. Grand Average ERNs and Political Ideology.	172
Figure 16. Scatterplots of Ideology and ERN data.	173
Figure 17. Measures of Ideology in Three Go/No-Go Samples	191

#### ABSTRACT

Do conservatives and liberals have differing sensitivities to avoidance, inhibition, and negative emotion? Do psychological factors beneath our conscious awareness underlie the political ideologies we embrace? Political science researchers have broken new ground over the past ten years in our understanding of the psychology and physiology of political ideology. However, large questions remain about how political ideology may be related to avoidance motivations and negative emotion. This work expands our current knowledge in this area by presenting three studies with multiple methodologies: original survey data, electroencephalographic measurements, and behavioral experiments in a lab setting. Working in the tradition of J.A. Gray's dual systems of behavioral motivation, I explore how political ideology is related to several related dispositional measures of behavioral avoidance, behavioral inhibition, and negative affectivity. Overall, and in contrast to literature expectations, my evidence suggests that liberals and conservatives do not have persistent differences in avoidance sensitivity or negativity bias. While strong evidence remains demonstrating important dispositional differences between liberals and conservatives, additional research will be required before researchers can conclude that conservatives are uniquely motivated by psychological avoidance or negative affect.

vi

# **CHAPTER ONE**

# INTRODUCTION

According to the Oxford New Monitor Corpus in October 2014, insults levied at Democrats and Republicans have taken an especially explosive turn in recent times. A sample of 1200 negative phrases gathered from online sources reveals incredible negativity and no small amount of creativity (Martin 2014). The right accuses the left of being loons, twits, nitwits, trash, hippies, morons, hypocrites, fools, scum, and elitists. The left responds in kind against the conservatives with labels such as fanatics, obstructionists, extremists, misogynists, ideologues, zealots, nut-jobs, thugs, and crazies. The only thing they seem to agree about is that the other guys can best be described as "idiots."

Although the lists of commonly used insults includes terms such as "extremist" and "ideologue," accusing someone of being an ideologue is not the same insult that it once was. The most common ideological labels used in American politics—liberal and conservative—are typically used as descriptors, not insults. Political candidates in the modern era often run toward these labels, particularly Republicans towards the term "conservative." Democrats often prefer "progressive" instead of "liberal" but the two terms are indistinguishable in terms of meaning. These ideological labels function as a kind of summary statement of the beliefs held by an individual. The summary statements are readily used to bolster credentials and signal intentions. Politicians selfidentify as either conservative or liberal, using these labels to help their political brand.

But this is a modern innovation. As I will explain in Chapter Two, for two centuries, political ideology *itself* was an insult, a slur thrown at you by your political rivals. An accusation of ideology meant you had tremendous bias, close-mindedness, and living in false consciousness. Your political opponents used these charges to galvanize their supporters towards (or against) political revolution. Being called an "ideologue" was an incendiary accusation that you made a practice of systematically distorting facts, figures, and events to obfuscate the truth and manipulate people. Political ideology was thought to be a blinding worldview which made one unreliable and untrustworthy.

#### The modern understanding of political ideology

Beginning with Philip Converse's seminal paper (1964), political ideology came to take on a useful technical definition. Converse described political ideology as a structured collection of abstract values by which one understands the political and social environment. Political ideology crystalizes and communicates shared values in political outlook. In other words, political ideology is a lens by which to understand the world.

Converse famously introduced the idea of constraint to political ideology. If a person is truly ideological, holding viewpoints on some issues ought to constrain their viewpoints on others. For example, imagine your friend is pro-gun, pro-religion, pro-free

trade, anti-minimum wage, and pro-small government. Scanning that list of issues, you might surmise that your friend was motivated by several abstract principles, including favoring the free market, distrust of national government, and favoring freedom, individualism, and responsibility.

Given that belief structure, now take a moment to predict your friend's stance on nationalized healthcare. Given the abstract values described above, and the closeness of these views to American conservatism, you might guess they would be opposed to nationalized healthcare. Are you right? Probably, yes. While it is possible that your friend's viewpoints are not motivated by abstract principles—perhaps they have randomly acquired their views purely by luck or purely in mimicry of another person—it seems unlikely they would have acquired this interrelated portfolio without an underlying set of abstract political values to build on. This is political ideology. As I will discuss further in Chapter Two, while Converse found that few people outwardly recognized their beliefs as ideological, we have good reason to suspect that many people hold beliefs in recognizable ideological patterns without being able to articulate them.

How common is political ideology? Converse, writing in the 1960s, estimated that only 1 in 10 Americans were truly ideological, a very low number that seems to have surprised him and our discipline. To assess their political beliefs, Converse interviewed a sample of Americans and had independent judges assess whether abstract political values were directly expressed during the conversations. Michael Lewis-Beck and colleagues, updating Converse into the 2000s with a deliberately similar technique, note that this number had doubled to 1 in 5 Americans (Lewis-Beck et al., 2008). The remaining Americans held political beliefs not anchored to abstract values, but instead clustered around either group interests or the current state of the country. Or they simply had no issue content whatsoever to the political views they reported (for example, identifying with a political party but being unable to articulate even a vague understanding of what the party stood for).

Still, if only 20 percent of the American public is ideological, why are political scientists so interested in political ideology? As noted by social psychologist John Jost, a person need not be consciously aware of their ideology (Jost et al. 2006). They are not required to recognize or understand their own belief structure for it to be guiding their political behavior. Ideological principles may be latent or implicit to the individual, an unidentified engine under the hood of a car driven to work every day. The studies of Converse and Lewis-Beck put a high burden on participants. These participants must have sophisticated political awareness to be able to verbalize how their abstract conception of politics fits into representational political schema. Jost argues that a latent belief structure can guide attitude formation without the individual actually being aware of it, or without them being able to name it. One does not need to be able to describe or categorize the physics of internal combustion engines in order to drive the car. Regardless of your understanding of what is happening under the hood, your ride is still deeply affected by the engine. Consistent with this argument, as will be detailed in

4

Chapter Two, researchers today have found considerably more evidence that the mass public is ideological than what Converse was able to detect earlier.

The concept of ideology as an abstract belief structure has proven to be exceptionally useful for political scientists. According to one estimate, more than 50 percent of all papers published in the American Political Science Review (the top journal in our discipline) include the word ideology (Knight 2006, 620). One central concern in the political psychology subfield is the question of where do beliefs come from, and many literatures within the political behavior subfield have found use for Converse's definition of political ideology.

## Political ideology, emotion, and political psychology

Where does political ideology come from? Our best answer is that political ideology is shaped by culture, environmental forces, elites, personal relations, personality dispositions, and evolutionary origins. Our field is increasingly taking a wide viewpoint on the origins of political attitudes. On the cutting edge of these questions are the new political behaviorists who are integrating insights from many scientific fields to improve our understanding of political attitudes, fields including political science, social psychology, political psychology, and cognitive neuroscience.

For example, political scientists are increasingly noticing the connection between emotion and politics. George Marcus and colleagues proposed the Affective Intelligence theory, demonstrating how emotional states of enthusiasm and anxiety have a large impact on political learning and political information processing (Marcus et al. 2000). Ted Brader's study of campaign advertisements outlined how politicians can (and do) use emotionally infused advertisements to affect political attitudes and behavior (Brader 2005, 2006).

But nowhere has the study of ideology been more riveting than in the political psychology and biology of political ideology. Kevin Smith, John Hibbing, John Alford, and their colleagues and students report study after study of the physiological correlations to political attitudes (e.g. Alford et al. 2005; Alford and Hibbing 2008; Smith et al. 2011), drawing particular attention to the role of threat and negative affect in political conservatives (Oxley et al. 2008; Dodd et al. 2012). John Jost and colleagues have connected threat, fear and uncertainty to political conservatism, arguing that political ideology is motivated by the individual's cognitive, emotional, and social needs (Jost et al. 2003). This recent wave of heavy lifting on the subject of political ideology has been initiated by political scientists growing access to methodologies, training, and data from other disciplines.

Political scientists have long been interested in how political attitudes are formed, and now we are realizing the importance of understanding how emotion relates to political attitude formation. Psychologists and neuroscientists have waged pitched battles over the relation of cognition to emotion (Zajonc 1980; Lazarus 1982; Zajonc 1984; Leventhal and Scherer 1987; Gray 1990; Pessoa 2008), and these arguments have defined significant portions of psychology and cognitive neuroscience (these six papers listed immediately above have over 11,000 citations). We are still waiting for the dust

6

to settle on a conclusive understanding of emotion and cognition, but it has become increasingly clear that emotion has a highly integrated role in cognitive processing. And, so, political science—a "borrower" discipline from its very founding—will also need to consider more fully the importance of emotion for cognition.

#### Rising study of physiological politics in political science

The core project of the political psychology of ideology is to understand how differences in political ideology are related to the vast array of cognitive, emotional, and physiological factors that predict or explain political attitudes. Broadly, most of the work from researchers in this field operate from one of two different perspectives.

The first perspective was developed by political scientists John Hibbing, Kevin Smith, John Alford, and the University of Nebraska-Lincoln Political Physiology Lab. Their theory, which I call the Negativity Bias Theory, focuses on emotional sensitivity, specifically the theory that conservatives are dispositionally more sensitive to negative emotion. Their perspective on the psychological differences between liberals and conservatives has been shaped by dozens of studies that look at genetic and physiological differences in people. In 2014, Hibbing and colleagues published a paper called "Differences in Negativity Bias Underlie Variations in Political Ideology" which summarizes a great host of their work, and integrates the findings into a central theory.

The second perspective comes out of social psychology. The 'Motivated Social Cognition Theory' is based on the work of psychologist John Jost and many colleagues, and it argues that conservatism is motivated by a network of psychological factors related to fear, threat, and reducing uncertainty. According to Jost and colleagues, two primary factors divide liberals and conservatives: first, advocating versus resisting social change (as opposed to tradition) and second, rejecting versus accepting inequality (Jost et al. 2003, Jost 2006, Jost and Amodio 2012). In their important seminal paper (Jost et al. 2003), they conduct a meta-analysis on the psychology of political conservatism, and they discovered that these two factors are highly correlated with personality traits, dispositions, and psychological motivations related to fear, threat, and reducing uncertainty.

In our literature so far, these two perspectives have been considered largely complementary. The findings and theories have substantial overlap and there is much cross citation. In my view, however, the two theories are conceptually different. Hibbing and colleagues have focused on physiological differences between liberals and conservatives, and they tie their theory together through emotions. Psychologist John Jost's work focuses on differences in psychological processes and motivations, but does not home in on negative affect specifically. Jost's work has a strong connection to motivational avoidance. Both of these projects are tremendously important to work on political ideology by political psychologists. However, given the breadth of psychological factors and physiological dispositions that humans have, it is not surprising that a great deal of additional work must be done to deepen our understanding of the political psychology of ideology. Our theories can only mature by enlarging the scope of psychological factors that are being investigated.

# Purpose of Dissertation

This dissertation has one primary objective: to expand our knowledge about how negative emotional sensitivity and avoidance motivations are related to political ideology at the root level of the individual, operating before (or outside of) political stimuli. I propose one useful place to expand our understanding of negative affect and avoidance motivations is through the work of Jeffrey Allen Gray, a British psychologist and cognitive neuroscientist. Gray developed a dual model of behavioral motivation to explain how and why cognition and emotion lead to behavior. According to Gray, animals have a appetitive system of behavioral activation that engages in goal pursuit, while also having a aversive system of behavioral inhibition that monitors the environment for changing information and threats. As I will explain, Gray's theories combine cognition and emotion in a way that would benefit the theories of Hibbing's Negativity Bias Theory and Jost's Motivated Social Cognition Theory. By exploring the meaning of Gray's theories for political ideology, we will have an opportunity to sharpen several elements of the political psychology theories that have been proposed.

This dissertation presents data from three studies aimed to clarify the specific role of avoidance motivation and sensitivity to negative emotion in political ideology. These sets of questions have always been interdisciplinary in nature, and this dissertation is no different. I employ survey methodology alongside cognitive neuroscience methodology to approach these important questions from multiple perspectives. Along the way, I take the opportunity to outline how the theories of Hibbing and Jost have similarities and a few important differences in predictions.

This dissertation will proceed as follows. Chapter Two details the use of political ideology by political scientists and clarifies how the biology and politics research program contributes to these important questions. Chapter Three summarizes the core theories of the political psychology of ideology that are under investigation in this dissertation. Chapter Four describes and characterizes the participant pool for two behavioral experiments while also testing for dispositional correlations. Chapter Five presents the results of an electroencephalography study of frontal cortical asymmetry and its implications for political ideology. Chapter Six presents additional electroencephalography work, this time from a behavioral experiment called the Go/No-Go task. Chapter Seven concludes the dissertation with a discussion of implications.

10

# **CHAPTER TWO**

# LITERATURE REVIEW

Do psychological factors beneath our conscious awareness underlie the political ideologies we embrace? Political science researchers have broken new ground over the past ten years in our understanding of the psychology and physiology of political ideology. This chapter outlines a general research program for studying dispositional differences in liberals and conservatives. My approach is interdisciplinary, and I borrow from political science, social psychology, and cognitive neuroscience to better understand the study of political ideology.

A brief preview of the chapter's findings may be useful here. The reader should see the importance of political ideology for understanding political attitude formation and should better understand the complexity of measuring ideology in the mass public. The reader will also see that while political scientists have made serious advances in the past two decades to understand the cognitive and affective processes of political attitude formation, we still have many questions remaining. We will see that political psychology literature increasingly finds that political conservatism is correlated with negative affect and/or avoidant behavioral motivations, but these associations leave several important questions unexplored. Finally, the reader will also note the importance of multidisciplinary work for this research program. Psychology and cognitive neuroscience have considerably more advanced techniques and theories to investigate cognitive and affective differences between individuals. However, psychology often neglects to fully consider the multidimensionality of political ideology, which compromises the applicability of those findings for political science questions. Approaching our questions from multiple disciplines bolsters several weaknesses in analysis.

There are two major theories concerning the psychological differences between political liberals and conservatives. First, the Motivated Social Cognition Theory posits that differences in psychological factors (traits, dispositions, cognitive styles) suggest conservatives are more motivated by needs to reduce uncertainty, ambiguity, threat, and disgust than liberals (Jost et al. 2003). Later, I outline an interpretation of this theory as describing conservatives as being motivated by avoidance. Second, the Negativity Bias Theory argues that conservatives have more sensitivity to negative emotion.

These two theories do not directly compete; in fact, there appears much common ground between them, including considerable cross-citation. At their core, however, I believe they are studying different psychological mechanisms and potentially ought to have different predictions about the physiology of political ideology. I will argue below that avoidance sensitivity and negativity bias are different processes, and they need to be studied as such, with a kind of precision that is available to us from other disciplines. In this research program of the psychological and biological determinants of political ideology, it is time for researchers to be more precise with how specific psychological mechanisms may, or may not, be related to political beliefs.

This dissertation takes up that charge by employing Jeffrey Allen Gray's theories (1972, 1981) on behavioral motivation, which have direct connection to both avoidance sensitivity and negativity bias. J.A. Gray hypothesized that animals have two complementary systems that underlie cognition, affect, and behavior. The behavioral activation system (BAS) facilitates behavior towards the completion of an objective or goal. We use this system hundreds of times a day; from reaching for a cup of coffee, to opening a doorknob, to speaking with a friend. The behavioral inhibition system (BIS) halts ongoing behavior in response to the introduction of novel information. This is a surveillance system, helping us react to unexpected changes in the environment. For example, we may be in the process of drinking coffee when we see a dead fly floating at the top of the cup. Seeing the fly cues a disgust reaction, and we halt our planned action to drink a sip of coffee. Introducing Gray's theories to the study of political ideology will help specify and articulate the mechanisms underneath both of the established theories.

Establishing the place of this project within the larger literature of political science and political psychology requires several steps. First, we engage the current political science literature on political ideology, including the history, relevance, and structure of political ideology in the literature. Second, we move into the current state of the psychological and physiological findings on political ideology, including an

investigation into the two major theories in the literature—the Motivated Social Cognition theory and the Negativity Bias Theory. Third, we discuss the political implications of Gray's behavioral motivation systems, and how it intersects with the current political psychology theories of political ideology.

#### Political Ideology: History and Present

Political attitudes are ubiquitous, and they have created the sociopolitical world around us. Political communities are characterized by sharp disagreement on societal rules, values, and expectations. These differences in opinion occur within regional areas, within small communities, within social groups, within economic classes, and even within families. Larger, more heterogeneous populations have larger differences in opinion. Even in small political communities, where shared backgrounds and mutual experiences yield a fairly common way of life, political disagreements can be surprisingly divisive. Sometimes, the family unit is the most heated battleground! Perhaps this is due to the paradox of it all—a unit of people that have so much shared environment are expected to have a common politics. When they do not, it is puzzling.

Why do human beings have so many central disagreements over how a society should be structured? No doubt the incredible variety of human experience would yield an incredible variety of political opinions. However, interestingly, for all the diversity of human experience and political attitudes, political disagreements tend to actualize as a general debate between only two opposing camps. One side largely values traditions, existing institutions, communities and families over government, and promotes safety, economic freedoms, and economic inequality. This perspective has been labeled the "right-wing" perspective. The other side—the left-wing—largely values government as a corrective tool to balance unequal starts and protect disadvantaged groups, while promoting privacy at home and in religion.

In the United States and other places, the right-wing approach is called conservative and the left-wing approach liberal, and these perspectives emphatically disagree over the core values, policies, and politics that a society should embrace. There are some political issues that have no disagreement between liberals and conservatives; these valence issues can be important to partisan politics (Stokes 1963, Green 2007). But while we all agree that decreased crime and increased economic growth are good, it is puzzling and intriguing that we find ourselves mostly divided into two opposing camps, who frequently have orthogonal opinions on a variety of public policies. Partisanship, elite opinion leadership, and media all contribute to the polarization of our politics. But there is an important role for the individual self, which is our central concern here. Explaining why people have political disagreements requires an understanding of political ideologies, the term political scientists use to refer to belief structures of political attitudes.

#### Approaches to Studying Political Ideology in Political Science

Political scientists have a number of approaches they have used to examine the nature and origins of political ideology. Traditional researchers have focused on environmental factors as the chief agents for explaining political behavior. They have

made tremendously important contributions. Political behaviorists have sought the origins of political attitudes in socioeconomic classes, racial groups, ethnic groups, religious affiliations, social groups, work socialization, family socialization, cultural environment, elite influences, and particularized historical contexts. This work on political attitudes has been dominant in the contemporary study of political behavior.

More recently, political science has had an explosion of research into the psychological, physiological, and biological factors in the formation of political attitudes. These researchers have looked at correlations between political values and genes, physiology, and the brain. "Over the last three decades, we have witnessed the emergence of neurological, biological, endocrinological, and physiological paradigms for the study of human behavioral differences" (Hatemi and McDermott 2011, 1).

Political science has taken enormous strides in the past decades towards a better understanding of political behavior. From the sociological studies of Lazarsfeld and colleagues (1944), to the social psychological school of Campbell and colleagues (1960), to Downsian economic voting and rational choice (1957), and even to newer theories of political attitude formation such as Zaller (1990), political science has a long and distinguished literature for explaining how people act politically. Interdisciplinary insights have fueled each of these forward movements, and it is no accident. Political scientists have often benefited from borrowing heavily from the methodological techniques of other disciplines, and the four seminal works above are perfect examples of this trend. Lazarsfeld was writing from a sociologist's point of view and used a sociological framework to analyze political behavior; Campbell and his colleagues borrowed survey methodology and attitude theories from social psychology; Downs was a young economist applying market principles to voting behavior; and Zaller borrowed theories and understandings from cognitive and social psychology in attitude formation and applied it to questions about survey answers.

Outside of these trailblazing political behavior researchers, the discipline as a whole has frequently reached out and incorporated other fields of study. Political scientists reach out to philosophy, ethics, and history as quickly as it reaches out to economics, psychology, and—increasingly—biology. In the next section, we explore in detail how political ideology has been studied in political science in the past.

## **Origins and Definitions of Ideology**

Political ideology has long been a topic of interest to political scientists, and it holds a unique position in political science scholarship. Political science has been generally characterized as a "borrower" discipline (Rigney and Barnes 1980; Miller 1997; Amadae and Bueno de Mesquita 1999; Pieters and Baumgartner 2002; Bartels and Brady 2003; Knight 2006; Sigelman and Goldfarb 2008, 2012). However, ideology is one of the few constructs widely studied in contemporary political science to originate directly within political science itself (Knight 2006, 619). In her study of the history of ideology in political science literature, Knight (2006) reports that the term "ideology" has grown substantially over time, and it now occupies a central role in research. During the first decade of the American Political Science Review, only 2.6 percent of articles included "ideology" or a variant (620). By 1976, 50 percent of articles published in the APSR included the term ideology, and the percentage remained high into the 2000s (620).

Specific definitions of political ideology vary widely, but political science has worked with a few core definitions over the years. The original usage of ideology in politics was as slander against intellectual and political opponents in the 19th Century, used by figures such as Napoleon and Marx against opponents whom they saw as being dogmatically unreasonable and biased against their own views about government and political economy (620). Ideology at this time was synonymous with bias and falseconsciousness. This "blinding worldview" usage of ideology is characterized by the idea that an ideological person has a worldview permeating their ideas and that worldview insidiously causes them to be unable to see alternative or contrasting viewpoints. Early political scientists also used ideology in the "blinding worldview" sense in our journals (albeit with much less fiery rhetoric than a political speech by Napoleon or Marx). It is important to note, though, that prior to 1930, the use of "ideology" in political science literature is overall quite scarce (620). It is unsurprising that a term used by political revolutionaries to score political points and brand opponents was not used frequently by more restrained political scientists, but it also demonstrates that political scientists had not yet come to understand how ideology could be a tool for understanding belief structures in the mass public.

World War II increased attention to the study of ideologies, as communism, fascism, socialism, and Marxism became hotly studied topics in political science literature (Knight 2006, 621). Ideology during this period continued to be defined as a collection of interrelated beliefs or a worldview. As before, the "blinding worldview" was employed against opposing viewpoints in a slanderous way, boldly painting opponents as biased or close-minded. It is during this period, however, that ideology as "blinding worldview" began to give way to a new usage of "worldview system," dropping the insulting or accusatory tone. By "worldview system," I mean that political scientists began to recognize that structures of interconnected values and attitudes existed in states and individuals, and these worldview structures did not automatically suggest a bias. For example, our literature began to understand that the much-praised liberal democracy was itself an ideology. A political scientist writing in 1942, for example, approvingly called the Monroe Doctrine an "ideological fence against fascism" (Wilcox 1942). So, we have two different usages of ideology being used side-by-side in the World War 2 period: the "blinding worldview" and the "worldview system." I want to note in passing that both usages are still wreathed in the idea of conflict between world-views. Also, again, it is worth mentioning that "ideology" was still a relatively uncommon subject in political science literature through the 1930s and into the 1940s (Knight 2006, 619-620).

Ideology continued to evolve through the 1940s and beyond, gaining new relevance in the midst of all the "-isms" of the post-war era. The rise of "-isms" as a

powerful force in the global political environment gave rise to the use of ideology in political science literature. Knight notes that 70 percent of articles mentioning ideology between 1947-1956 also mentioned communism or Marxism (2006, 621). Ideology continued to be connected to "-isms" into the McCarthyism Cold War era and beyond, but we can notice that political science's positivist turn seemed to have contributed to the abandonment of the "blinding worldview" usage in favor of the "worldview system" usage (the negative connotation of "ideology" remained, however, and remains to this day—but instead of a blatant accusation, it had been reduced to a connotation at worst). For one example of the "worldview system" usage gaining prominence, we can observe that Samuel Huntington, writing in 1957, defined ideology as "a system of ideas concerned with the distribution of political and social values and acquiesced in by a significant social group"—a clean, clinical definition of what was once used in uproarious, accusatory political speeches a hundred years prior. Knight notes this transition: "although the coherent and relatively stable set of beliefs or values has remained constant in political science over time, the connotations associated with the concept have undergone transformation" (625).

Even in this later era, however, the new and more neutral usage of ideology as "worldview system" still had a broad, complex, and sometimes contradictory usage within the discipline (Gerring 1997). There was confusion as to the proper level of study for ideology; it could be variably applied to states, governments, institutions, groups, or the individual (Minar 1961). Minar, writing in 1961, summarized the then-current usage of ideology as "particularly vulnerable to vagueness and overgeneralization," but nonetheless noted with admiration that "the complete study of ideology is perhaps necessarily eclectic as to level and technique, as the best in the traditions of the discipline of political science has tended to be" (331). Interestingly, Minar's review of ideology appears to have predicted Converse's later work on ideology: Minar illustrates the importance of using psychological methodology to study ideology (329) and suggests that the difficulty of understanding ideology "perhaps be avoided when hypotheses are developed at a 'finer' level, i.e. when analysis proceeds on an individual psychological basis" (329). Minar directly influences Converse's groundbreaking study (and, highlights once again the "borrower" nature of political science).

It was in the 1960s that political science began to more narrowly and consistently define ideology as a value-neutral theoretical construct. The discipline's contemporary use and conception of political ideology begins with Phillip Converse (2006[1964]). Converse himself initially avoided using the term "ideology" because it was "thoroughly muddled," preferring the term "belief structures" in order to narrow down one particular facet of ideology he believed especially worthwhile for study (3). In essence, he was trying to separate out two terminological uses of ideology in order to focus on a third use, ideology as a belief structure. In other words, he was culling the blinding worldview and worldview system usage. Knight remarks "it is important to see Converse's contribution in the context of the times. By proposing that ideology was reflected in the representation of preferences on a liberal-conservative continuum, the spatial definition removed the negative implications of bias or false consciousness" (2006, 623). "Belief structure" continues to be used in political science to this day, but as Converse explains, the "obvious overlap" between belief structures and ideology are apparent and hard to separate functionally (3). In contemporary research, when political scientists speak of ideology they are nearly always speaking of a kind of belief structure, and consequently, terminology has converged (see Knight 2006). It converged for Converse, too; even as he was trying to avoid the muddled usage of ideology, he still used the term interchangeably with belief structure (2006[1964], 3). In the very same article that Converse shies away from using the term ideology, he nonetheless finds himself using it again. He writes that "belief systems that have relatively wide ranges...are broadly called ideologies and we shall use the term for aesthetic relief where it seems most appropriate" (2006[1964], 5).

Seemingly following Minar's suggestions, Converse proposed what Gerring (1997) and Knight (2006) have called the core definition of ideology: ideology as coherence and issue constraint. The field followed Converse on this choice. Gerring writes, "if all the senses of the term ideology are attended to, I would argue that only one trait meets [the criterion of universality]. The importance of coherence—aka 'consistency' or 'constraint'—is virtually unchallenged in the social science literature" (1997, 980). Knight writes in agreement, "this consensus is reflected in the 'core definition' of coherence, regardless of what other conceptual baggage might be imposed on the concept, the notion of ideology remains a relatively stable set of interrelated ideas" (2006, 623). Knight also mentions the connection between ideology and conflict: "beyond this, the crucial element of contrast—the notion that one set of beliefs competes against another—is implicit in discussion of party ideologies, 'ists,' and 'isms'" (623).

For Converse, the central defining element of political ideology is the idea of coherence, which is operationalized as issue constraint. For the collection of attitudes to have an underlying structure (a "belief system" or an "ideology"), outsiders who are given one attitude as a starting point should be able to predict additional attitudes held by the ideological individual. Holding one attitude should constrain what other attitudes that person may hold, if they have an ideological structure to their political attitudes. The collection of attitudes that a person could hold would be interrelated along ideological lines. In contrast, holding a random collection of attitudes without any consistency or restraint between them is non-ideological—there is no structure to the belief system. Knight (2006) identifies this as the core definition of ideology (despite widely varying usage from the 1800s until today) and she also offers that this definition is the most popularly used conception in political science since the 1960s due to Converse's influence on the issue. Converse's work on defining ideology has been tremendously influential in political science, but also in other social sciences (Kinder 2006, Bennet 2006, Jost 2006).

From Converse, political scientists inherited a definition of ideology that they (and other social scientists) could use to help understand belief structures in the mass

public. As noted above, over 50 percent of articles in the flagship journal of political science make a mention of the word ideology from the 1970s into the 2000s (Knight 2006, 620), demonstrating the importance and utility of the Conversian understanding of ideology for political scientists both past and present, with no signs of slowing down in the future.

#### Structure of Ideology: Left and Right

Having outlined the origins of political ideology in political science, we move now to an examination of the structure of ideology today. The divide between the political right and the left is a surprisingly timeless paradigm for politics. From the Spartans and Athenians in ancient Greece to the French Revolution and beyond, politically interested human beings in the past and present have found themselves divided on questions of tradition versus change, security versus freedom, and hierarchy versus individualism. Human beings will continue to have these conflicts of vision no matter which planet we find ourselves inhabiting.

One part of what makes the left-right division in politics so striking is its ubiquity. The left-right construct is not only useful to social scientists for describing political behavior—it is also a heuristic that is popularly used by the public. Political scientists have often thought of ideology as a continuum, and we have used spatial ideological modeling to great effect in describing political agents like legislatures (Poole and Rosenthal 2001, Poole 2005), presidents (Treier 2010) and courts (Martin and Quinn 2002). And, even in popular media, one can hear frequent use of spatial concepts for organizing political

groups. Presidential candidates appeal to the "wings" of their party (the more radical party members figuratively spread out from the center) during the primary election. Once selected by their party, they appeal to "the middle" towards moderate voters. In the 2012 presidential election, Republican Rick Santorum was tied to the "Far-Right" for his political beliefs on abortion, while Democrat Barack Obama was labeled by political opponents as a member of the "Far-Left" because of his universal health care plan. Republican Mitt Romney's primary campaign in 2012 was described as a way to "move to the right" to appeal to more radical conservatives who were skeptical of Romney's politically moderate tenure as governor of Massachusetts. Democratic-leaning independent Bernie Sanders was originally a "far-left" protest candidate of Democratic centrist Hillary Clinton's 2016 presidential campaign. Politically sophisticated American citizens and media use these terms in everyday discussions. As Jost (2006) writes, "even casual observers of today's headlines, newscasts, and late night talk shows cannot escape the feeling that ideology is everywhere" (Jost 2006, 652). A Google web search in May 2016 reveals over 185 million results for "liberal," 137 million results for "conservative," and 49.3 million results for "ideology."

What does it mean to be left or right in the United States? Ellis and Stimson provide excellent summaries of the popular conception of the two political ideologies in Ideology in America (2011). In the United States and in some other places, liberal is associated with leftist politics, while conservative is associated with rightist politics. To be liberal in the United States means you embrace the idea of equality of opportunity for all, and you see government as the primary tool for balancing out unequal starts (Ellis and Stimson 2011, 3). Liberals believe that a market economy provides benefits to society, but that it has dangerous outcomes to disadvantaged groups, thus requiring government to be a firm regulator (4). The role of government is also to establish standards for the social order, in markets and for public goods (4). Liberals embrace freedom for private, non-economic choices in lifestyle and religion, with religion considered to be outside the proper scope of government. To be conservative in the United States means that you believe that families and communities are the driving forces of society, not government (5). Government often lacks the moral imperative and practical feasibility to correct market failures and address other economic considerations (5). The free market may have issues, but government is ill-equipped to make adjustments, often making problems worse instead of correcting them (6). Conservatives support economic freedoms and income inequality, believing that both of these tend to lead towards a greater economic prosperity for society over time (5). Government should be limited, but they do have responsibility for providing safe and effective market transactions, as well as help enforce property rights and private contracts (5). In terms of private social behavior, conservatives tend to split into two camps. One group favors a strong government for promoting traditional values, enforcing order in society, and preserving the role of religion in public life (6). A different group is more libertarian in outlook, and they instead desire freedom in social affairs in parallel with freedoms in economic life (6).

Psychologist John Jost has studied political ideology from a social psychologist perspective, focusing on the psychological factors that may motivate conservative beliefs. Jost and colleagues have determined two primary factors divide liberals and conservatives most fundamentally: (a) advocating versus resisting social change (as opposed to tradition) and (b) rejecting versus accepting inequality (Jost et al. 2003, Jost 2006, Jost and Amodio 2012).

While other systems have made interesting contributions to the study of political belief structures and ideology (for example, the Moral Foundations Theory of Haidt and Graham 2007, Graham et al. 2009, Haidt et al. 2009 and the Cultural Cognition Theory of Gastil et al. 2011), none of them outright reject the basic liberal versus conservative pattern. Generally speaking, the liberal versus conservative dichotomy is preserved quite well in new systems, although most conceptualize liberal and conservative differences in a multidimensional framework instead of one-dimensional. For example, a common multidimensional model of political ideology divides social issues from economic issues. Even with two dimensions, one can see the liberal versus conservative conflict. Social liberals conflict with social conservatives alongside economic liberals and economic conservatives. We can increase specificity of ideology with new systems, but the left-right divide remains surprisingly intact. Overall, I believe the ubiquitous left-right divide occupies a central place in the study of political scientists and psychologists

27

to meet. However, we do need to take seriously the consequences of multiple dimensions of political ideology. We take up this conversation in the next section.

# **Dimensions of Political Ideology**

Typically, political scientists and researchers in related fields have argued for a simple one-dimensional scale of general liberalism versus conservatism. Researchers have articulated ideology as a one-dimensional construct (Knight 1999, 2006; Jost et al. 2009). This scale has served the research literature generally well over time, and it holds much predictive power (Knight 1999; 2006; Jost et al. 2009). The most popular scale of measuring political ideology is self-identification on a seven point Likert scale (see, for example, the time-honored American National Election Studies). In this scale, individuals are asked to place themselves on a seven point, fully labeled scale from "Very liberal" to "Very conservative," with the midpoint of four labeled "moderate." However, political scientists focused on the measurement of ideology in the mass public have often found ideology to be more complex than this simple scale, requiring two new understandings. First, ideological appears multidimensional, and the dimensions are most likely economic policy and social policy. Second, there appear to be differences in how ideology manifests in elites versus the mass public.

Political elites fit the one-dimensional scale well, given their strong attitudes and their well-sorted party identification (Kritzer 1978; Jennings 1992; Fiorina et al. 2005; Abramowitz and Saunders 2008; Lewis-Beck et al. 2008). Additionally, the Conversian definition of ideology as attitudinal constraint seems to work well for political elites (Kritzer 1978, Jennings 1992, Lewis-Beck et al. 2008). However, even for political elites, we have some cause to question what the scale is actually measuring (see, for example, Conover and Feldman 1981). It is plausible that the one-dimensional scale is descriptive of whatever current, ongoing elite ideological conflict is happening, rather than a stable grouping of values. If true, conservatism on the scale may not be defined by underlying or abstract conservative values, but instead by whatever self-described conservative elites have packaged together. The debate over same-sex marriage in 2013 within the Republican party is an example of elites trying to redefine what conservatism is, rather than trying to define their position against a static benchmark for what conservatism means (see Socarides 2013).

The validity of measuring the ideology of the mass public on a one-dimensional scale has generated a lengthy discussion in the political science literature, as numerous researchers both past and present have argued that the one-dimensional model is underspecified, simplistic, or simply empirically absent from the mass public (Stimson 1975, Conover and Feldman 1981; Kerlinger 1984; Peffley and Hurwitz 1985; Page and Shapiro 1992; Jacoby 1995; Kinder 1998; Stimson 2004; Jacoby 2009; Haidt et al. 2009; Treier and Hillygus 2009; Ellis and Stimson 2009, 2012; Popp and Rudolph 2011; Carmines et al. 2012, Hussey 2012). Measuring ideology in the mass public has been of tremendous interest in the political science literature. The first argument about the nature of belief systems in the mass public claims that the public is mostly non-
ideological (Converse 2006 [1964]); consequently, both one-dimensional and multidimensional conceptions are equally ill-suited to describing the public.

A secondary argument, however, posits that the mass public is multidimensional in its belief structures (Treier and Hillygus 2009; Hussey 2012; Carmines et al. 2012), and the mismatch between elites (who are one-dimensional) and the mass public (who are multidimensional) creates problems for measuring belief structures in the mass public. These studies note the special challenges of being a cross-pressured voter in a political environment where elites are one-dimensional. A voter who is liberal on one dimension (e.g. social issues) and conservative on the other dimension (e.g. economic issues) becomes underrepresented by elites, leading to lower political participation (Hussey 2012; Carmines et al. 2012) as well as confusion about their place on a one-dimensional ideology scale (Treier and Hillygus 2009; Hussey 2012).

Here is an example of that potential confusion. Libertarian voters (socially liberal but economically conservative) have trouble answering our ideological placement questions. When asked by the American National Election Studies to place themselves on a one-dimensional scale of liberalism-conservatism, libertarians are conflicted. Do they circle the middle reflecting the sum of their diverse attitudes? This is problematic for researcher interpretation because self-placement in the middle could also imply they are apathetic, moderate, or that they do not understand what the labels mean. But in this case, none of those possible explanations are true. These libertarian voters are both politically interested and fairly radical in their beliefs; they did not know how to place themselves because the scale simple does not fit their belief structure. There are other problems, too. Libertarians who routinely vote Republican may self-place themselves on the conservative side of the spectrum, misinterpreting that the surveyor is asking them about partisanship or perhaps identifying as conservative because the label is politically charged at the moment (think of the Tea Party movement).

The conclusion from this line of research suggests that a multidimensional scale will reduce measurement error. If the mass public is ideological at all, that is. We pick up this topic in the next section.

Why does the dimensionality of political ideology matter for the current research project? First, virtually all of the political psychology literature on the connection between ideology and various psychological factors measure ideology on a onedimensional scale. Political science literature suggests, however, that using a onedimensional scale may not fit the public very well, presenting the possibility that most work on this subject could be improved by introducing multidimensionality to their measures. Second, having difficulty capturing the mass public on a one-dimensional scale has traditionally been interpreted as evidence that the public is non-ideological, as most scholars conclude based on the Conversian tradition. This could be incorrect, however. If the public were actually multidimensional in their political ideology, trying to fit them on a single dimension would similarly fail. Moving forward, to study ideology properly requires social scientists to include multidimensionality.

#### Political Ideology in the Mass Public

Does the mass public even think about politics in ideological terms or understand what political ideology is? If political ideology is having a relatively stable and logically coherent belief system, few people meet that standard, according to Converse (2006 [1964]). Using American National Election Study data from 1956 to evaluate the public's view, Converse estimates that only about 12 percent of the electorate uses ideological thinking to organize their political beliefs, based on the attitudinal constraint definition of ideology.

Converse finds that only a small fraction of the mass public organizes their political beliefs according to broad and abstract principles. So, then, how do they organize their beliefs, if at all? Converse grouped survey responses into five categories of attitude structures: ideologue, near-ideologue, group interests, nature of the times, and no issue content. Ideologues are people who exhibit some evidence of ideological thinking, by articulating abstract principles and using conceptual dimensions to evaluate political actors. Near ideologues are those who used ideological thinking in a brief or incomplete way, yet did make at least a minimal appeal to abstract principles. Group interests were people who did not use ideological precepts and instead made evaluations based on the favorable or unfavorable treatment of groups in society (working against "the common man," favoring "big business"). Nature of the times describes people who praised or blamed political actors based on the actors' association with either good times or bad times in the recent past. Finally, there is a residual category for respondents who had virtually no structure to their political attitudes and were unable to articulate an organization for their beliefs at all.

This data is from one study in the 1960s and no matter how incredibly influential this piece is, we ought to look at more data. Lewis-Beck et al. (2008) closely followed Converse's methodology on attitude structures and updated observations about the mass public using 2000 and 2004 American National Election Studies data. The number of people who are ideologues and near-ideologues had increased from 11.5 percent to 19.8 percent of the population since Converse published his research in 1964. 20 percent of the population is twice what Converse found, although it still remains a minority of the mass public.

Political scientists disagree over these claims, however. Political scientists have tended to agree that political elites are polarized around ideology and partisanship. Political elites are broadly defined as people with high levels of political knowledge, political interest, and political participation (sometimes this group is also called political sophisticates or the political class). High levels of political engagement and political sophistication characterize them, and they also have strong, durable political beliefs (Abramowitz 2010; Claassen and Highton 2009; Levendusky 2009). Political elites can include elected government officials, but it can also refer to private citizens with an intense interest in political affairs. Political scientists are in agreement that political elites are polarized along the Left-Right continuum, and they have grown more ideologically polarized over time. The theme of elite polarization on a liberalconservative continuum is a major one in current political behavior studies (Hetherington 2001; Fiorina et al. 2005; Lewis-Beck et al. 2008; Abramowitz and Saunders 2008; Fiorina and Abrams 2008).

While political scientists agree about elites, they have an ongoing (and passionate) debate about political polarization in the mass public. A popular and influential book by Morris Fiorina (Culture War?, 2005) argues that while elite polarization is dangerously increasing over time, everyday citizens themselves have virtually no ideological polarization. In Fiorina's analysis, he demonstrates relatively small differences in opinion between red states and blue states (49), as well as only mild differences in opinion between self-placed Republicans and Democrats on contemporary political issues (65). Claassen and Highton (2009) find support for Fiorina, and they suggest that party elite polarization has been increasing and that the public still continues to be unmotivated to become politically aware. RePass joins in this debate on the side of Fiorina, demonstrating in his analysis that only 7.4 percent of potential voters in the United States were found to be both strongly conservative and strongly Republican, while only 3.2 percent were both strongly liberal and strongly Democratic (2008). RePass also reports that 62.4 percent of the American public is functionally non-ideological, meaning they do not adhere to abstract ideological principles on either the left or the right (2008). This line of argument tends to lead towards Conversian conclusions about the mass public—regular people simply are not animated by ideological principles. Fiorina's work describes a mass public that is

distinctly moderate and characterized by rather strong agreement about the issues (instead of being locked in ideological conflict).

Abramowitz and Saunders (2008) disagree intensely with Fiorina, arguing that a substantially large segment of the mass public is ideological and growing more ideological over time. Abramowitz and Saunders find little evidence of polarization in the least politically engaged third of the public (agreeing with Fiorina here) but they find that most engaged third is significantly polarized (2008, 546). Moreover, they also argue that polarization actually has some benefits for increasing political interest and participation, by energizing voters and increasing political participation. This debate between the Fiorina camp and the Abramowitz camp continues to this day to be fairly spirited, inspiring new research (Fiorina et al. 2008; Abramowitz 2010). These scholars all agree that political elites are sharply polarized, and have grown more polarized over time. They disagree over the ideological polarization of the mass public.

Arguments over the polarization the public may give us suggestions about whether the public's attitude structures are ideological in nature, but it does not fully satisfy the basic Conversian project of exploring whether the mass public adheres to ideological belief systems because political polarization does not require political ideology but could be dependent mostly on partisan identity (or even, as recently suggested by Iyengar and colleagues, by emotional affect (Iyengar et al. 2012); see also Huddy et al. 2015). The most direct evidence we have, therefore, is from the studies of Converse and followers (Lewis-Beck et al. 2008; see also McGuire 1986; Tedin, 1987). In his original work, Converse demonstrated that few in the mass public thought about politics in terms of ideology, but there has been a strong growth of ideological thinkers in more recent time, according to Lewis-Beck and colleagues (who have deliberately replicated Converse's methodology as closely as possible). Overall, it seems that political scientists mostly agree that the number of people in the mass public who *fully* structure their political attitudes according to ideological principles is small but perhaps growing. Political science is not the only social science to look at political ideology, however, and we will see that political psychologists and social psychologists have interesting insights that political scientists have overlooked.

In a 2006 retrospective on his work, Converse eagerly commented on the view born from his work that citizens mostly have nonattitudes and/or are non-ideological (Converse 2006). Converse quickly addresses the issue as a strawman argument of the worst kind. He states, "to my eye, the worst common misinterpretation of the essay attributes to me the claim that most citizens have only "nonattitudes" on questions of public policy. It is my contention, I am told, that "real" policy opinions are in very short supply" (300). In his own words, he states a viewpoint that the public has many ideological beliefs, with only 22 percent of his original sample have completely nonpolitical viewpoints (301). He says, "In short, I have found it very hard to understand this misreading about few in the electorate holding any "real" policy attitudes. It is almost as though the misreader believes our argument to be that citizens must be either full ideologues or near to it (12 percent of the electorate) or else that they have no "real" policy attitudes. This is quite unimaginable to me" (303). Converse himself believes that "full ideologue" is perhaps a very tall bar to leap for the mass public, and a failure to hit that threshold does not indicate that people are completely nonideological.

Indeed, is it possible that these scholars have simply "defined away" ideology by making its requirements too steep? (Jost 2006, 653). Converse had a tremendous impact on psychologists studying political ideology, effectively closing down the study of political ideology (651). "The deadening impact of these conclusions on the study of ideology in social, personality, and political psychology can scarcely be exaggerated" (Jost 2006, 651-652). Jost criticizes the entire group of ideology scholars for having a standard for ideological thinking that was too strict, with too many prerequisites (650-652).

Following Tedin 1987, Jost proposes the definition of political ideology is better thought of as an interrelated set of moral and political attitudes possessing cognitive, affective, and motivational components (Jost 2006, 653). This standard, more flexible and broader than Converse's, "gives ordinary citizens a reasonable chance of empirically satisfying the criteria of being ideological" (654) and allows Jost to issue a call for psychologists and political scientists to reopen the study of political ideology. Jost points us to Kerlinger (1984) who writes as if to summarize this point:

Whether conservatism and liberalism are typically conceptual tools for the man-in-the-street is not the central point. For the scientist, too, liberalism and conservatism are abstractions like any other abstract concepts he works with: introversion, intelligence, radicalism, achievement, political development and the like. To be sure, most people don't recognize their abstract nature and certainly don't use them as social scientists do. Nevertheless, they are quite familiar with their behavioral and environmental manifestations. (217)

Political scientists have been primarily concerned with the question of whether elites and the public organize their politics around ideological principles and how consistent they are in the expression of those principles—thus emphasizing defining and measuring ideology instead of explaining why individuals, groups, or societies gravitate towards ideological persuasions (Jost 2006, 654). The insight of Jost and Kerlinger is that human beings do not have to intellectually understand ideological thinking in order to be found using it. When looking at the most influential work from political scientists on this question, Jost's charge seems fair—Converse and his followers have frequently defined ideological thinking in terms of whether citizens can explain what ideology is, whether they can accurately frame their opinions in ideological terms, and whether they can accurately place political actors on a political ideology continuum. These are important questions worthy of continued analysis. However, just because people do not understand political ideology does not mean they are not ideological. Moreover, as Sullivan and colleagues taught us about the mass public and ideology, people are highly sensitive to the method of measurement employed (Sullivan et al. 1978). Ask questions as Converse did, and the public largely looks non-ideological. However, change the questions as Nie and colleagues did (Nie and Anderson, 1974; Nie et al. 1976), and ideology in the mass public swells to much higher rates. In this respect, ideology is similar to religious belief—it is difficult to parse out, but there is little doubt that

religiosity varies widely among the population and can be a driving force in some people's lives.

Jost (2006) marshals some empirical support for the claim that the mass public is more ideological today than in the quieter 1950s-60s when Converse was collecting his data. First, note that Stimson (1975) argued that more than a majority of the actual electorate had evidence of an ideological belief structure according to Conversian definition (414). Second, Judd and Milburn (1980) analyzed data from the 1970s and found it to "pose a substantial threat to Converse's original hypothesis that the attitude responses of the public at large are unstable, nearly random responses" (82). In an analysis of ANES data conducted by Jost, he finds that over two-thirds of respondents since 1972, and over three-fourths since 1996 were willing to place themselves on a scale of liberalism-conservatism, even when provided explicit "don't know" answer item options. Overall, self-placement on the scales are reasonably accurate (Conover & Feldman, 1981; Evans, Heath, & Lalljee, 1996; Feldman, 2003; Knight, 1999). Additionally, the scale has tremendously strong predictive usefulness, explaining 85% of the variance in presidential vote choice (Jost 2006, 658).

Some political scientists have joined this chorus as well. In their book *Predisposed*, authors Hibbing, Smith, and Alford write: "Ideology is not, as Converse and his many followers claimed, merely the ability to describe currently popular labels or to endorse collections of positions that meet with the approval of political scientists" (2014, 56). Like Jost, they argue that ideological thinking is a core part of human nature: "Ideology is us" (56). Ideological thinking is due to bedrock social and political problems that draw roots from our evolutionary heritage. "The [political] division is real and unavoidable, and it centers on distinct orientations to mass-scale social life...ideology...is not, as the 'end of ideology' school asserted, a concept that just popped out fresh and new from Renaissance thought, only to fade from sight with the end of the Cold War" (55-56).

Ellis and Stimson (2012) similarly reject Conversian claims about ideology, writing that "whatever the well-documented weaknesses and randomness of individuallevel political opinions (see Converse 1964 for the still authoritative treatment), we expect to find power and simplicity when we look at the summary preferences of the whole electorate over time" (37). In their analysis, Ellis and Stimson find that large segments of the population that can be categorized into four ideological groups (operationally vs. symbolically, liberal vs. conservative). Ellis and Stimson easily concede that the voters may not be capable of articulating their own underlying belief structure, but the big picture of the data tells a clear story. For example, in their most recent year of data (2006) they find that 25% of the public is operationally liberal but symbolically conservative (Figure 5.5, 98). This "mismatched" ideological specification would have failed Converse's test, effectively labeling a quarter of the population non-ideological under his terms. But, in reality, this group has a consistent set of professed values (typically conservative) and also a consistent design for how they want government to function (typically liberal). It is not a contradiction or a confusion. They are churchgoing, gun-owning patriots who support minimum wage and believe it is the role of government to provide housing for the poor (Table 5.5, 105). They may be *conflicted* conservatives (Ellis and Stimson 2012, 111) but they are consistently so in their attitudes about social life and government prerogatives.

Jost, Hibbing, and other political psychologists propose that ideological thinking plays a critical role in social cognition—the way human beings encode, store, process, and retrieve social information. Studies from these researchers reveal meaningful political and psychological differences between liberals and conservatives that co-vary with dispositional and situational variables. Political ideology is tied more deeply into the human experience than we may have realized, and this area deserves more attention from political scientists.

Next, we turn to new research in political science and political psychology that explores the psychological and physiological correlations to political ideology. Two large theories have emerged to describe the underlying mechanical structures of a host of various studies on the psychological and physiological traits of political adherents. We outline these two theories and point to an area of the literature that needs better specification in order for us to understand how liberals and conservatives differ from each other in their traits and dispositions.

# New Approaches to the Study of Political Attitudes

The study of biological origins of politics could be considered to have started shortly after the discipline itself was founded (Aristotle's famous "man is a political animal"). In the modern state of the discipline, E.O. Wilson's writings on sociobiology (Wilson 1979) kickstarted an intense debate over biological determinism that appears to have resulted in a cooling of the entire research program. More recently, renewed interest from political scientists and psychologists have reinvigorated the field of study. They have faced less criticism from other academics compared to the earlier generation, perhaps partially because of increased scientific literacy across fields, or perhaps due to increasing tolerance for other methodologies, or perhaps partially because of a concerted effort by these new wave biopolitical scientists to express the unification of nature and nurture, instead of being ambivalent or depicting them as opposing forces. Very recently, scholars have begun to connect some of the dots between political science and biology by calling for a new look at the evolutionary origins of politics (Hatemi and McDermott 2011).

What is "biopolitics" as a subfield of political behavior? Somit and Peterson define biopolitics as a "short-hand term used to describe the approach of those in the profession who believe that biological concepts—especially evolutionary theory, which treats behaviour as the product of both nature and nurture—and biological research techniques can help us study and understand political behaviour better" (1999, 559). Alford and Hibbing write "a biological approach to political science is hardly new... in fact, biopolitics stretches back nearly as far as behavioral politics; both approaches found their first enthusiastic practitioners in the behavioral revolution of the 1960s" (2008, 184). The first review of the biopolitics literature was published in 1972, summarizing the findings and theoretical underpinnings of forty articles (Somit 1972). Biopolitics received formal disciplinary recognition in 1973 when the IPSA Research Committee on Biology and Politics was formed (560). The Association for Politics and Life Sciences was formed in 1981 and launched a dedicated journal, Politics and the Life Sciences, in 1983 (560). By 1998, Somit and Peterson note that the biopolitics literature had increased to over 1200 articles (1998, 560).

However, the state of biopolitics research in political science proper has always been questionable even as it was growing. Somit and Peterson note the vast majority of this literature was featured in journals outside of political science, and the top tier journals were largely abandoning the publishing of biopolitics work (1998, 561). Somit and Peterson (1999) express disappointment in the lack of biopolitics research in political science of that time, and they attribute this dearth of research to (a) losing shared theoretical space with rational choice, (b) residual distaste for ill-founded social Darwinism accusations and implications of racist associations with earlier biopolitics work, and (c) hesitation of new scholars to engage the work because of negative career implications. Alford and Hibbing note the departure in methodology between early behaviorists and early biopolitics [had] remained largely theoretical, descriptive, and speculative" (Alford and Hibbing 2008, 184). Even the negative associations with social Darwinism aside, biopolitics appears to have stagnated because political science grew increasingly interested in empirical study, and early biopolitics researchers did not share this interest, or had methodological difficulty.

This changed in the 2000s. A groundbreaking study by Alford et al. (2005) used twin data to reveal the surprising inheritability of political attitudes and party identification. Fowler and colleagues followed swiftly afterward with two studies demonstrating the heritability and genetics of political participation (Fowler et al. 2008; Fowler and Dawes 2008). These studies were published in top tier journals and were empirical in methodology. They opened a floodgate of empirical biopolitics work within the pages of political science's top journals (including, also, high-profile criticism; see Charney 2008; Charney and English 2012).

#### Why Biological Models?

Political behavior researchers have traditionally used two approaches for explaining human action in politics—situational and individual (Hatemi and McDermott, 2011, 13). Socialization models exemplify the situational approach, positing that the environment creates, shapes, and reinforces the political behavior of the individual that ascends from that environment. The individual approach, in contrast, is exemplified by rational-choice modeling, which asserts that individual behavior responds to changing incentive payoff schedules. As Hatemi and McDermott note, "in their most reduced form, behavioral models argue that all behavior results from social conditioning (Campbell et al. 1960), while rational-choice models assume preferences are exogenous, fixed, and given, and remain agnostic, if not unconcerned, about their source (Bueno de Mesquita 1983)" (Hatemi and McDermott, 2011, 13). Both of these perspectives are environmental in the sense that environmental changes drive all of the important effects in the models. Changing socioeconomic status or changing the payoffs for defecting is proposed to substantially change the behavior of subjects.

Models from both approaches have had a huge impact on the discipline of political science as a whole, and they continue to be productive lines of research that increase understanding and knowledge of political behavior. Over time, however, many other scientific disciplines challenged the assumptions of both socialization/behavioral models and rational-choice models (Green and Shapiro 1994; Robson and Kaplan 2003). Hatemi and McDermott note that "numerous challenges from economics, psychology, neuroscience, and other fields have found that not all people are socialized to act the same way; rational-choice models also hold limited explanatory capacity by remaining almost exclusively focused on choices motivated by unrealistically narrow conceptions of self-interest (Dawes and Thaler 1988; Tversky and Thaler 1990; Fehr and Gächter 2000; Gintis 2000; Henrich et al. 2001, Fehr and Fischbacher 2004)."

These environmental models necessarily are silent on the human being prior to interaction with the environment. This is not a fundamental flaw to the research program, but we no longer need to treat the pre-environment psychological/biological being as a black box to be ignored. We already intuitively understand that humans do not enter into the environment as blank slates. People within a common environment (say, a small community) can have rather divergent opinions on politics, or could have

45

very different preferences for monetary inducements. Race, gender, social class, and culture have powerful effects on political behavior, but within these subgroups is disagreement about political candidates and political issues. Even within families there is a surprising diversity of opinion—two siblings raised in the same household have an environment in common and yet regularly hold different political beliefs. The point here is that the environmental effects are not universal—every human reacts differently to every environmental effect. We are beginning to fashion the tools we need to start understanding why and how people are different prior to the important environmental effects.

Given our evolutionary history, purely economic models of human behavior seem dubious as explanations for the sum of human political behavior. "The process of natural selection is based upon adaptive traits beginning at a much earlier period in human development, where pure economic power-seeking and self-interest were not the only potential adaptive traits, if such traits were adaptive at all. Certainly many important human social and political traits, including detecting kin, selecting mates, foraging for food, avoiding predators, and detecting cheaters, evolved in a context prior to modern market conditions" (Hatemi and McDermott 2011, 13). We should not be surprised that it takes an exceptionally large monetary incentive to get human beings to defect from those who share close kinship, given how central such an impulse seems to be to our evolutionary psychology (Trivers 1971; Kruger 2003; Axelrod and Hamilton 2006). Economic models also seem to be unable to explain our tendency to want to punish defectors even when it costs us personally to do so, and we expect to not recover the costs of punishment (Gintis 2003).

Biological models have the potential to contribute to the existing sociological and economic behavioral models in political science by offering a theoretical explanation for these behavioral effects that the other models consider paradoxes (Hatemi and McDermott 2011). But even more centrally, evolutionary political science offers to investigate a large paradox in the study of political behavior: "simply put, our current theories and examinations largely assume that all people are biologically the same when it comes to politics, which is a radical notion considering how remarkably diverse humans are in virtually every other domain" (Hatemi and McDermott 2011, 18).

With these insights in mind, calls for an ambitious program of political neuroscience began in the 2000s from a number of researchers connected closely to biopolitics in political science (Marcus et al. 1998, McDermott 2004, Schreiber 2006, Fowler and Schreiber 2008, McDermott 2009; Hatemi and McDermott 2011; Schreiber 2011) as well as in political psychology (Lieberman and Schreiber 2003, Cacioppo and Visser 2003, Amodio and Jost 2007, Jost and Amodio 2012). As we shall see in the next section, political neuroscience offers new methodological tools to study our core questions about political ideology. For the present study of political ideology, biological investigations promise to be revealing—political ideology is likely an evolutionary adaptation that human beings evolved in response to the social nature of human groups (Thornhill and Fincher, 2007; Hatemi and McDermott, 2011; Fincher et al. 2008).

### Two Psychological Theories Emerge

The recent reawakening of the biopolitics and political attitudes research program in political science has been coupled with a similar revival of interest in political ideology in social psychology. While there is a myriad of studies in the literature tying various psychological, physiological, neurological, and endocrinological factors to political attitudes, two overarching theories have emerged for explaining why these differences are there. First, the Motivated Social Cognition theory posits that differences in psychological factors (traits, dispositions, cognitive styles) suggest conservatives (compared to liberals) are more motivated by needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al. 2003). Second, the Negativity Bias Theory argues that conservatives have more sensitivity to negativity than liberals, causing them to register greater physiological responses to such stimuli and devote more psychological resources to them. Considerable cross-citation exists between the two theories and much common ground. Below, I outline the two theories and their findings. We are on the precipice of additional theoretical development (see, for example, unpublished work by Tritt et al. on the arousal model of political conservatism). However, the two major theories represent the most well-developed and empirically supported literature that we have today.

Having established a history of research on political ideology, the next chapter outlines a research program designed to improve our understanding of the two major theories of political ideology from a political psychology perspective. It also introduces the work and ideas of Jeffrey Allen Gray, an influential psychologist who can help specify and clarify the psychological processes studied by the teams from Hibbing and Jost.

## **CHAPTER THREE**

## THEORY

This chapter builds the theoretical foundations for the empirical investigation in the remainder of the dissertation. The motivating question for this dissertation is as follows: do conservatives and liberals have differing sensitivities to avoidance, inhibition, and negative emotion? Do psychological factors beneath our conscious awareness underlie the political ideologies we embrace?

As reviewed in the last chapter, there are two major theories concerning the psychological differences between political liberals and conservatives. Psychologist John Jost and his colleagues formulated the Motivated Social Cognition Theory. This theory posits that conservatives are motivated by behavioral avoidance. In Jost's terminology, conservatives are more motivated by needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al. 2003), which is due to differences in dispositional traits (compared to liberals). The second theory, from John Hibbing and colleagues, posits that conservatives have more sensitivity to negative emotion. I term this the Negativity Bias Theory.

This chapter provides theoretical and literature justification for my characterization of these two theories. It is important to recognize that these two theories are not competing theories—I argue they are interested in different

psychological processes. Jost's work has an affinity with a psychological disposition called avoidance sensitivity, while Hibbing's work focuses instead on negative emotion. By the end of this chapter, we will see that Jeffrey Allen Gray's theories on behavioral motivation have a direct connection to both avoidance sensitivity and negativity bias. His theories, and the neuroscience related to them, can considerably aid the research program of understanding political ideology from a psychological perspective. Thus, the goal of this chapter is to articulate these two theories, integrate Gray's theoretical insights to this project, and explain why neuroscience methodology is essential to sorting all of this out.

### The Negativity Bias Theory

Hibbing, Smith, and Alford (2014) present the Negativity Bias Theory as a summation of a decade of research into the physiology of political ideology. These scholars have been the primary progenitors of the modern "rediscovery" of the biology and politics research program, a revival that begins in the mid 2000s. In summarizing the entire empirical biopolitics literature in political science, as well as drawing deeply from the political psychology literature, Hibbing and colleagues hypothesize that conservatives have a stronger negativity bias than liberals. Negativity bias means that "negative events are more salient, potent, dominant in combinations, and generally efficacious than positive events" (Rozin & Royzman 2001, 297).

Human beings, on average, tend to have negativity bias. People in general are more sensitive to negative faces, words, and social information, and biological systems tend to have measurably higher levels of activation in response to negative than positive stimuli (Rozin & Royzman 2001). However, Hibbing and colleagues note that this sensitivity is measurably stronger in conservatives than liberals and this theory explains nearly all of the hundreds of political psychological studies of correlates to political attitudes.

Consequently, not only do political positions favoring defense spending, roadblocks to immigration, and harsh treatment of criminals seem naturally to mesh with heightened response to threatening stimuli but those fostering conforming unity (school children reciting the pledge of allegiance), traditional lifestyles (opposition to gay marriage), enforced personal responsibility (opposition to welfare programs and government provided healthcare), longstanding sources of authority (Biblical inerrancy; literal, unchanging interpretations of the Constitution), and clarity and closure (abstinence-only sex education; signed pledges to never raise taxes; aversion to compromise) do as well. Heightened response to the general category of negative stimuli fits comfortably with a great many of the typical tenets of political conservatism (23-24).

The evidence for increased negativity bias in political conservatives is readily available through a group of studies focused on physiological correlates of political ideology. Oxley et al. (2008) present evidence that conservatism is correlated with physiological reactions to non-political stimuli. In a group of non-student participants with strong political beliefs, individuals less tolerant of sudden noises and

52

fearful/disgusting visual images, tended to support increased defense spending, capital punishment, patriotism, and the Iraq War. Helzer and Pizarro (2011) find that asking research participants to wash their hands increased their stated preferences for conservative positions. Several studies have found that threatening stimuli are consistently more distracting for conservatives (Carraro, Castelli & Macchiella 2011, McLean et al. 2013). Negative stimuli such as angry faces attract the attention of conservatives more than they do liberals. Dodd et al. (2012) found conservatives spent significantly more time looking at negative images and were significantly quicker to "fixate" on those images (as measured by eye-track software) as well.

## The Motivated Social Cognition Theory

Political science has a consensus definition of political ideology as attitudinal constraint, and the discipline makes use of this definition to explore and explain political behavior. As mentioned earlier, however, some political psychologists have used a different definition and conceptualization of ideology. The major departure in theory is that political scientists have put abstract principles as the foundations of political ideology while these political psychologists believe root psychological needs are the foundation for the abstract principles. So, political scientists would believe a person is a liberal ideologue because they endorse certain abstract principles about fairness and equality. Endorsement of abstract principles leads to endorsement of political attitudes in an ideological way. Psychologists working on political ideology have typically taken a different perspective. In contrast, they would believe a person is a liberal ideologue

because they have psychological needs or sensitivities that make the abstract principles of fairness and equality appealing to the individual. Psychological needs lead to endorsement of abstract principles, which lead to endorsement of political attitudes in an ideological way.

The psychological theory has the potential to contribute much to political science's understanding of political ideology. In my view, the constraint theory may describe a defensible way to measure the presence of ideology, but it simply does not speak to why. Human beings seem strangely compelled by ideological beliefs and powerfully animated by ideas. Some have been willing to kill or die for abstract principles, committing great acts of atrocity or heroism (Jost and Amodio 2012, 55). The strength of these endorsements suggest that ideology is tied deeply to the self. Even the more everyday forms of ideological thinking can fundamentally bend the way that we are exposed to information and bias our information processing.

Ideology can motivate cognitive processes and behavior in several theoretical ways (Jost et al. 2009). First, ideology reduces uncertainty by suggesting preferences and providing a framework for analyzing new and old contexts and situations. Second, ideologies offer existential security, a way to cope with anxiety about death (see Becker 1985, also Greenberg et al. 1986 on terror management theory). Third, people are drawn to ideologies for reasons of affiliation, for belongingness and social strength as part of a group. People take an ideological perspective in politics that they find appealing—political scientists will not find that controversial—but they do so in part because the ideology itself appeals to psychological dispositions and cognitive processes within the individual. Ideology offers "certainty, security, and solidarity" (Jost and Amodio 2012). Political scientists may find this controversial, or they may not, but it is certainly the case that this angle is understudied in mainstream political science literature.

One of Jost's central arguments, however, is that not all ideologies are the same, and not all ideologies will satisfy the same needs. There are likely important psychological differences between varying ideologies, although researchers have not yet ventured outside of studying conservatism contrasted with liberalism (but see lyer et al. 2012 for a look at libertarianism). Jost and his colleagues have focused on the left-right divide between political liberals and political conservatives (in contrast to, say, religious ideologies or social ideologies such as Social Dominance Orientation (Sidanius and Pratto 1999), but there are, no doubt, other ideological belief structures. In terms of political ideology, political psychologists have mostly followed political scientists and accepted the unidimensionality of political ideology.

The Motivated Social Cognition theory comes directly from Jost and colleagues (Jost et al. 2003). In an extensive meta-analytic review of 88 studies in 12 countries, Jost et al. (2003) found evidence that political conservatives exhibit more death anxiety, are less tolerant of ambiguity, have less openness to new experiences, have less tolerance for uncertainty, have more need for order, structure, and closure, and have more fear of threat and loss when compared to political liberals.

Jost and colleagues are not alone in their study of the psychological underpinnings of political ideology. Other researchers have suggested that conservatives have an increased wariness of out-groups that may be related to heightened feelings of threat in certain contexts (see Skitka & Mullen 2002, 119; de St. Aubin 1996; Stone & Schaffner 1988). Right-wing authoritarianism has a long history of being correlated with and being activated by threat (Lavine et al. 2002; Lavine et al. 2005). Nail and McGregor (2009) observed a movement towards conservative political stances in eight of eight items in two independent surveys of adults, one taken before 9/11 and one taken afterwards. Survey respondents reported increased support for conservatives, George W. Bush, and increasing military spending, and less support for socialization of medicine. Weber and Federico (2007) found that anxious attachment styles were associated with right-wing authoritarianism and mediated by belief that the world was a dangerous place. They also found that avoidant attachment styles were associated with social dominance orientation. The perception of a dangerous world is correlated with right-wing ideologies (Jost et al. 2003), and it is especially strong among political sophisticates (Federico et al. 2009). Hatemi et al. (2013) find that fear dispositions have a modest but significant relationship with conservatism out-group attitudes. Shook and Clay (2011) found that conservatives were more susceptible to conditioning with negative stimuli than liberals, and conservatives were less susceptible to conditioning with positive stimuli compared to liberals.

These correlations suggest conservatives have an aversive cognitive style, behaviorally motivated to avoidance, and in particular, avoidance of threat and the unknown (see also Dodd et al. 2012). This evidence highlights the distinct possibility that political conservatives have a more avoidant cognitive style.

### **Two Theories Measuring Two Different Psychological Constructs**

The two theories discussed above have much common ground. Both sets of authors tend to cite each other in support of their respective findings. In my view, however, it is clear that each theory is focusing on a specific psychological process distinct from the rest. We would benefit from sorting these mechanisms that underlie the two theories, which would allow us to gain much more specificity for our claims.

It is valuable to understand how sensitivity to avoidance (Motivated Social Cognition theory findings) impacts political attitudes. It is also valuable to understand how sensitivity to emotional negativity is correlated to political attitudes (Negativity Bias theory findings). These findings, however, have different underlying mechanisms. Affect is not the same as personality traits, and neither are the same as cognitive style. Yet, research in this field tends to lump all the findings together as if it is one portrait of conservatives.

The Motivated Social Cognition theory posits that conservative belief structures are motivated by psychological needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al. 2003). The Negativity Bias Theory argues that conservative belief structures are correlated with sensitivity to negative affect (Hibbing et al. 2013). Both theories have consistent empirical support. While the two theories frequently crossreference each other, there is little explanation for why both theories could or should co-exist. Or, put another way, regardless of the surface similarities, these theories are ultimately describing different mechanisms as the roots for these correlations. But what are those mechanisms? Is a disposition towards avoidance the same as a dispositional sensitivity to negative affect? How does cognition and emotion intersect to create behavioral outcomes? What we could use is a better specified theoretical link between these two profiles of political conservatism.

## Gray's Theory of Behavioral Motivations

Jeffrey Allen Gray offers a theory of behavioral motivation, which he envisioned as two complementary systems that underlie cognition, emotion, and behavior. The behavioral activation system (BAS) facilitates behavior towards the completion of a direct objective or goal. All of the small behaviors we take during a day are for some purpose, some goal. The goal may be short-term, fleeting, poorly considered, or it may be long-term and strategic. Regardless of what the goal is, and how much we have considered it, we act because we desire some outcome. BAS facilitates action. In contrast, the behavioral inhibition system (BIS) halts ongoing behavior in response to new information. This is our monitoring system, designed to help us react to unexpected changes in our environment. The behavioral inhibition and activation systems blend our cognition with our emotion in order to generate behavior. Here is an illustration of how the dual systems operate. Sitting down with our laptop for work, we decide to have a sip of coffee. The BAS operationalizes the goal with our cognitive processes. We reach for the cup of coffee and bring it to our lips. We experience a mild sense of satisfaction, mild positive affect, which helps to motivate the action. Our goal is nearly achieved, something that we planned in our mind that was aided by emotion. As we look into the mug near our lips, however, we notice that a dead fly is floating on top of the coffee. Seeing the fly cues a disgust reaction, an arousing negative emotion, and we immediately halt our planned action to drink a sip of coffee. This is the BIS facilitating a rapid change of plans by interrupting our ongoing behavior in response to newly detected information. Again, the BIS integrates both cognition and emotion to trigger behavior.

BAS is about going while BIS is about stopping. BAS is connected to feelings of positive emotion, while BIS is associated with feelings of negative emotion. This theory has the potential to articulate the mechanisms underneath both theories as well as link them together directly. Below, we examine Gray's theories in more depth and outline several ways to study dispositional differences of political ideologues.

Researchers in psychology have a number of tools to investigate avoidance motivations and negativity bias, methods that can be used to help illuminate the psychological differences between liberals and conservatives. Below, I describe the most commonly used framework for studying emotion and behavioral motivation in psychology, and then outline methodological tools from cognitive neuroscience that can be used to explore the question of emotional sensitivity in political ideology.

Gray's dual behavioral motivation systems—the Behavioral Inhibition System (BIS) and the Behavioral Approach System (BAS)—are a pair of heuristic devices developed to help explain animal and human behavior (Gray 1972, 1978, 1981, 1988). BIS can be thought of as a process for stopping when a potential threat or reward is detected, while BAS is a process for going forward to enact a plan of action (Demaree et al. 2005). Worded another way, BIS is an aversive system, while BAS is an appetitive system (Carver and White 1994).

BIS is thought to facilitate attention or sensitivity to cues of punishment, danger, avoidance, and novelty. Gray suggests that BIS functioning is responsible for feelings such as fear, anxiety, frustration, and sadness in response to cues (Carver and White 1994; Gray 1972, 1978, 1981, 1988). BIS functions to interrupt current behavior in order to process these cues in preparation for a response. High BIS activation is associated with enhanced attention, arousal, vigilance, and anxiety, and very strong BIS corresponds to anxiety-related disorders (Fowles, 1988; Quay, 1988).

The complementary system to the BIS is the BAS, which represents a motivational system sensitive to signals of reward, nonpunishment, and escape from punishment. BAS facilitates approach towards a reward (going toward), but also facilitates active avoidance away from a punishment (going away) (Amodio et al. 2008).

BAS has been associated with feelings of optimism, joy, aggression, and anger (Gable et al. 2000; Gray and McNaughton, 2000; Wingrove and Bond, 1998; Harmon-Jones 2003).

The BIS and BAS scales developed by Carver and White (1994) have been used extensively by psychologists to measure dispositions for behavioral approach and behavioral inhibition, as well as emotion. BIS, in particular, is associated with selfregulation (Amodio, et al. 2008) but has not been examined in the context of political ideology. Marcus et al. (2000) use Gray's BIS and BAS systems in their theory of Affective Intelligence, which demonstrates that anxiety increases political learning and enthusiasm increases political participation. They do not connect it to political ideology, however, which presents a clear opportunity to look at BIS and BAS in a survey questionnaire as a way to test for the psychological dispositions that Hibbing and Jost utilize in their respective theories.

## Hemispheric Asymmetry and Psychological Dispositions

Neuroscience has contributed in a significant way to the study of emotional affect and behavioral motivation, and we will need these insights in order to explore the psychological dispositions of avoidance sensitivity and negativity bias. By borrowing some of these methodologies, many grounded directly in Gray's work, we can find new tools to study how political ideology may be related to emotion and behavioral motivations.

One such tool is the measure of hemispheric asymmetry. Tomarken et al. (1990), found that participants in an electroencephalography (EEG) experiment who had greater

resting right frontal activity responded with more intense negative affect to negativelyvalenced film clips, particularly those involving fear or threat (see also Wheeler et al. 1993). Davidson and colleagues (1993, 1998a, and 1998b) suggest relatively greater left frontal activity corresponds with trait tendencies toward a general withdrawal or avoidance system, a hallmark of Gray's behavioral activation system (BAS). This finding has been empirically verified by a number of different research groups (e.g., Carver and White 1994; Coan and Allen 2003, 2004; Harmon-Jones and Allen 1997, 1998; Sutton and Davidson 1997). Based on these findings, researchers now believe that the pattern of frontal hemispheric asymmetry is a biomarker for avoidance motivations and negativity bias. Yet, they were not able to distinguish between the two at the moment. As demonstrated, there are detectable left/right hemispheric differences in the strength of EEG signals in frontal electrodes, and this asymmetry may be a moderator and/or mediator of emotions, such as fear and anxiety (Coan and Allen 2004).

Coan and Allen, writing in 2004, report than the relationship between cortical asymmetries and emotion had been established by over 70 studies (2004, 7). In their review, they establish that resting levels of neural activity, as well as state-based activation, in the prefrontal cortex are correlated with trait predispositions and changes in emotional state. These findings suggest that brain systems tapped by frontal EEG asymmetries may moderate (in the case of activity) and mediate (in the case of activation) emotional responding. The consensus in this literature is that relatively greater right hemispheric frontal activity is associated with tendencies toward a general avoidance or withdrawal system, which is then often correlated to negative affect (Coan and Allen 2003a; Coan and Allen 2003b; Davidson 1993; Harmon-Jones and Allen 1997, Sutton and Davidson, Davidson 1998a,b). Davidson's influential approach/withdrawal motivational model of emotion proposes that left frontal activity (either as state or as trait) indicates a propensity to approach or engage a stimulus, while relatively greater right frontal activity indicates a propensity to withdraw or disengage from a stimulus.

Other researchers have confirmed this relationship. For example, Field et al. (1995) and Fox et al. (1996) found evidence that children with greater right frontal activity at rest were more inhibited socially, and scored lower on social competency. Schmidt and Fox (1994) found a relationship between frontal EEG asymmetry and measures of sociability in adults. Those scoring low on measures of sociability had relatively greater left frontal activity. Schmidt et al.1999 found shyness had greater right frontal activation. All of these studies from developmental psychology researchers suggest a relationship between avoidance and right frontal activity. EEG asymmetry promises to be a useful tool for studying avoidance sensitivity in political liberals and conservatives. Below, I briefly discuss political ideology as studied from a political neuroscience perspective, noting that no research has yet connected EEG asymmetry and emotional sensitivity with political ideology.

# The Go/No-Go Task and Psychological Dispositions of Conservatives

To date, three studies have explored the connection between dispositional styles in the brain and political ideology. None, yet, directly measure avoidance sensitivity or negativity bias for liberals and conservatives, which is the project that this dissertation picks up. Amodio et al. 2007 hypothesized that differences in the cognitive styles of liberals and conservatives might reflect basic differences in information processing mechanisms, such as those involved in conflict monitoring—a neurocognitive process for detecting discrepancies between response tendencies and one's higher-level intentions. To test this prediction, Amodio and colleagues compared participants' selfreported political orientation with behavior and neural activity on a Go/No-Go task (explored in detail in Chapter Six). Consistent with the model of political ideology as motivated social cognition, liberalism was associated with greater behavioral accuracy on No-Go trials of the task. Furthermore, liberals' EEG signal exhibited significantly larger event-related potentials (ERP), indicative of greater anterior cingulate cortex (ACC) response on No-Go trials than did conservatives, supporting the hypothesis that political orientation may be linked to basic neurocognitive processes for dealing with new and unexpected information.

Weissflog et al. (2010) also assessed ERP responses and political ideology, this time in a sample of Canadian university students who completed the Go/No-Go task. As in the Amodio et al. 2007 study, a stronger liberal orientation was associated with larger No-Go N2 amplitudes, indicating greater conflict-related ACC response, and thus replicating the results of Amodio et al. 2007. In addition, larger No-Go N2 and Event-Related Negativity (ERN) amplitudes in these college students were correlated with greater endorsement of egalitarian values and lesser endorsement of right-wing authoritarianism.

Although they did not look at political attitudes or ideology directly, Inzlicht et al. (2009) indicated that higher religiosity was correlated with smaller ERNs in response to errors on a color-naming Stroop task (1937). As Jost and Amodio (2012) note, there is a strong association between conservatism and increased religiosity, and so this study may also be considered broadly consistent with the results of Amodio et al. 2007.

The intent of the above discussion is to open the subject matter, but more discussion and clarification of these studies is explored in more detail in Chapter Six, which explains the Go/No-Go task conducted for this dissertation, and presents a comprehensive look at what these findings mean for political ideology. In sum, this chapter outlined the two major theories of the political psychology of ideology as being fundamentally about (a) sensitivity to avoidance and (b) negativity bias. Then, it introduced Gray's theory of behavioral motivation. By using tools for studying Gray's Behavioral Inhibition System and Behavioral Activation System, such as survey questionnaires, hemispheric asymmetry studies, and the Go/No-Go behavioral task, we will be able to focus on these two important psychology of ideology. The next three chapters present empirical studies of these psychological processes in a sample of liberals and conservatives, through a survey questionnaire and two laboratory experiments.
# **CHAPTER FOUR**

# SURVEY DATA AND DISPOSITION ANALYSIS

One of the central objectives of this dissertation is to expand research in the area of root physiological differences in avoidance sensitivity and negativity bias between liberals and conservatives through an investigation of Gray's theories of behavioral motivation systems. This chapter furthers this goal by (a) characterizing my participant pool for experiments in later chapters and (b) exploring correlational analysis between political ideology and self-reported dispositions towards behavioral inhibition and behavioral activation. Later chapters go beyond questionnaire self-reports to bring these questions into the lab.

This chapter describes and characterizes a large sample (n = 466) of non-student employees drawn from Loyola University Chicago who completed a detailed survey about their personality traits and social-political viewpoints. From this larger sample, I drew a subsample (n = 51) who participated in two behavioral neuroscience experiments that become the focus of this dissertation in later chapters. Therefore, this sample becomes an important starting place for characterizing the group of participants who go on to complete the lab experiments. Overall, we find that this sample is not nationally representative and skewed towards liberals, although a meaningful minority of participants do hold deeply conservative beliefs. This sample is considerably more representative than a typical sample of college undergraduates. Additionally, as introduced in the last chapter, Carver and White (1994) have developed a questionnaire that allows researchers to estimate dispositional sensitivity to behavioral inhibition and behavioral activation, which will help us investigate avoidance sensitivity differences between liberals and conservatives in a large sample size.

Behavioral motivation is an interesting, important psychological and physiological process that could be of importance for political ideology. Borrowing heavily from the theories of Gray (1990; Gray and McNaughton 2000), psychologists Carver and White (1994) devised a questionnaire to tap into behavioral motivations with a survey instrument. This measure has become a seminal contribution, with over 4,000 citations and continuing to grow in influence. While behavioral motivations will always be best measured through experimental manipulation and observation of actual behavior, there are numerous situations where a survey can provide important information in a cost effective way. Experimental costs for a large sample may be prohibitive, and/or time available for experiments may be scarce. A survey instrument can be issued quickly, at lower cost of time and effort. Finally, observational data from a survey can serve as an effective way to assess personality dispositions towards behavioral inhibition and approach in a large sample. Observational data about selfdescribed personality traits can be useful to investigators.

The rest of this chapter follows from these goals. Primarily, this chapter details the social, political, and demographic characteristics of the large sample, which itself will serve as the participant pool for two behavioral experiments in subsequent chapters. Additionally, this survey collected observational data on sensitivity towards Gray's dual systems of inhibition and activation, which is hypothesized to be related to political ideology. By combining the observational data in this chapter with the experimental data in later chapters, we use a multiple method approach to answering the central question of the relationship of Gray's BIS/BAS systems to political ideology.

### **Behavioral Inhibition and Activation**

J.A. Gray hypothesized that two general motivational systems for behavior underlie our emotional and cognitive processing (1990; Gray and McNaughton 2000). The behavioral activation system (BAS) facilitates action in pursuit of a desired outcome. The behavioral inhibition system (BIS) is a surveillance system, intended to monitor the environment for unexpected changes that may threaten or change our goal-directed behavior.

As described in the past chapters, BAS is about going while BIS is about stopping. Additionally, research has demonstrated that increased BAS has been connected to feelings of positive affect, and increased BIS is associated with feelings of negative affect. BAS is a process for going forward to enact a plan of action while BIS interrupts when a potential threat or reward is detected. For these reasons, Carver and White (1994) describe the BAS as an appetitive system, while BIS is an aversive system. BIS draws attention to cues of punishment, danger, and novelty, and is theorized by Gray to be responsible for negative emotions of fear, anxiety, frustration, and sadness. BAS motivates behavior towards a reward or a desired interaction and is associated with feelings of optimism, joy, aggression, and anger. Recall from earlier discussions that approach can be positively valenced (optimism, joy) or negatively valenced (aggression, anger) as long as the emotions motivate engagement with a stimulus instead of avoidance.

I have argued in past chapters that Gray's dual systems underlie two of the most prominent theories about the political psychology of political ideology. First, the Motivated Social Cognition Theory (of John Jost and colleagues) is built upon the theory that conservatives are motivated by a constellation of psychological factors related to fear, threat, and reducing uncertainty. These core psychophysiological functions, especially of fear and threat, are a kind of behavioral avoidance clearly rooted in Gray's behavioral inhibition system. While it is true that uncertainty avoidance is not purely the same psychological construct as behavioral inhibition, the overlap is substantial. As Jost and colleagues write, uncertainty avoidance in conservatives is part of a general predisposition towards caution and general avoidance (Jost et al., 2003; Jost et al., 2009). Jost and colleagues also frequently mention a study of children that describes "inhibited" preschoolers as growing up to become political conservatives (Block and Block 2006; as cited in Jost et al., 2009; Jost, 2009). Given that motivational avoidance and inhibition have substantial overlap, and Jost's theories find great empirical support for the claim that conservatives are more avoidant, we can also predict that conservatives would be more inhibitory as well.

According to the Negativity Bias Theory, conservatives exhibit stronger reactions to negativity, and focus more visual attention to negative stimuli (Hibbing et al 2014). Negative affect includes many different emotions, such as fear, threat, sadness, and anger. (The connection between discrete emotions and political ideology is picked up with force in Chapter Five of this dissertation.) Negativity bias is also connected to feelings of fear and threat, and they motivated behavioral avoidance of the stimulus that triggers the negative emotion.

# Hypotheses

In sum, the key hypothesis for how Gray's dual systems may be related to political ideology is that conservatives should have greater behavioral inhibition. Jost's work on conservatives and avoidance and Hibbing's work on negative emotions each point to this conclusion.

Behavioral activation predictions are less clear, and will be approached as exploratory in this work. Behavioral activation is related to sensitivity to approach, as well as positive emotion. Because being sensitive to negative emotion cannot speak to whether someone is also sensitive to positive emotion, we do not have direct hypotheses from current work. There is nothing in the Jost or Hibbing theories that suggest conservatives or liberals have a relationship to behavioral approach or positive emotion. Behavioral inhibition and behavioral approach are conceptually separate from each other, not two ends to the same phenomena. Positive and negative emotion, too, are not inverses of each other, but separate constructs.

# Method

#### Participants

Participants were recruited from the campus staff of Loyola University Chicago from February 2012 until November 2012. This recruitment method was inspired by Kam et al. 2007 as a good convenience sample to utilize when an undergraduate sample poses research problems. As Sears 1986 noted, college aged students are likely to have more weakly held social and political attitudes (522) and may exert more cognitive effort than the typical person due to the emphasis on accuracy in a school setting (525). Of particular importance is their young age and their (understandable) lack of broad life experiences compared to national populations.

I sent 1144 emails to non-student adults employed by Loyola University Chicago, inviting them to take our online survey on political attitudes and personality. The survey was created and administered on Opinio software

(http://www.objectplanet.com/opinio) with access provided by Loyola University Chicago. My outreach included all employees at the Lake Shore Campus or the Water Tower campus, except those who were academic faculty or higher-level administrators (above the title of "director"). Three follow-up attempts were made. First, a reminder email was sent after 30 days. Second, I also mailed physical copies of the survey through interdepartmental mail approximately three months after my initial email contact. Third, I sent a final email reminder. Overall, I received a total of 466 completed surveys (40.3% response rate). I did not offer compensation for completing the survey; however, the materials mentioned that successful completion of the survey may make one eligible for a follow-up neuroscience study which would pay more than \$40/session. 280 of those who completed the survey (60%) also volunteered to be considered for the follow-up study in the lab. Eventually, a selection of 51 of these volunteers became subjects of the behavioral and neuroscience studies in later chapters of this dissertation.

Participant demographic background. This sample was disproportionally female (286 females, 176 males, and 4 refused). The median age of the sample was 38 years old, and age ranged from 20 to 93 years old (*mean* = 41.33, *sd* = 13). Ninety-three is not an error: a small number of participants of advanced age live and work at Loyola University Chicago in various departments, but are especially found in numerous Jesuit groups and organizations housed on campus. 72 percent of respondents identified as white, 11.6 percent as black, 6.3 percent as Asian, and 10.1 percent as other/multiracial. The sample was disproportionally educated compared to the national population, which is to be expected given these are workers employed by a university. 38 percent hold a bachelor's degree and 36 percent hold a master's degree. 8 percent of the sample had a professional degree or Ph.D. Only 16.8 percent of the sample did not have at least a bachelor's degree.

*Party identification*. Party identification fit expectations (Kam et al. 2007) that the campus staff would largely resemble the Democratic politics of the surrounding region generally. 79 percent of the respondents identified as Democrats, 9 percent as independents, and 11 percent as Republicans. These numbers include those who said they leaned toward a party. Only 11 participants in the entire sample considered themselves "strong Republicans" (2.4 percent), an exceptionally low amount compared to the national population. This sample skewed heavily Democratic, an expected result for a university located in Chicago.

#### Measurement

Broadly, this survey questionnaire addressed two categories of attitudes: political ideology and behavioral motivations. Full question wording is provided in Appendix A.

Political ideology. Three different question sets were used to assess political ideology, with each designed to tap into a different conceptual element of political ideology. Much previous research has relied on the 7-point self-identification Likert scale. Following the standard set by the American National Election Studies, participants are asked to place themselves on a 7-point scale between "very liberal" and "very" conservative, with the middle labeled "centrist." Self-placement on this scale is strongly predictive of party identification as well as vote choice in national elections. The self-placement scale also allows a measure of how the individual sees their own political ideology, providing a summary judgment of how the individual would selfclassify their own belief structure. However, there are some potential weaknesses. First, the respondent must have a sense of what the terms conservative, liberal, and centrist mean, which requires a certain amount of political knowledge. Second, like all self-reports, there are a number of opportunities for purposeful and inadvertent misrepresentation. Third, this scale presumes a one-dimensional ideological spectrum, which, as discussed in prior chapters, may be problematic.

To compensate for weaknesses with the self-report, I also used a variation of the Wilson-Patterson Conservatism scale of political ideology (1968), which asks respondents for their opinion on a variety of contemporary public policy topics and then aggregates the results into an overall ideology score. Political ideology has often been conceptualized as the summary of your various political attitudes—after all, if you take the liberal view on most political arguments, you are probably liberal. One way to measure political orientation, thus, may be to ask your opinion on a variety of political topics and then see what the aggregate picture is. The Wilson-Patterson scale has been a popular measure of political orientation by using the summation of various political attitudes on political issues of the day. These question items need to be updated to the contemporary political setting, and so I adopted the modified Wilson-Patterson battery utilized by Smith et al. (2011). Twenty questions are included in the Smith et al. variant.

Additionally, although the theoretical conception of political ideology should be universal to human beings, there is little question that answers to contemporary political ideology question batteries are going to be bounded by cultural and national contexts. This sample is drawn from the United States, and the question sets listed above have been primarily used on subjects in the United States. To broaden our reach, I use four questions adopted from the World Values Survey that offer attitudes on competition, income inequality, public ownership, and private responsibility. These questions have been used reliably across the globe to look at attitudes toward political economy issues, especially concerning economic justice. These questions are useful as a way to measure collectivism versus individualism in economic policy, and they offer a more diverse measuring stick with applications outside of the United States. This scale is referred to as the World Values Survey Economic Justice scale.

*Multidimensional ideology*. The Wilson-Patterson items collected here are usually utilized as a measure of unidimensional ideology, running from liberal to conservative. However, the underlying items of the scale could be analyzed along additional dimensions. Commonly, ideology is broken into two dimensions: economic and social (Treier and Hillygus 2009; Ellis and Stimson 2012). Economic (or, fiscal) conservatism suggests a strong preference for free markets and opposition to government regulation and taxation. In contrast, social conservatism is a skepticism towards social changes and preference for more traditional social mores, such as the traditional family structure. Splitting ideology in two dimensions allows for a better measurement of libertarian (economically conservative, socially liberal) and authoritarian (economically liberal, socially conservative) ideological archetypes.

I account for, and anticipate, multidimensional ideology in two ways in this chapter. First, I utilize factor analysis below to determine how well multidimensional ideological values fit this sample. Second, however, as this sample is not nationally representative, it would not have been appropriate to use the factor analysis to derive dimensions. There is no guarantee that employees of Loyola University Chicago would

75

represent the normal dimensions of political ideology (social and economic factors). So, I created two subscales designed to tap into economic and social dimensions based on the well-known and reliable Wilson-Patterson scale and prior researchers (Treier and Hillygus 2009). The economic conservatism subscale includes these items: welfare spending, tax cuts, small government, and foreign aid. The social conservatism subscale items chosen were: prayer in school, legal pornography, illegal immigration, death penalty, the Patriot Act, biblical truth, gay marriage, and legal abortion. These two subscales are theoretically derived, based on researchers work on the multidimensionality of political ideology (Treier and Hillygus 2009; Ellis and Stimson 2012).

Behavioral Inhibition and Behavioral Activation Scales. BIS and BAS were assessed in the conventional manner described in Carver and White (1994). All BIS/BAS questions ask the respondent to say whether they strongly agree, somewhat agree, somewhat disagree, or strongly disagree with the statement. These question batteries assess an individual disposition towards each motivational system (aversive and appetitive.) BIS has a seven items and BAS has thirteen. Specific items are listed in Appendix A with the rest of the survey.

# Results

# Descriptive Statistics

# Table 1. Descriptive Statistics of Survey Measures.

	n	mean	sd	median	min	max	range	skew	kurtosis	se
age	456	41.33	13.42	38	20	93	73	0.57	-0.66	0.63
Self-report Ideology	455	3.04	1.41	3	1	7	6	0.67	-0.21	0.07
Wilson-Patterson (Full)	466	<mark>6.</mark> 7	3.79	6.25	0	16.5	16.5	0.36	-0.68	0.18
Wilson-Patterson (Economic Subscale)	466	1.66	1.3	2	0	4	4	0.28	-1.03	0.06
Wilson-Patterson (Social Subscale)	466	2.62	2.25	2	0	8	8	0.47	-0.93	0.1
World Values Survey Economic Justice Scale	444	19.51	6.5	19	0	36	36	0.32	0.16	0.31
Behavioral Inhibition	457	12.8	3.22	13	2	21	19	-0.09	0.23	0.15
Behavioral Activation	443	23.91	4.76	24	10	38	28	0.24	-0.06	0.23

Table 1 displays the descriptive statistics for the major variables of interest. The varying sample sizes of the statistics reflect the existence of missing data from the survey instrument.



The self-report ideology scale (*mean* = 3.04, *median* = 3) skews leftward, reflecting greater self-described liberalism in the sample. While the seven point scale has representation at each score, extreme conservatives are scarce. Figure 1 illustrates how the distribution of ideology differs from a nationally representative survey. The American National Election Study of 2012 includes the same survey question that I utilize to generate a self-report of political ideology on a seven point scale.

The Wilson-Patterson Conservatism scale has a similar left-ward skew (*mean* = 6.7, *median* = 6.25). The theoretical range goes from 0 to 20, but the most conservative members of this sample cap at 16.5. (Participants that answered neither "support" nor

"oppose" are scored as a 0.5, as per Smith et al. 2011.) As with the self-report measure, moderate liberals and strong liberals are overrepresented while conservatives are underrepresented. Both of the Wilson-Patterson subscales follow similar patterns as the full scale, except the sample is far more socially liberal than economically liberal.

The World Values Survey Economic Justice Scale is considerably more moderate than the other measures, reflecting two insights. First, the WVS is written for a global audience, and Americans are more economically conservative than most countries. Second, this scale has no social conservative elements in it, and thus, the very liberal viewpoints on social issues does not weigh the measure toward the left in this scale as much as others. This scale features of mean of 19.51 and median of 19, on a 0 to 36 point scale, and participants in the survey had values at both poles of the scale. While there is a left-ward skew to the data, attitudes about economic justice are not as strongly left as the other ideological measures, suggesting that the sample may be more socially liberal than it is economically liberal (which is consistent with the story told by the Wilson-Patterson Social conservatism scale, which has a median value of 2, with a scale maximum of 8). Figure 2. Distribution of Political Ideology.



Figure 2 visually represents the distribution of ideology in the survey sample across the three ideological measures of self-placement, Wilson-Patterson Conservatism scale, and the World Values Survey Economic Justice scale. Note that the range of all xaxes are drawn from the scale minimum to the scale maximum, *not* the observed range of the data from this sample. For example, although no subject scored a maximum of

20 (100 percent conservative) on the Wilson-Patterson scale, the x-axis includes that maximum value of 20. Drawing x-axes to the scale dimensions facilitates better visual interpretation of the ideological preferences of the sample within each plot, as well as comparison between each plot.

Figure 3. Distribution of BIS/BAS Scales.



81

Figure 3 demonstrates the distribution of Carver & White's BIS/BAS scales of behavioral motivation in the sample. Both shapes are essentially normally distributed around the middle of each respective scale, with few outliers and zero extreme values to the far left or right.

# **Correlation Matrix**

Table 2. Correlation Matrix of Important Variables.

	Self-report	W-P (Full)	W-P (Econ)	W-P (Social)	Econ. Justice	BIS
Self-reported ideology						
Wilson-Patterson (Full)	0.690***					
Wilson-Patterson (Econ)	0.507***	0.778***				
Wilson-Patterson (Social)	0.631***	0.891***	0.521***			
Economic Justice	0.551***	0.661***	0.603***	0.523***		
Behavioral Inhibition	-0.164***	-0.227***	-0.170***	-0.179***	-0.154**	
Behavioral Activation	-0.054	-0.031	0.043	-0.076	0.041	0.103*

\* p < 0.05; \*\* p < 0.01; \*\*\* p < 0.001. Correlation tests are two-tailed.

Note: All ideology measures scaled so that lower value is more liberal, higher is more conservative.

Table 2 displays the correlation matrix for the variables of interest in this study,

with Pearson's correlation tests. Below is a visualization of these relationships.



These relationships can also be clearly seen in Figure 4. The darkness of the shading and the size of circle both indicate the strength of the correlation, with blue coloration representing a positive correlation and red being negative. Any relationship that is not statistically significant to the p = 0.05 standard is represented with a blank (all white) space on the matrix. As can be observed, all relationships between the variables

do reach statistical significance, except for those between the BAS measure and any political ideology measure.

The table and visualization demonstrate that all five ideological measures are tightly correlated, forming a strong cluster of related measures. Additionally, all measures of ideology are negatively correlated with the behavioral inhibition scale measure, with mild strength. In contrast, BAS has no relationship with any ideological measure, and only a mild positive relationship with BIS.

#### Scale Reliability

Often, social science research relies on a bundle of questions that are used to approximate some kind of underlying construct. For example, some researchers believe that political ideology cannot be accurately assessed by the simple question, "what is your political ideology?" Participants may not be consciously aware of their political ideology, they may not know what political ideology means, they may be overly influenced by a negative association with a political label, or for various other reasons may misrepresent or miscalculate their own political ideology. Researchers facing this challenge may wish to employ a battery of questions such as the Wilson-Patterson scale, which attempts to assess political ideology by asking twenty questions about contemporary political issues. By looking at the big picture of these twenty issues, it may be possible to triangulate the political ideology of a person without needing to rely on the subject's self-assessment. In these cases, it would be useful for researchers to know (and, also, to demonstrate to reviewers) that the bundled items were appropriately chosen to estimate the latent construct.

In situations like this, social scientists have come to rely on Cronbach's alpha to measure the internal consistency of a scale. Cronbach's alpha is a coefficient that examines to what degree the items are related as a group. A higher alpha coefficient signifies more internal consistency, meaning that the items are more closely measuring the same thing. In practice, researchers frequently employ Cronbach's alpha to justify their scale as assessing a single unidimensional latent construct, a practice that has been widely criticized in the psychometric literature. Psychometric researchers have established that Cronbach's alpha cannot determine the dimensionality of the underlying construct, but instead merely the degree to which the items are consistently related to the construct (Green et al. 1977; Schmitt 1996). The underlying construct may be unidimensional or multidimensional and still yield a high alpha score.

Interpreting alpha scores is based on consensus and rules of thumb, much like assessing p-values. The general practice is that an alpha score greater than 0.70 is "acceptable," with 0.80 considered "good," and greater than 0.90 to be "excellent" (George and Mallery 2003). These are guidelines and not hard rules, however. Alpha scores of 0.60 are sometimes used with caution, particularly if the scale has been validated elsewhere in a more nationally representative sample. It is not unusual for studies with non-representative samples to have a weaker Cronbach's alpha coefficient. Such studies can still justify the use of a scale if it has empirical support from more

definitive articles with more complete samples.

Table 3. Cronbach Alpha Scores of Scale Measures.

	Cronbach's Alpha
Wilson-Patterson (Full)	0.803
Wilson-Patterson (Econ)	0.611
Wilson-Patterson (Social)	0.764
Economic Justice	0.700
Behavioral Inhibition	0.801
Behavioral Activation	0.843

Table 3 lists the Cronbach's alpha scores for the scales employed in this study. The Wilson-Patterson Full Scale, and the twin behavioral motivation scales of Carver and White (1994) all maintain healthy alpha scores of around 0.80, signifying "good" internal reliability. All three scales are widely used in academic research, and so it is not a surprise that they replicate well with this sample. The two Wilson-Patterson subscales derived from theoretical perspectives have more questionable alpha scores, with 0.611 and 0.764. Prior literature has not attempted to derive an economic conservatism and a social conservatism factor structure from the Wilson-Patterson battery, and these alpha scores suggest caution when interpreting findings. As this sample is not nationally representative, we might not be surprised to discover that the theoretical factor structure of economic and social conservatism does not fit well the liberal denizens of the workforce of Loyola University Chicago. (We will see shortly that economic conservatism is not so clear cut in this sample according to factor analysis.) Because of the clear utility in investigating ideology with these two factors, this work will proceed in utilizing the two scales, but interpretation will be exploratory and cautious. Finally, the World Values Survey Economic Justice scale is also novel to this work. With an alpha score of 0.700, it hovers in the "acceptable" range of internal consistency.





Proportion in survey that gave liberal (dark) and conservative (light) response

Earlier, we have already viewed the distribution of aggregate Wilson-Patterson scores in this sample. Figure 5 explores the ideological polarization of the sample on a per-item basis within the Wilson-Patterson scale. Each item was labeled simply with the labels on the y-axis (e.g. "Prayer in Schools" and "Gay Marriage") and participants are directed to either "support" or "oppose" the topic by choosing the answer that is closest to their belief. These answers are coded as liberal (or conservative) if they support a liberal (or conservative) belief. For example, welfare spending is traditionally a liberal belief—if a participant chose "support", that would be coded as a liberal answer.

Items in Figure 5 are arranged from the item which received the greatest proportion of conservative responses (patriotism) through the item that received the greatest proportion of liberal responses (women's equality). Overall, there is considerable political diversity across the span of issues, although there is a noticeable left-ward skew to the attitudes reported. Only three items had more conservative support than liberal support (patriotism, free trade, and small government). On the 17 other items, the liberal viewpoints is endorsed by a majority of respondents. Although failing to capture the majority on most issues, a healthy conservative minority is readily apparent in most of the remaining items.

There are three exceptions, however—three issues that have very little support for the traditional conservative position in this sample. Gay marriage, pollution control, and equality for women each feature 95% of the sample taking the liberal side of the issue. While women's equality and pollution control could be explained by changing

89

societal attitudes towards women as well as the largely bipartisan support of pollution control in a major American city, the lack of political diversity for the issue of gay marriage is somewhat puzzling. Gay marriage is one of the most polarizing issues throughout the 1990s and 2000s, defining one of the sharpest cleavages between liberals and conservatives. It is possible that the conservative, educated denizens of Chicago had already shifted the gay marriage argument out of the culture wars, in acknowledgement that times have changed or will change (this survey was administered in 2012, and shortly afterward in 2015, the Supreme Court in *Obergefell v. Hodges* ruled that gay marriage was a constitutional right, effectively winning that pitched battle for the liberals.) More likely, though, the lack of conservative opinion about gay marriage reflects the dearth of social and religious conservatives in my sample.

Overall, the distribution of Wilson-Patterson items helps illustrate the left-ward skew of the subjects, but also note that conservative viewpoints are still represented. Below, we turn to a factor analysis in order to sort out the dimensionality of the Wilson-Patterson data.

# Wilson-Patterson Factor Analysis

As discussed in earlier chapters, political scientists increasingly view political ideology as multidimensional, and yet, the self-report measure and the political issue attitude battery employed here are unidimensional, assuming participants to be either liberal or conservative, or somewhere in-between. Yet, in the public, we know this simple picture to be false. There are some in the public who are socially liberal, yet

90

economically conservative. They may or may not identify as political libertarians, or even understand what that term means; and yet, they fit a profile that cannot be easily placed on a bidirectional scale.

This sample is not nationally representative, so an exploration of the underlying factor structure of ideology in this sample cannot easily serve to teach us about the multidimensionality of attitudes in the general public. Based on prior research that does use representative samples, we have expectations that there ought to be an economic factor and a social factor. Earlier, this chapter describes how I developed two subscales of the Wilson-Patterson battery that attempts to capture those two elements.

There are least two reasons to explore the underlying factor structure of this sample, even if it is not nationally representative. First, we ought to see how well the theorized social and economic factor structure fit this sample. Does an economic factor emerge? Is a social factor observable? And, second, to what degree are there any latent ideological factors at all within the sample? If issue attitudes did not cluster whatsoever in this sample, it would suggest that Conversian political ideology (where issue attitudes are constrained around some kind of latent underlying value structure) is not a relevant issue.

Confirmatory factor analysis requires the researcher to stipulate a theory for how many factors should be present. I hypothesize that there will be two factors, a social and an economic kind of conservatism, based on contemporary research on this subject. My confirmatory factor analysis was conducted on the twenty WilsonPatterson items, utilizing a method of oblique rotation called promax, and with a model specification of maximum likelihood. The factor analysis was performed in R Statistics, using both the base package native to R as well as the supplemental R package "psych" written by William Revelle of Northwestern University (Revelle 2016). This factor analysis procedure will yield two sets of loadings (one for each proposed factor) for each item of the Wilson-Patterson scale. Figure 6. Factor Analysis Scatterplot.



Confirmatory Factor Analysis of Wilson-Patterson Items in Sample

I will present two visualizations that help illustrate the factor structure apparent in this sample. First, Figure 6 represents the factor analysis as a scatter plot. Each axis represents one of the factors, and the W-P item is plotted on the Cartesian plane. In this figure, you can observe two clear clusters of attitudes have emerged (I have colored them black and gray). The gray circles are centered at the factors' mean loadings, and are sized to help visually connect the clusters.

The black items load very strongly on Factor Two (y-axis), while being fairly weak on Factor One (x-axis). This factor includes many socially and religiously conservative issues: legal pornography, legal abortion, gay marriage, biblical truth, and prayer in school. I interpret this factor as a good representation of the "social" conservatism factor we expected to find. Note that equality for women, punishing illegal immigration, and death penalty are also socially conservative viewpoints, but they did not load strongly on Factor Two as would have been expected. Equality for women simply had no variability in this sample, and so the factor analysis could not adequately sort this item. On the death penalty, Illinois has had extreme problems with the death penalty for decades, with a Republican governor in 2000 declaring a moratorium on death penalty cases after thirteen people were found wrongfully convicted. So, while the death penalty ought to be in this factor, the context of living in Illinois probably altered the normally expected factor structure.

The gray items, in contrast, do not have so clear a theme. These issues span from international affairs and national defense, to free market economics, to immigration and the size of government. Factor One does not appear to be an economic factor, as expected, but instead it is a kind of general conservatism. For example, gun control, patriotism, the Patriot Act, and the death penalty are not primarily economic issues. 94



# Wilson-Patterson Two Factor Loadings

In Figure 7, the loading information is presented in a different form which will aid in demonstrating how cleanly the factor analysis divides the items. The length of the bar represents the strength of the loading. Generally, the items cleanly position in one factor and not the other (the exception being free trade and equality for women, which had almost zero variability within the sample anyway). Overall, we find support for the idea of a two-factor system of political ideology in this sample, although the factors are not cleanly a social factor and an economic factor.

# **Ideology and Behavioral Motivations**

Lastly, we turn to Gray's behavioral motivation systems, the behavioral inhibition system and the behavioral activation system. This is an active test of my hypotheses in this chapter, where I expect conservatives to have more self-reported inhibition compared to liberals. Utilizing Carver and White's (1994) scale, we compare how much behavioral inhibition and activation is self-reported by liberals and conservatives in the sample.

Figure 8. Behavioral Inhibition and Two Measures of Ideology.



Wilson-Patterson Conservatives report lower behavioral inhibition N = 457



We observe a modest negative correlation between behavioral inhibition and political ideology, with conservative participants reporting less behavioral inhibition. This is precisely the inverse of what we would hypothesize, given the theories of Hibbing and Jost. Contrary to expectations, it is the liberals in this sample who self-report having a disposition towards behavioral inhibition.

#### [r = -0.046, p = 0.34] Behavioral Activation Scale 30 ė ¢ • ¢ ¢ . ¢ • ė ė ٠ ۲ ۲ • • ¢ ¢ • å • Ċ 20 -• • é . ¢ ۲ ¢ é 10 -Moderately \_ Liberal Very\_ Conservative Very\_ Slightly\_ Slightly \_ Conservative Moderately\_ Conservative Moderate

# Figure 9. Behavioral Activation and two Measures of Ideology

Ideology is not correlated with behavioral activation scale N = 432

Self-report Political Ideology



W-P ideology is not correlated with behavioral activation scale N = 443

For exploratory purposes, we also investigated how self-reported disposition towards behavioral activation may be related to political ideology. As observed in Figure 8, there is a robust null finding for this relationship.

# Discussion

This chapter contributed to the larger dissertation project in two ways. First, this chapter detailed the politics and traits of the larger sample from which experiments in the next two chapters will draw their participants. Second, this chapter explores the correlation between various measures of political ideology and Gray's behavioral inhibition system and behavioral activation systems by way of the Carver and White (1994) questionnaire. Future chapters, which investigate experimental findings from lab

work, have a smaller sample size than what was presented here. Thus, this survey work allows a test of the hypotheses about avoidance motivations and negative affect in a large sample before moving to smaller ones.

The hypothesis for the behavioral inhibition system did not go as expected. Contrary to expectations gleaned from Hibbing and Jost, behavioral inhibition (at least, the self-report) appears to be associated with political liberals in this sample. Also, behavioral approach appears unrelated to political ideology. Future chapters will continue exploring these hypotheses by going beyond the survey questionnaire and into the laboratory.

# **CHAPTER FIVE**

# **HEMISPHERIC ASYMMETRY ANALYSIS**

This chapter focuses on dispositional differences between liberals and conservatives in terms of their trait avoidance and trait emotional sensitivity. A trait is a relatively stable disposition that gives the individual a propensity to act in a certain way. Our personality is comprised of thousands of traits. We can readily think of examples of personality traits such as careful, trusting, or pessimistic. A more trusting individual has a propensity to trust, independent of context. Traits often lead to behavior, but they are not determinative. They only prejudice behavior, not dominate it. There are many situations where even a trusting person will choose not to trust. All other things being equal, however, a person with the trusting trait would be more likely to trust. Thus, traits are conceptually independent from circumstances. Interactions like context, environment, mood, and stimuli are temporary and fleeting (experiencing a "state"), while the trait is an enduring characteristic of the individual. A state is a temporary experience, while a trait is a core disposition to the individual.

My overarching argument is that approach-avoidance processes and emotional processes need to be better separated in order to distinguish and judge the two mainstream theories of psychological political ideology. As described in an earlier chapter, although both the Motivated Social Cognition Theory and the Negativity Bias
Theory often rely on the same evidence and have common ground, they actually make two separate claims about conservatives. The psychological processes of interest in this chapter are *trait* differences between liberals and conservatives, specifically *trait avoidance* and *trait negative affectivity*. These individual dispositions are related to everyday behavior, and both theories make predictions for how these traits ought to be related to political attitudes. Below I describe the predictions each theory would make for these traits.

Recall that the Motivated Social Cognition theory argues that conservatives are more motivated by needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al. 2003). Conservatives are sensitive to avoidance signals, and this sensitivity biases how they view the world, turning them towards a politically conservative belief structure. One way to explore conservative sensitivity to avoidance is by using electroencephalography, measuring electrical signals from the scalp. Cognitive neuroscientists believe that frontal hemispheric asymmetry, a pattern of neural activity in the brain, is correlated with avoidance sensitivity. Thus, *trait avoidance* may be a biomarker of being politically conservative.

The Negativity Bias Theory, on the other hand, actually makes a different argument about trait differences between liberals and conservatives. These scholars argue conservatives are more sensitive to negative emotions than liberals, which causes stronger physiological reactions and focuses greater attention on such stimuli (Hibbing

102

et al. 2014). Thus, the Negativity Bias Theory posits that *negative affectivity* may be a biomarker of being politically conservative.

This distinction between *trait avoidance* and *trait negative affectivity* seems subtle, but it has deep ramifications for what we believe is the psychological origins of political ideology. Trait avoidance is a disposition that moderates tendencies to avoid and withdraw from novel stimuli. Trait avoidance is behavioral, not specifically emotional. A person with strong trait avoidance will have a relatively greater propensity to avoid. This propensity may manifest in myriad ways, such as faster response times to disengage or more awareness or alertness to dangers in the environment. Trait avoidance is a low-level behavioral process, a reaction that insects or small mammals could have. Reacting to smelling a predator or inhibiting the consumption of a piece of bread about to go into your mouth because you see mold are both examples.

This behavior is different than higher order level reasoning. Anticipating the foul mood of a work colleague and deciding to avoid interrupting them for a coffee break may appear to be "behavioral avoidance." In fact, it may indeed be a kind of behavioral avoidance, in some sense, but it is a far more sophisticated kind of cognitive processing than the prior examples. It is less automatic and less instinctual. Only humans have relationships with work colleagues, and only humans are capable of the higher cognition to anticipate and avoid their foul moods. Behavioral avoidance, as I mean it, is a kind of fundamental reaction that all multi-cell organisms possess and utilize hundreds to thousands of times a day.

While trait avoidance is a motivational proclivity, encouraging behavior, trait negative affectivity is an emotional sensitivity, encouraging a *feeling*. Emotions serve the purpose of facilitating behavior, but they are not behavior themselves. Trait negative affectivity is a disposition that enhances tendencies to feel negative affect, giving a tendency to experience a broad range of negative emotions in response to environmental stimuli. A person with greater trait negative affectivity will experience emotions like anger, disgust, and anxiety more regularly than others, and will have relatively more negative affect in response to negative stimuli. Low trait negativity means a person is relatively less affected by negative mood states and negative stimuli. Individuals may have clinical levels of trait negative affectivity, resulting in anxiety, depression, and/or poor self-concept that can interfere with life. Non-clinical populations also have individual variability in negativity sensitivity. For example, two individuals who watch the same sad movie may feel differing degrees of sadness. If one of the individuals felt greater sadness across several sad movies, they may have stronger trait negative affectivity than in the companion. Neither individual is necessarily abnormal in this example. In this case, negative affectivity is simply a characteristic of the person, one of thousands of personality traits.

Trait avoidance and trait negative affectivity overlap in psychological political ideology studies, but they ought to be considered conceptually distinct. Tritt and colleagues argue, for example, "political psychology research may have similarly confounded valence and arousal, leading to the false conclusion that negative valence

per se is associated with conservative political beliefs. The confounded nature of arousal and valence is reflected in Hibbing et al.'s interpretation of experimental, psychophysiological, neurobiological, and personality research" (Tritt et al. 2014, 330). Many negative emotions motivate behavioral avoidance. However, some emotions that are negative are not *avoidant*. Valence (the feeling of positive or negative) and arousal (the feeling, or lack thereof, of energy and attention) have often been confused in emotional studies (see Harmon-Jones et al. 2010; Tritt et al. 2013). Harmon-Jones and colleagues stress the importance of differentiating valence from arousal: "By exploring the cortical regions underlying emotion processes, the research has suggested the importance of delineating emotional experience from emotional expression and *emotional valence* from *motivational intensity and direction*" (Harmon-Jones et al. 2010, 459, my emphasis added).

One important case study might illustrate this point. Consider the emotion of anger, an important discrete emotion that both theories fail to consider fully. If considered more fully, in fact, the two literatures would have opposing hypotheses about political conservatives. Anger is a negative emotional state that motivates approach behavior like yelling, hostility, confrontation, and even violence (Ekman & Friesen 1975; Plutchik 1980; Berkowitz 1993; Blanchard & Blanchard 1984; Lagerspetz 1969). Harmon-Jones and colleagues believe "the valence of the emotion may be separable from the motivational direction of the emotion, so that negatively valenced emotions such as anger can be approach motivating" (Harmon-Jones et al. 2010, 459). The Motivated Social Cognition theory predicts conservatives to be *avoidant*, and therefore would predict conservatives to have less trait anger than liberals because anger is related to *approach*. In contrast, the Negativity Bias Theory posits that conservatives are sensitive to *trait negative affectivity*. As anger is a negative emotion, the theory predicts that conservatives have more trait anger than liberals. The theories are at crossroads on anger, and this example illuminates the motivation of this study to examine trait avoidance and trait negativity affectivity individually.

Interestingly, both theories mostly avoid discussions of anger, not integrating it directly with their respective theories, nor discussing anger as a division point between Motivated Social Cognition Theory and Negativity Bias Theory. In fairness, of course, studying psychological anger as a discrete emotion is very new to political science (but see Ryan 2012). In fact, the study of discrete emotions in political science is itself fairly new (but see Hatemi et al. 2013; and Clifford and Wendell 2015). Hibbing and colleagues (2014) include only a glancing mention of anger, in passing, and without data or consequence to their argument. They write, "when 'emotionally ambiguous' faces are shown to research participants, individuals on the political right are more likely to report that the face is expressing a threatening or dominant emotion, such as anger." The word anger is otherwise absent from the rest of the article. Consequently, anger is not given full consideration in the theory, or distinguished from other negative emotions. Jost et al. 2003 give a similar treatment to anger. Anger is mentioned in a subheading (361), but merely lumped alongside fear and threat. Grouping anger with fear and threat is problematic because while all are negative emotions, fear triggers behavioral *avoidance* while anger triggers behavioral *approach*. These discrete emotions have the same negative valence, but they motivate completely different behavior. Furthermore, the evidence marshaled by Jost and colleagues is exclusively about reacting to fear and threat (361). There is no direct reference to anger as anything other than a synonym of feeling threat, and no data exploring the relationship of anger to ideology whatsoever. Also, arguably, the link between anger and ideology that they explore is in the specific context to Right Wing Authoritarianism and parenting styles (and, so, not emotional anger, nor broad political conservatism) (Jost et. al 2003, 347; see also Jost et al. 2009).

The discussion above on anger risks being taken as simply a cheap shot against two big targets—is it not unfair to take criticism to these seminal works simply because they have left work to be done? Let me be especially clear about why this discussion of anger matters to the present work. Anger specifically illustrates the potential problems with conflating emotional valence for behavioral motivations in psychological work. Anger has a negative valence, and yet, also encourages behavioral approach, unlike most other negative emotions. But, more importantly, the discussion of anger is *also* serving as an illustration of the broad consequences of being too vague about the underlying psychological factors that contribute to political ideology.

Right now, as paradigmatic and ground-breaking as they are, the theories rely on much of the same literature but Motivated Social Cognition interprets conservatives as avoidant, while the Negativity Bias Theory interprets conservatives as sensitive to negativity. Neither theory has taken up the challenge of sorting out whether these conservatism effects are (a) behavioral avoidance, (b) negative affectivity, (c) both, or (d) neither. It is time to dig in deeper into these parallel claims. I argue that trait avoidance and trait negative affectivity are two different psychological dispositions, and this chapter is an opportunity to evaluate both theories on their own appropriate claims.

Thus, it is clear that investigating the theories based on their claims of conservative dispositions requires two different (but related) investigations. I argue that testing baseline dispositional differences is an important way to draw contrasts between the two theories, in an attempt to understand better these psychological differences between liberals and conservatives. Motivated Social Cognition needs to test *trait avoidance*, as it posits that conservatives have more trait avoidance. Negativity Bias Theory needs to test *trait negative affectivity*, as it posits that conservatives have more sensitivity to negative affect. This chapter tests these theories in two ways: first, we examine a dispositional difference in resting brain activity that has been linked to trait avoidance, and second, we compare dispositional differences in trait-based negative affectivity.

### Frontal Asymmetry and Trait Avoidance

Cognitive neuroscientists have used trait frontal hemispheric asymmetry for more than 25 years to investigate approach and avoidance sensitivities in human beings. Today, some of the seminal papers on trait frontal hemispheric asymmetry from Davidson and colleagues have over 900 citations (Sutton and Davidson 1997, Davidson 1992). Coan and Allen, writing in 2004, report than the relationship between cortical asymmetries and emotion had already been established by over 70 studies (2004, 7). In their metareview, they establish that resting levels of neural activity in the prefrontal cortex appear to be correlated with trait predispositions towards emotions and behavioral motivations. "Findings from numerous studies reflect an emerging consensus that relatively greater trait left frontal activity is associated with trait tendencies toward a general appetitive, approach, or behavioral activation motivational system, and that relatively greater trait right frontal activity is associated with trait tendencies toward a general avoidance or withdrawal system" (Coan and Allen 2004, 11). Below I describe these findings, and then turn towards how it may be useful in evaluating the Motivated Social Cognition Theory and the Negativity Bias Theory.

Very early in the study of hemispheric asymmetry—indeed, of all neuroscience researchers connected hemispheric activity patterns with emotions. Over 70 years ago, World War I soldiers with damaged right or left anterior cortices were discovered to have differing experiences with positive and negative affect (Goldstein 1939). Subjects with lesions to the left frontal region (leaving the left side malfunctioning and the right side intact) were more likely to exhibit depression symptoms (Black 1975, Gainotti 1972; Robinson and Price 1982).

These earlier efforts with clinical populations eventually led to research on normal populations, too. Measuring electroencephalography, Tomarken et al. (1990),

found that subjects with greater resting right frontal activity responded with more intense negative affect to negatively valenced film clips, particularly those involving fear or threat. Tomarken and colleagues later proposed that trait positive affect is associated with greater left than right frontal cortical activity, whereas trait negative affect is associated with greater right than left frontal activity (Tomarken et al., 1992). Another study using film clips similarly revealed that having more right frontal activity led to more intense reports of negative affect, and having more left frontal activity led to more intense reports of positive affect (Wheeler et al. 1993).

The relationship between hemispheric asymmetry and emotional sensitivity had early support, but is now challenged by contemporary research. Indeed, these early findings were real, but new research clarified that emotional affect may have been a confound. Davidson and colleagues (1993, 1998a, and 1998b) pushed the study of hemispheric asymmetry away from positive and negative affect, instead suggesting that hemispheric asymmetry corresponded with an approach-avoidance system, a hallmark of Gray's theories of BIS/BAS behavioral motivation. Several follow-up studies by Davidson and colleagues confirmed the relationship (e.g. Reuter-Lorenz and Davidson, 1981; Sutton and Davidson 1997; Davidson 2004). Harmon-Jones and colleagues also found support for the approach-avoidance theory, noting that "past research had essentially confounded emotional valence with motivational direction, and researchers were claiming that relatively greater left than right frontal cortical activity reflected greater approach motivation and positive affect, whereas relatively greater right than left frontal cortical activity reflected greater withdrawal motivation and negative affect" (Harmon-Jones et al. 2010, 454-455). Davidson's findings have been empirically verified by a number of different research groups (e.g., Carver and White 1994; Coan and Allen 2003, 2004; Harmon-Jones and Allen 1997, 1998; Amodio et al. 2008).

The consensus in this literature is that relatively greater right frontal activity is associated with tendencies toward behavioral avoidance, instead of emotional affect (Coan and Allen 2003a; Coan and Allen 2003b; Davidson 1993; Harmon-Jones and Allen 1997, Sutton and Davidson 1997, Davidson 1998a,b; Harmon-Jones et al. 2010). Other researchers have confirmed this relationship. For example, Field et al. (1995) and Fox et al. (1996) found evidence that children with greater right frontal activity at rest were more inhibited socially, and scored lower on social competency. Schmidt and Fox (1994) found a relationship between low adult sociability and relatively greater right frontal activity. Schmidt et al. 1999 found shyness had greater right frontal activation.

All of these studies from developmental psychology researchers suggest a relationship between avoidance and right frontal activity. Davidson's approach/withdrawal motivational model of emotion also proposes that left frontal activity (either as state or as trait) indicates a propensity to approach or engage a stimulus, while relatively greater right frontal activity indicates a propensity to withdraw or disengage from a stimulus. Yet, as described above, testing the avoidance motivations of conservatives would appear to be a significant part of the larger project of the Motivated Social Cognition Theory. (Also, it is worth mentioning that the Negativity Bias Theory may also be interested in these results, given that while emotion is technically a confound in these hemispheric studies, it can still be instructive because most avoidance motivations are, in fact, triggered by negative emotions.) Consequently, frontal EEG asymmetry promises to be a useful tool for studying avoidance sensitivity in political liberals and conservatives. To date, no study has looked at resting hemispheric data to evaluate whether liberals and conservatives have more approach or avoidance sensitivities. This chapter thus fills a lacuna in the psychological study of political ideology.

## **PANAS-X and Sensitivity to Negative Emotion**

To examine trait differences in how liberals and conservatives experience emotion, we will need to use a tool designed specifically to evaluate trait (not state) emotions. Recall that this chapter focuses on trait dispositions, or the characteristic propensity to feel an emotion independent of context. We are interested in trait negative affectivity, not the effect of putting subjects in a negative state condition. One method of operationalizing trait negative affectivity is by having participants self-report how often they experience these emotions in their last few weeks. The Positive and Negative Affect Schedule (PANAS) is a scale developed by Watson and colleagues (Watson et al. 1988; updated to PANAS-X in Watson and Clark 1994) to measure positive and negative affect in contextual (state) and characteristic (trait) ways. The scale is extremely popular in psychology and has more than 17,000 citations. As a trait measure, PANAS-X has been shown to be significantly correlated with corresponding judgments made by well-acquainted peers (Watson and Clark 1994, 19). Self-reporting a negativity bias, for example, was correlated with peers claiming that you were sensitive to negativity. Additionally, test-retest scores were strongly consistent, and PANAS-X scores were significantly correlated to weekly mood ratings in a sample of 239 undergrads studied for a minimum of seven weeks (Watson and Clark 1994, 19-20).

Before continuing, what exactly is positive and negative affect? Positive affect represents the extent to which an individual feels enthusiastic, active, and alert (Watson et al. 1988). Higher positive affect is a state of full energy, concentration, and pleasing engagement, whereas low positive affect is characterized by lack of energy and sadness (Watson et al. 1988). Negative affect is the extent to which a person feels general distress and unpleasant engagement. It includes mood states such as anger, contempt, disgust, guilt, fear, and nervousness, while low negative affect is a state of benign calmness (Watson et al. 1988). It is worthwhile to note here that positive affect and negative affect are theoretically (with empirical justification) thought of as separate emotional factors or dimensions (Watson and Clark 1994). An individual could have a strong trait of positive affect and a strong trait of negative affect, together. This individual would be a (relative) roller-coaster of emotion—higher highs and lower lows—compared to someone with low affect in both dimensions. For example, the Negativity Bias Theory states that conservatives are more trait negative. This does not axiomatically mean the theory would also posit that conservatives are less trait positive. The dimensions are largely independent from each other (Watson et al. 1988, 1064; Watson and Clark 1994, 1-3)

As the Negativity Bias Theory posits that conservatives are more sensitive to negative affect, it can to be tested via the PANAS-X scales. Although the Motivated Social Cognition Theory ultimately makes a different claim than conservatives have more negative affect, the results of this study would still prove useful to that study. Affect and behavioral motivations have much in common, even though they are commonly conflated.

## Hypotheses

Thus, the two major theories of psychological political ideology have predictions for trait dispositions related to avoidance and negativity bias (see Table 4, below). Concerning frontal hemispheric asymmetry and *trait avoidance*, the Motivated Social Cognition Theory posits that conservatives should have greater right-side asymmetry, reflecting a greater sensitivity to avoidance compared with liberals (**H1**). The Negativity Bias Theory agrees that conservatives should have more avoidance sensitivity, but as I have argued, this is technically confounding avoidance sensitivity for negativity bias. Consequently, a better test of the negativity bias would be measuring *trait negative affectivity*. The Negativity Bias Theory predicts that conservatives should have more trait negative affectivity than liberals (**H2**). Additionally, in terms of the discrete emotion subscales, the Negativity Bias Theory predicts that conservatives should have more trait fear (**H3**) and trait sadness (**H4**) than liberals. Again, the Motivated Social Cognition Theory does not necessarily dispute **H2**, **H3**, *or* **H4**. After all, there is substantial overlap between trait avoidance and trait negative affectivity. However, interestingly, we can use trait hostility (anger) as a differentiator between the theories. Anger is both affectively *negative* but motivationally *approaching*, where the other emotions are affectively *negative* and motivationally *avoiding*. Motivated Social Cognition Theory predicts conservatives should have *less* trait hostility (anger) than liberals, while Negativity Bias Theory predicts conservatives should have *more* trait hostility (anger) (**H5**).

Both theories remain agnostic about positive emotion scales. As mentioned above, positive and negative affect sensitivity are independent from each other (positive affect is not simply the absence of negative affect, but instead is its own emotional dimension). Although Hibbing and colleagues (Dodd et al. 2012) find that liberals have increased sensitivity to *positivity* compared to conservatives, this finding is not emphasized or explored in Hibbing et al. 2014, the foundational paper of the Negativity Bias Theory of political ideology, and so is not emphasized here, either.

115

Table 4. Disposition Hypotheses.

		Motivated Social	Negativity
		Cognition	Bias Theory
Trait Avoida	nce		
H1. Conserv	atives have more trait avoidance	Yes	Yes
-			
Trait Negativ	e Affectivity		
H2. Conserv	atives have more trait negative affectivity	Yes	Yes
H3. Conserv	atives have more trait fear	Yes	Yes
H4. Conserv	atives have more trait sadness	Yes	Yes
H5. Conserv	atives have more trait hostility (anger)	No, less	Yes

# Method

### Participants

466 adults employed by a large university in the Midwest completed a screening survey of their political attitudes and personality dispositions. Details about this survey are in Chapter Four of the dissertation. Four months later, a subsample of 51 was selected to perform laboratory behavioral experiments. As the original pool was substantially skewed towards liberals, this sample was screened for political attitudes in order to promote political diversity in the sample. The screening was simply a random selection of 20 liberals, 15 moderates, and 20 conservatives, determined by a combination of their self-reported ideology, and the aggregation of their political attitudes on various issues (the Wilson-Patterson scale, detailed in full in Chapter Four). If an invitation was declined or ignored, a new random participant of their group would receive an invitation. Over the next six months, 20 liberals, 14 moderates, and 17 conservatives agreed to the additional testing session and came into the laboratory. (More detailed political characteristics of this sample are of direct interest to this chapter, and will be described in additional detail below.)

Fifty-one adults (age 24-52 years, *mean* = 30.7, *sd* = 5.8) served as participants (32 female, 19 male). Participants self-described as right-handed (49 selected "somewhat" or "strongly" right-handed). Although it is speculated that handedness may be related to hemispheric asymmetry patterns, and most research consequently selects right-handed participants only, differences in handedness have no impact on the findings of this chapter. The testing session averaged less than two hours and 30 minutes, including breaks and capping preparation (described below). Participants completed four tasks during the session, and had short breaks of 5 minutes between tasks. Participants were permitted to stand up and/or leave the testing room, but were encouraged to avoid doing so because of the tedious capping process. As it happens, no participants left or required to be recapped during their testing session. Participants were compensated at a rate of \$20/hour, with a minimum payment of \$40 and a maximum of \$50 for the session.

# Procedure

After arriving at the laboratory, participants were briefed on the tasks, and asked to sign consent forms. Electroencephalography requires wearing a tight fitting cap that includes a set of sensor nodes for reading electrical signals from the scalp. The capping process requires between 20-40 minutes of preparation time, as electrode nodes are first gelled with a saline solution to increase conductivity, then snapped into the cap one at a time, and then finally the readings system is tested. The electrodes do not break the skin or provide any discomfort to the participants. While being capped, but before the experiments began, participants completed a brief questionnaire about personality as well as the PANAS-X trait and state emotional batteries (Watson and Clark 1994). Participants first completed three tasks and had several breaks for approximately 110 minutes. Then, they began the resting electroencephalography task that is the subject of this chapter. Resting electroencephalography, a reading of electrical signals from the scalp while the participant was at rest, was recorded while participants sat quietly in a sound-attenuated room for an 8-minute resting period (alternating one minute blocks of eyes-open and eyes-closed in a counterbalanced pattern, as is the convention). The activity lasted about 10 minutes in total, and it was the final activity completed by participants before they were dismissed from the session.

### Measuring Alpha Asymmetry

Resting alpha waves were collected in a 64 channel system designed by Biosemi. Alpha waves are one variety of brain waves that can be detected by electroencephalography. Alpha waves appear to originate from the occipital lobe during an awake, but resting, state. According to Coan and Allen, "evidence suggests that activity within the alpha range (typically 8–13 Hz) may be inversely related to underlying cortical processing, since decreases in alpha tend to be observed when underlying cortical systems engage in active processing" (Coan and Allen 2004, 9). The detection of *more* alpha waves, thus, represents *less* cortical processing. Measuring alpha is therefore an indirect way to measure cortical processing. As our goal is to assess the degree to which one hemisphere may be more active while at rest, like all other frontal asymmetry researchers, we need to measure alpha waves in order to infer cortical activity.

Alpha was collected in the frequency range between 8-13 Hz, in 0.5 Hz bands. The raw data was collected by the Biosemi system (compiled in a BDF file format for each participant) and manually checked for eye movement artifacts using BESA Research, Windows-based software designed to analyze digitally recorded EEG. Electroencephalography is hyper sensitive to muscle movements, and the slightest twitch of the eye can result in a noticeable data artifact which needs to be removed from analysis (as individuals have different propensities to blink). In order to reduce volume conduction contributions to EEG, data were transformed to a current source density (CSD) Laplacian 27- channel virtual montage (Allen & Reznick, 2015; Stewart et al., 2014). The application of a CSD montage has been recommended to yield a stable, localized estimate of frontal alpha (Stewart et al., 2014).

One challenge for comparing results between labs is keeping consistent electrode placement, and only careful work in the lab with measuring tape can minimize these differences between subjects. Another complication is that different EEG systems today use varying electrode node schemes over time, with most systems today having 32, 64, or 128 channels. Consequently, researchers use *virtual channels* to allow comparison of results between labs. Whether EEG signals are measured with a 32, 64, or 128 channel system, software can be used to convert these measurements to a common cranial mapping system called the 10-20 system. Thus, our 64 channels of recorded data were converted to virtual channels corresponding to the traditional 10-20 system, thus allowing comparison to other asymmetry literature. 10-20 refers to the fact that the actual distances between adjacent electrodes on the scalp are either 10% or 20% of the surface area, creating a spreading pattern where electrodes are equidistant from each other.



Note: Colored nodes are the electrodes used to compute frontal cortical hemispheric asymmetry

As observed in Figure 10, each site has a letter to identify the lobe (<u>F</u>rontal, <u>T</u>emporal, <u>P</u>arietal, or <u>O</u>ccipital) or to signify that it is <u>C</u>entrally located. Numbers are used with the letters to designate the hemisphere location, with even numbers on the right and odd numbers on the left. A "z" (zero) is used for an electrode on the midline. The codes "A", "Pg" and "Fp" identify the earlobes, nasopharyngeal and frontal polar sites respectively.

The first step of capping correctly is to note two anatomical landmarks: first, the nasion which is the distinctly depressed area between the eyes, just above the bridge of the nose; and second, the inion, which is the lowest point of the skull from the back of the head and is normally indicated by a bump. From these landmarks, researchers can determine the sizing of the cap, and the placement of the cap on the head, to ensure that electrical nodes are consistently arranged from subject to subject.

Electroencephalography data needs to be referenced (all signals recorded are necessarily relative, and so there is need for a base reference). In this system, all sites were referenced to the mastoids. To filter electrical interference from the signal, a band filters of 0.01 Hz and 60 Hz was applied, as is standard convention. Cortical hemispheric asymmetry is measured with frontal electrodes (placement of these electrodes are noted with color). Following conversion to the virtual channels, amplitudes were natural logged by site, and then asymmetry scores were calculated the conventional way (see Coan and Allen 2004), by subtracting the left score from the right score at each frontal node pairing (FP2 – FP1, F4 – F3, F10 – F9, and F8 – F7). The original idea for creating a difference score comes from the theory that the two hemispheres inhibit each other, so a greater *difference* in resting state scores suggests one hemisphere is more dominant (and there are approach-avoidance consequences for this dominance). The subtraction also takes into account individual differences in skull mass or other physical factors,

creating a metric that accounts for the *relative* asymmetry of that individual. After the creation of the node subtractions, all four were averaged into an aggregate frontal asymmetry subtraction score.

Table 5. Frontal Asymmetry Score Interpretation.

Frontal Asymmetry Score	Alpha Waves	Neural Activity	<b>Behavioral Motivation Implication</b>
Negative	More alpha on left side	Right hemisphere more active	Avoidance Sensitivity
Zero	Equal alpha on both sides	Equally active sides	No relationship
Positive	More alpha on right side	Left hemisphere more active	Approach Sensitivity

For the remainder of the chapter, we will use the frontal asymmetry score to measure and discuss the findings of this experiment. Table 5 summarizes interpretation. The frontal asymmetry score represents the relative activity of the right and left hemispheres, with a score of zero meaning symmetrical activity. In interpreting this scale, a positive frontal asymmetry score means more alpha waves on the rightside, more left frontal activity, and increased approach sensitivity. A negative frontal asymmetry score means relatively more alpha waves on the left-side, more right frontal activity, and increased avoidance sensitivity.

### Measuring Trait Emotions

The PANAS-X revised scale was administered to all participants in order to assess the degree to which they are sensitive to emotions. PANAS-X questions allow emotional sensitivity to be assessed dimensionally—positive affect and negative affect. Each participant uses a five point scale to self-report how much they have felt a discrete emotion over the past couple of weeks (trait), or how much they feel it now, in the moment (state). A total of 30 discrete emotion items are included (see appendix for full list).

Positive affect and negative affect are themselves comprised of a factor

structure of discrete emotions. In their creation and analysis of the PANAS-X scale,

Watson and Clark (1994) identify a number of subscales for each affective dimension.

Positive Affect is comprised of Joviality, Self-Assurance, and Attentiveness. Negative

affect includes Fear, Hostility, and Sadness. A summary of the two dimension scales, the

six discrete emotion subscales, and the individual emotion items that comprise each

scale is included in Table 6.

Table 6. PANAS-X Scale Compositions

Gene	ral Dimensional Sca	les
	Negative Affect	afraid, scared, nervous, jittery, irritable, hostile, guilty, ashamed, upset, distressed
	Positive Affect	active, alert, attentive, determined, enthusiastic, excited, inspired, interested, proud, strong
Basic	Negative Emotion	Scales
	Fear	afraid, scared, frightened, nervous, jittery, shaky
	Hostility	angry, hostile, irritable, scornful, disgusted, loathing
	Sadness	sad, blue, downhearted, alone, lonely
Basic	Positive Emotion S	cales
	Joviality	happy, joyful, delighted, cheerful, excited, enthusiastic, lively, energetic

```
Self-assurance
proud, strong, confident, bold, daring, fearless

Attentiveness
alert, attentive, concentrating, determined
```

Note. Adapted from Watson and Clark, 1994, "The PANAS-X: Manual for the Positive and Negative Affect Schedule - Expanded Form"

As can be observed in Table 6, discrete emotions are aggregated into two

general dimensional scales, negative affect and positive affect. For example, fear,

hostility, and sadness are all negatively valenced discrete emotions, while joviality, self-

assurance, and attentiveness are all positively valenced discrete emotions. An individual

with greater trait negative affectivity would be expected to react more strongly to negative emotions, and would do so independently from state-based contextual factors, such that would *trigger* an emotional response. If two people both had an emotional response triggered by the environment, the one with greater trait negative affectivity would be expected to react more strongly to that trigger, all else being equal. PANAS-X also allows the creation of several subscales of interest, as well. As noted above, discrete emotions like fear, hostility (anger), and sadness are commonly invoked by the Negativity Bias Theory and other emotion researchers working in political psychology. *Measuring Ideology* 

Political ideology questions were asked to all participants several weeks prior to the laboratory experiment. Below are the five ways ideology was measured in this sample, and the samples distributions on those scales. Cronbach's alphas reported in subsequent paragraphs are from the full survey sample of 466 participants, as alphas from the smaller sample of 51 include more random variability and are not as conceptually useful. That said, alphas from the smaller sample did not differ in a meaningful way from the numbers reported below. Other details about the scale construction and measure validity are described in Chapter Four. Table 7. Descriptive Statistics of Lab Participants.

	n	mean	sd	median	min	max	range	skew	kurtosis	se
self-report ideology	49	3.31	1.65	3	1	6	5	0.09	-1.4	0.24
Wilson-Patterson (Full)	51	6.94	3.82	7	0	14	14	-0.07	-1	0.54
Wilson-Patterson (Economic Conservative)	51	2.04	1.33	2	0	4	4	-0.12	-1.14	0.19
Wilson-Patterson Social Conservative)	51	2.41	2.12	2	0	7	7	0.43	-1.06	0.3
World Values Survey Economic Justice Scale	51	21.16	7.4	24	0	34	34	-0.46	-0.27	1.04

Self-reported ideology is the conventional seven-point scale used by the American National Election Studies, spanning "Very liberal" (1) to "Very conservative" (7) with "Moderate" (4) being the middle anchor. Liberals range from 1 to 3, moderates are 4s, and conservatives range from 5 to 7. Although some literature has pointed to interesting alternatives for self-reports (Wood and Oliver 2012), this is still the dominant way of measuring ideology in a survey. Also, this scale (and a 10 point variant of it) is used almost exclusively by psychologists doing work in the Motivated Social Cognition Theory area (e.g. Jost et al. 2003, Jost and Amodio 2012).

The Wilson-Patterson scale of political attitudes (Wilson 1968; updated by Smith et al. 2011) is a battery of twenty issue attitude items, which are additively combined into a general conservatism scale ( $\alpha$  = .80). As ideology is sometimes conceptualized as multidimensional, I also created social conservatism ( $\alpha$  = .76) and economic conservatism ( $\alpha$  = .61) subscales based off of the Wilson-Patterson scale detailed above. A full description of how these subscales were constructed is available in Chapter Four. In short, however, these two subscales are based on theoretical grouping, and based on general multidimensional work (e.g. Treier and Hillygus 2009). Social conservatism includes school prayer, legal pornography, illegal immigration, death penalty, the Patriot Act, biblical truth, gay marriage, and abortion. Economic conservatism includes welfare spending, tax cuts, small government, and foreign aid. More research needs to be done on the psychological underpinnings of *multidimensional* ideology (Hibbing et al. 2014), and some evidence points to the idea that ideology is not a singular concept (see Chapter Two for more details). It will be useful here to see how social and economic conservatism might differ, and this is an important way that political scientists can improve on the work of psychologists in the study of the psychology of political ideology.

Finally, I used the World Values Survey asks four Likert-style questions about economic justice. (Exact question wording is available in the appendix.) These items have been combined into the World Values Survey Economic Justice battery, which measure attitudes about income redistribution and economic inequality ( $\alpha = .70$ ).

## Results

### Table 8. Correlation Matrix of Emotional Measures.

		Ideology Measures						
		self reported	1550 Attitudes	Economic	Socialissues	Economic lustice		
	Trait Avoidance							
	Frontal Asymmetry	0.23	0.355*	0.235‡	0.346*	0.1		
nits	Trait Negative Affect	-0.08	-0.157	-0.149	-0.08	-0.294*		
2	Fear	-0.185	-0.161	-0.239‡	-0.028	-0.356*		
ical	Sadness	0.003	-0.158	-0.067	-0.131	-0.242‡		
olog	Hostility	0.001	-0.034	-0.107	-0.025	-0.072		
Psych	Trait Positive Affect	0.1	0.206	0.197	0.15	0.175		
	Joviality	0.02	0.141	0.109	0.122	0.131		
	Self-assurance	0.139	0.253‡	0.303*	0.176	0.254‡		
	Attentiveness	0.158	0.187	0.17	0.108	0.108		

\* p < 0.05; ‡ p < 0.10. Correlation tests are two-tailed.

Note on ideology scales: higher values reflect more conservatism. Self-reported is the ANES 7 point scale. Issue attitudes is the full Wilson-Patterson scale. Economic and Social issues are Wilson-Patterson subscales. Economic Justice comes from the World Values Survey.

The Motivated Social Cognition Theory predicted that conservatives would have more trait avoidance compared to liberals (**H1**), which is operationalized in this study as a lower frontal asymmetry score. However, I find no significant relationship between frontal asymmetry and self-identified political ideology, and it trends in the opposite direction than predicted (r = 0.23, p = 0.11). I also find that the Wilson-Patterson conservatism scale is correlated *positively* with the frontal asymmetry score (r = 0.36, p= 0.01), exactly the opposite of the expectations. This finding also holds for the W-P social conservatism scale (r = 0.35, p = 0.01), and it trends in the same pattern for the economic issue subscale (r = 0.24, p = 0.10). The economic justice subscale of the World Values Survey, which focuses on liberal-conservative beliefs about economic inequality, is not correlated with frontal asymmetry (p = 0.49). Whether ideology is self-reported or survey-determined, there is no evidence that conservatives have more trait avoidance based on frontal asymmetry measures. If anything, these results suggest *liberals* have the neural correlates for having more trait avoidance.

What about trait sensitivities to negative emotions? No statistically significant relationship was found between self-identified conservatism and trait negative affectivity (H2) (p = 0.59). No statistically significant relationship was found between the Wilson-Patterson ideology scale (p = 0.27), the W-P social conservatism subscale (p = 0.58), or the W-P economic conservatism subscale (p = 0.3). The World Values Survey Economic Justice scale does have a surprise negative correlation to trait negative affectivity (r = -0.36, p = 0.01), which suggests that more trait negative affectivity is correlated with political liberalism (not with conservatism, as was expected).

Trait fear appears mostly unrelated to political ideology, with a trending finding on economic conservatism, and a statistically significant relationship on the World Values Scale, where once again the relationship directions are *contrary* to theoretical expectations (**H3**). Self-reported ideology (p = 0.20), W-P scale (p = 0.26), or W-P social conservatism (p = 0.84) yield no relationship. Again, the Wilson-Patterson subscale for economic conservatism trends towards having a relationship to trait fear, but in precisely the opposite direction than expected (r = -0.24, p = 0.09), with it being associated with political liberalism. Economic conservatives in this sample may have moderately *less* trait fear sensitivity.

Trait sadness (**H4**) and trait hostility (**H5**) have essentially no relationship to any measurement of political ideology in this sample (*p* values range from 0.27 to 0.95 for both trait sadness and trait hostility, see Table 8). Early in this chapter, I mentioned that trait hostility (anger) may prove to be an interesting dividing point between the Motivated Social Cognition Theory and the Negativity Bias Theory, but as it turns out, both theories fail their hypotheses here.

Although both theories are agnostic about the relationship of political ideology to positive affect, we have the data to explore. First, recall that Watson and Clark 1994 posited trait negative and positive affectivities ought to be considered separate dimensions (and not two sides of the same construct). In this data, I find that the two characteristics are only weakly negatively correlated (r = -0.26, p = 0.06), confirming their research. Second, in terms of the relationship of trait positive affectivity to political ideology, I find rather strong evidence of no relationship whatsoever (see Table 8). One correlation manages to catch, as economic conservatives have more trait selfassurance (r = 0.30, p = 0.03). We should be careful about extrapolating too much on that finding, given that it was precisely one significant result of twenty tests (exactly 5% of the tests) of trait positive affect. Self-assurance has two trending relationships, both in the opposite direction to expectations.

### Discussion

This chapter focuses on ideological differences in trait avoidance and trait negative affectivity. These individual dispositions are related to everyday behavior, used by human beings from dozens to hundreds of times a day. We investigate the claims of the Motivated Social Cognition Theory and the Negativity Bias Theory, being especially sensitive to the idea that the theories subtly posit different psychological undercurrents for political ideology (albeit, processes that are quite related).

Recall the Motivated Social Cognition theory argues that conservatives are more motivated by needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al. 2003). Conservatives are thought to be sensitive to avoidance signals, biasing their beliefs towards political conservatism. I use frontal hemispheric asymmetry to test the idea that trait avoidance may be a biomarker of being politically conservative.

The Negativity Bias Theory, on the other hand, argues conservatives are more sensitive to negative emotions, causing stronger physiological reactions to negativity (Hibbing et al. 2014). I used a battery of survey questions that tap into emotions to test the idea that trait negative affectivity may be a biomarker of being politically conservative.

The results are surprising in every possible way. The data show there is virtually no evidence that liberals and conservatives hold trait differences in trait avoidance or sensitivity to negative emotion. In fact, if anything, liberals in this sample appear to be more sensitive to avoidance, which stands out from the vast majority of political psychology work on this subject. Where I had hoped to contribute by parsing out two different psychological processes (behavioral motivations like avoidance from negative affect) in order to clarify our two main theories, instead I fail to find evidence that supports either theory.

Overall, ideology was measured in five ways, and in no form of measurement is there a relationship between trait emotional sensitivity and ideology. These results urge caution moving forward for both psychological theories of political ideology, and begs further work to establish where the psychological differences noted by Jost, Hibbing, and colleagues are emerging. Those results have strong support over dozens of papers, yet, my work here does not support their claims. The next chapter continues the investigation of how avoidance sensitivity and negativity bias may be different between liberals and conservatives.

### **CHAPTER SIX**

# **GO/NO-GO BEHAVIORAL INHIBITION ANALYSIS**

This chapter expands our investigation of dispositional differences between liberals and conservatives by looking at conflict monitoring in the brain. Conflict monitoring is the cognitive process that helps resolve issues between goal directed behavior and our ever changing environment, allowing us to detect potential conflicts and facilitate behavior to resolve detected conflicts. Conflict monitoring is part of a behavioral surveillance system, with the central purpose of resolving a conflict between desired outcomes and a habituated response.

Conflicts between goals and ongoing behavior occur regularly in human experience. For one silly—but instructive—example, consider the children's game, Red-Light-Green-Light. In the game, one child plays as the Stop-light and the rest are at a starting line some distance away. The game begins when the Stop-light yells, "Green light!" At this cue, the children at the starting line race towards the Stop-light at high speed. However, at any time, the Stop-may call out, "Red light!" which forces the children to immediately halt moving. Children who cannot successfully stop are removed from that round of the game. As the game continues, the Stop-light alternates between calling for green light and red light in an attempt to get the dashing kids to make errors and drop out of the game. The first child to tag the Stop-light without violating the red light signal is the winner. The Stop-light wins if they can successfully force errors in all the runners, leaving no runners remaining on the field.

What makes Red-Light-Green-Light difficult for children (and their 32 year old uncle) is that the players have a common goal that is difficult to achieve. They want to be the first to touch the Stop-light, which is accomplished by running more quickly than the other players. However, the runners must simultaneously monitor changing signals in the environment to stay "safe" in the game. Failure to stop immediately after the red light signal results in disqualification. A runner in this game needs to be goal-directed, but also needs to remain constantly aware of changing signals in the environment. And, if your niece really wanted to see her uncle out of the game, she could yell out, "Green light<u>ning</u>!" which falsely impersonates a green light signal to begin running again. (Do not underestimate the strategic shrewdness of your nieces.) To succeed in this game, you need constant surveillance paired with fast footwork. The challenge of the game is in successfully reaching your goal rapidly while being alert to unexpected signals that threaten immediate failure.

Red-Light-Green-Light is a metaphor for conflict monitoring in animal behavior. We are constantly engaged in small goal-oriented behavior, while also simultaneously being constantly vigilant for new signals in the environment. Conflict monitoring is a cognitive process that allows humans to detect potential conflicts between behavior and goals as well as instigate new behavior to resolve the conflict. Our conflict monitoring processes are constantly vigilant, and may trigger dozens of times per day, at many different scales. You reach down for an apple at the grocery store, and place it in your basket. You repeat this twice more. However, when you reach down for the fourth apple, your eyes catch the sheen of dusty green-brown mold. Your goal to place this apple in your basket was interrupted by a new signal in the environment. Conflict monitoring is the process that halts your habituated behavior in response to this new information. Other day-to-day examples are numerous. While briskly walking through a cafeteria towards a vender, you suddenly need to halt because another person walks into your path without seeing you. You might be taking notes during a phone call and realize that the pen you are using has run out of ink, requiring you to shift attention from the phone call to the drawer under your desk for the backup pen. Your conflict monitoring processing is what helps you identify when a shift in attention and behavior may be required, especially in circumstances when the ongoing behavior is prepotent, habitual, or intuitive. We need our conflict monitoring process the most when we are not concentrating or focusing on what we are doing, and the behavior is routine.

Conflict monitoring is broadly categorized by researchers as part of the behavioral inhibition system, which has been a theme of this dissertation. As a reminder from earlier chapters, behavioral inhibition is the system that triggers motivation to halt ongoing behavior in response to the environment, especially concerning negative stimuli and events that cause punishment, frustration, and anxiety (Gray 1982; Gray & McNaughton 2000). The behavioral inhibition system is a broad psychological construct that involves numerous other cognitive processes, of which conflict monitoring is one example.

This chapter follows recent cognitive psychology work by looking at conflict monitoring in the brain, examining behavioral and neural data responses on a Go/No-Go task. I record electroencephalography to measure how the brain reacts when it detects a conflict between goal and behavior. Our key question for this chapter is: are there individual differences between liberals and conservatives in conflict monitoring processing?

As we will see, only limited attention has been given to the question of whether liberals and conservatives would be expected to have differences in conflict monitoring processes. Much like personality differences, we might expect some individuals to have a stronger conflict monitoring response than others, allowing them to make fewer response errors to simple behavioral tasks. This chapter is a direct replication of a study by David Amodio, John Jost, Sarah Master, and Cindy Yee (2007), which hypothesizes that because liberals are more cognitively flexible, they ought to have stronger conflict monitoring systems. This study, cited more than 350 times, is a seminal contribution to the area of inquiry sometimes called *political neuroscience* (Jost and Amodio 2012), and seeks to bolster and extend Jost's extremely popular theory of conservatism as motivated social cognition. Replication of this study is important because the paper itself is important to the project of biology and politics. When political scientists and political psychologists defend the utility of neuroscience for studying political attitudes, the Amodio et al. 2007 paper is always cited, and it is sometimes the only such paper cited in a review paper.

Prior chapters in this dissertation have raised more questions about the psychological roots of political ideology than answers—indeed, results so far from this sample have largely *failed* to support the idea of significant differences between liberals and conservatives in psychological factors of root-level emotional sensitivity and approach and avoidance motivations. Chapter Four dealt with self-reported behavioral inhibition and behavioral activation in survey data, finding a few unexpected relationships but mostly a host of null results. Chapter Five moved beyond self-report into an investigation of trait level differences in approach-avoidance and emotional sensitivity by the reading of electrical signals off the scalp in search of brain patterns connected to negative and positive emotions. These results suggested very few differences in emotional sensitivity between liberals and conservatives, which goes against expectations from virtually any researcher working on the political psychology of political ideology. In this chapter, we move past self-reported traits and trait-level neurophysiological patterns, towards actual behavioral data on a behavioral task.

When this chapter closes, we will continue to find a counter-intuitive (but robust) story of no significant psychophysiological differences between liberals and conservatives, this time in the context of conflict monitoring processes during a Go/No-Go task. I also address the replication efforts of two additional studies—one which intended to directly replicate Amodio et al. 2007 as I have done, and one which
collected similar data to Amodio et al. 2007 but did not discuss the implications of their data. Thus, this chapter ultimately looks at all available conflict monitoring in the brain of liberals and conservatives across four sets of researchers. Amodio et al. 2007 are cited frequently, but no review paper has yet looked at the aggregate findings across all four of these research teams.

At the close of this chapter, having reviewed new and old evidence of conflict monitoring and political ideology, I believe the Amodio et al. 2007 findings need more attention from scholars. The original data presented in this chapter, on a sample of 51 non-student adults, do not replicate any part of Amodio et al. 2007 at all. One study (Weissflog et al. 2013) has mixed results for replication, and a second related study (Inzlicht et al. 2009) fails to replicate. Implications of this struggle to replicate this important work are discussed at the conclusion of this chapter.

## **Conflict Monitoring and Executive Functioning**

To think and act, the human brain must engage in several attentional and monitoring behaviors. Cognitive scientists broadly refer to these processes as executive functioning. Executive functioning is a set of cognitive processes that are necessary for the cognitive control of behavior. Executive functioning allows people to identify, pursue, retool, and resolve goal-directed behavior. Some of these executive functions include control of attention, inhibition control, and memory processes. Our capacity to plan, problem solve, and logically reason are all due to our executive functioning. The broad array of executive functions in the brain are tremendously influential on behavior, and impairment in executive functioning is linked to a suite of cognitive and personality disorders.

Executive functioning is about the control of action, where "deliberate, conscious control of activity is desired rather than those that are automatic" (Norman & Shallice, 1986, 1). Norman and Shallice (1986, 2) posit five types of situations that require deliberate attentional resources:

- 1. They involve planning or decision making.
- 2. They involve components of troubleshooting.
- 3. They are ill-learned or contain novel sequences of actions.
- 4. They are judged to be dangerous or technically difficult.
- 5. They require the overcoming of a strong; habitual response or resisting temptation.

In common between these situations is the need to control the response in order to avoid error.

Researchers have classified the cognitive process designed to identify potential response conflict and motivate effective performance to overcome the conflict as conflict monitoring (Amodio et al. 2008; Kerns, 2004). Response conflict is the occurrence of incompatible response tendencies, or an instance where reflexive or habitual response appears to conflict with the individual's immediate goals. Conflict monitoring, therefore, helps control our control (Kerns, 2004). Conflict monitoring in this dissertation is synonymous with error detection and behavioral conflict detection.

One of many behavioral tasks in psychology labs known to tap into conflict monitoring is the Stroop color-naming task, one of the most famous and important experimental tasks ever devised. In the generic variant of the Stroop, participants view a printed word that represents a color (e.g. "RED," or "BLUE"), and they are tasked with speaking the written word aloud. This task is relatively simple during the congruent trials, where the word "RED" is written in red ink. However, incongruent trials are mixed in, where the word "RED" is written in *blue* link. Incongruent trials require the participant to suppress the reflexive impulse to say the color of the ink in order to correctly say the written word aloud. Participants take longer and makes more errors on incongruent trials because of the increased difficulty.

The Stroop task requires individuals to engage in conflict monitoring. It requires subjects to override a strong, prepotent response in order to accurately complete the trial. Subjects in the Stroop task are habituated to quickly viewing a word and quickly responding by naming a color. Once the prepotent response is established, it requires executive control to step in and inhibit the response. Conflict monitoring is about detecting a potential conflict between goal and behavior, and cuing the brain to attempt to resolve the conflict.

The Go/No-Go task is another behavioral task for measuring individual differences in conflict monitoring. Like the Stroop color-naming task, the Go/No-Go task creates a habitual prepotent response of pressing the "Go" button, which must be unexpectedly inhibited when the rare "No-Go" signal flashes on the screen in place of the "Go" signal. This chapter utilizes the Go/No-Go behavioral task, and more details about the experiment are below.

Although accuracy on the tasks is one reasonable measure of conflict monitoring, researchers have also expanded the scope of their investigation into the neurocognitive mechanisms responsible for conflict monitoring. Cognitive scientists studying the brain have learned that conflict monitoring largely involves the anterior cingulate cortex and the dorsolateral prefrontal cortex (Botvinick et al. 2001; Botvinick et al. 2004; Kerns 2004). Neuropsychological studies of lesion patients have helped demonstrate the importance of ACC related executive function for everyday cognitive and social behavior (Bush et al., 2000; Holroyd & Yeung, 2012). Below, we outline the role of the anterior cingulate cortex as it relates to conflict monitoring in more detail.

# Anterior Cingulate Cortex and Psychological Processes

Conflict monitoring involves functions of attention, cognitive control, and memory. Processes like this are virtually never fully isolated to a single brain region. It is important to understand that many regions of the brain are related to emotion and cognition, complexly interweaving many different brain regions. Strict topographical understandings of brain function tend to underestimate the interplay that occurs between many regions in response to a single stimuli. And, of course, region-of-interest studies (necessarily) tend to downplay the role of the whole-brain in favor of the particular region being highlighted and investigated in a particular study. Figure 11. The Anterior Cingulate Cortex.



Those warnings understood, research has nonetheless demonstrated a consistent special role for the anterior cingulate cortex (ACC) in conflict monitoring and conflict detection. The anterior cingulate cortex is the frontal area of the cingulate cortex, a centrally located region immediately above the corpus callosum (see yellow region in figure). Over the past few decades, researchers have broadly understood that the ACC is related to cognition and emotion in the human brain, with particular specialization in attention and self-regulation (Botvinick et al., 2001; Botvinick et al., 2004; Bush et al., 2000; Kerns, 2004; Van Veen & Carter, 2002). Research also suggests that emotions can moderate and mediate executive action and attention, as well (Kanske & Kotz, 2011).

As research on the anterior cingulate cortex has continued, a picture has been sketched of its particular specialization in emotional and cognitive behavior. Researchers have found the ACC is particularly active compared to other regions when the cognitive processes involve attention, detecting errors, and self-regulation (Amodio & Frith, 2006; Botvinick et al., 2001; Kanske & Kotz, 2011). Additionally, distinctive ACC activity spikes when individuals engage in top-down control to resolve detected conflicts during information processing (Botvinick et a. 2001; van Veen & Carter 2002).

Researchers also believe that the anterior cingulate cortex has a broad role in directing attention during behavioral tasks in the laboratory. During a computerized behavioral task with an accuracy component, a powerful response in the ACC can be observed when the subject realizes that they have made an error when attempting to be accurate. This error-detection processing appears to originate from the ACC (Gehring et al. 1993), and researchers have also found general conflict-monitoring processing also tied to the ACC (Luu et al. 2003). In theory, the ACC serves an adaptive purpose, helping us recognize errors between our passive behavior and our active goals, particularly if our current habitual behavior needs to be rapidly interrupted in response to new information or a new threat.

Additionally, the ACC has also been connected to modulation of autonomic activity (Devinsky et al. 1995), suggesting a role in subconscious or unconscious behavior direction. Lavin et al. 2013 note that neuro activity generated in the ACC is produced when subjects process conflicting social information, as well as detecting pain. Singer et al. (2004) and Jackson et al. (2005) linked ACC activity with empathy when viewing pain in others.

The anterior cingulate cortex has also been connected to many related psychological processes, including self-regulation and anxiety. Gray and McNaughton (2000) looked at animal models of mice, lesion studies of human beings, and the known effects of anxiolytic drugs (like Xanax) to formulate their theory of anxiety as an alarm system, with the ACC as a "cortical alarm bell" (pg. 137). The ACC alerts the mind in response to states of uncertainty, simultaneous activation of conflicting goals, and erroneous responding.

Gray and McNaughton 2000 specifically note that abstract conceptual goals are likely to be regulated in much the same way as simple, concrete goals. It should be expected, according to them, that more advanced cognitive processing will follow the same general rules. Anxiety about completing a dissertation follows the same broad neurocognitive rules of anxiety as walking into an ominous cave where predators may be present.

Research continues to confirm the Gray and McNaughton model of anxiety, and the role that the ACC plays in the detection of states that trigger anxiety. The ACC is important for the types of inhibited responding that is characteristic of anxiety (Hajcak & Foti 2008; Hajcak et al. 2003) and for the minimization of prediction errors (Ridderinkhof et al. 2004). A prediction error is when you expected an outcome but you turned out to be wrong. This concept is highly compatible with the idea of conflict and error detection noted above.

The ACC, thus, is part of the general system for regulating and modifying behavior by signaling when control is needed, usually as a result of some anxietyproducing event such as the commission of an error (Holroyd & Coles 2002), the detection of conflict (Yeung et al. 2004) or the experience of uncertainty (Critchley et al. 2001; Hirsh & Inzlicht 2008).

You can see that researchers are still triangulating the detailed functions of the anterior cingulate cortex, but work so far has demonstrated a number of pathways for how the ACC affects, facilitates, or mediates a particular kind of attentional information processing concerned with making mistakes and correcting behavior. In other words, the anterior cingulate cortex seems to play an especially important role in conflict monitoring.

Below, we turn to the Go/No-Go behavioral task and Gray's dual-systems of behavioral motivation, to try to understand how conflict monitoring is related to Gray's system of behavior inhibition and what this could mean for political ideology.

#### Go/No-Go and Gray's Behavioral Motivation Systems

The Go/No-Go behavioral task has been used by researchers as a measure of individual differences in behavioral inhibition. In past chapters, I argued that Gray's dual theory of behavioral motivations is potentially important to understanding psychophysiological differences between liberals and conservatives. Below, I sketch an outline for why Gray's work is relevant to conflict monitoring in the anterior cingulate cortex.

To review quickly, Gray (1982; 1990; Gray & McNaughton 2000) proposed that animals have a behavioral activation system (BAS) that motivates behavior towards goals and a behavioral inhibition system (BIS) that halts ongoing behavior in response to novel or unexpected stimuli. The behavioral activation system is conceptualized as the *go* system, motivating behavior towards the pursuit of goals. Goals, in this context, refer to the hundreds of directed impulses and small plans we have on a daily basis.

Colloquially, we often tend to think of goals as big picture ideas—you have a goal to complete this research paper, or a goal to leave work early enough to walk the dog before dinner. But Gray's definition of goal-directed behavior goes to a much smaller level (we might colloquially think of them as micro-goals). Under this framework, it is considered goal-directed behavior to wheel your rolling office chair over to the phone, so that you can place a call. It is also a goal to pull a tissue from the box to blow your nose. These goals require behavior, and Gray's theory conceptualizes the behavioral activation system as the mechanism for initiating goal-directed behavior.

The problem for us living things is that we cannot pursue these little goals with reckless abandon. Our environmental situation or context can change rapidly without notice, requiring us to change our immediate plans and halt ongoing behavior. For example, your plan to roll the chair over to the telephone needs to quickly be changed when you hear the sharp, distressed bark of the dog underfoot. When you hear the

146

noise of the bark, or feel the bump of his tail under the chair wheels, you will immediately *halt* your executing plan in reaction to the new information. To inhibit our current intended behavior requires the behavioral inhibition system. You should think of the behavioral inhibition system as being the *stop* mechanism. It halts ongoing behavior, attempts to disengage us from danger, and allows us to create a new behavioral plan in response to the new information.

Simultaneously in our minds, we are engaged in goal-oriented behavior while also keeping persistent surveillance over our changing environment. This is the heart of Gray's dual process system—humans and animals alike have a behavioral activation system that is engaged in current goal-directed behavior while also an active behavioral inhibition system that is quietly surveying the environment for emergent new information.

Now, a critical question has emerged about what behavioral inhibition means in psychological research. Amodio et al. 2008 note, "the primary source of ambiguity concerns whether BIS is associated with the tendency to halt ongoing behavior or to engage in active avoidance behavior in response to a potential threat. On one hand, much research has operationalized BIS in terms of behavioral inhibition, as originally suggested by Gray... on the other hand, many researchers have described BIS in terms of behavioral avoidance" (12). These two conceptualizations of the Gray system inhibition versus motivational avoidance—are rather different. Behavioral inhibition is the halting of ongoing behavior based on new information—it is the interruption of action. In contrast, behavioral avoidance is actively avoiding something based on new information—it is the action of avoidance. Behavioral inhibition is about *stopping a plan*, whereas behavioral avoidance is about *starting a plan of avoidance*.

Both functions could be linked to the concept of conflict monitoring as they are related to detecting a conflict, stopping conflictual behavior, and redirecting behavior to avoid the conflict. Still, conceptual clarity is necessary because *stopping a plan* and *starting a plan* could be fundamentally different behavioral motivations, threatening to confuse the underlying psychological mechanisms that we look at in this dissertation. Especially as they intersect with approach-avoidance theories.

Amodio et al. 2008 examine this conceptual confusion directly. Based on a review of the literature, they conceptualize BIS as specifically about inhibition. They argue that Gray's original conceptualization is theoretically more akin to halting ongoing behavior rather than initiating avoidance behavior. Then, using behavioral and electroencephalography measures, Amodio and colleagues find that the BIS self-report (based on the questionnaire from Carver & White, 1994) is correlated with ACC related activity on a Go/No-Go task. This means the BIS self-report is related to an attentional system for monitoring response conflicts. (A deeper discussion of BIS, BAS, inhibition and motivational avoidance is included in Chapter Four.) Additionally, they find that self-reported BIS is correlated with increased error-related negativity in the anterior cingulate cortex when subjects make errors on No-Go trials. In other words, when subjects are supposed to inhibit their button-pressing, but fail to do so, the anterior

148

cingulate cortex has a sharp moment of activity to reflect the detected error. This further establishes the relationship between Gray's BIS scale and ACC related activity during the Go/No-Go task.

Thus, this chapter considers the BIS system a method of *halting* ongoing behavior, suggesting that it is important or essential to the process of conflict monitoring. Additionally, I conceptualize the Go/No-Go task as a behavioral measure designed to test conflict monitoring in a way consistent with Gray's theories of a behavioral inhibition system. The experiment in this chapter is the behavioral and neural complement to the survey data on BIS in Chapter Four. Below, we complete the literature review by looking at how conflict monitoring in the anterior cingulate cortex, a feature of Gray's behavioral inhibition system, could be related to political ideology.

#### Go/No-Go Task and Political Ideology

The Go/No-Go task has already been applied to political ideology twice and religious belief once. Before wading in, though, we should pause on what we might expect to find. From the summary of literature above, we understand that conflict monitoring is a general mechanism for detecting when one's habitual response tendency is mismatched with responses required by the current situation. We understand that conflict monitoring is conceptually related to Gray's behavioral inhibition system, a powerful cognitive feature of how we interact with the world on a minute-by-minute scale. We also understand that conflict monitoring is not restricted to autonomic body systems or low-level cognition—it could just as easily be related to abstract thought. Given the above, how do these processes influence high level abstract thinking, such as liberal and conservative political beliefs?

Generally speaking, the political psychology of ideology literature predicts conservatives to be both motivationally avoidant and inhibited in cognitive style. Jost's Motivated Social Cognition Theory and Hibbing's Negativity Bias Theory address this issue only in passing, as our investigation in Chapter Two details. Recall that prior theories in political psychology have conflated approach-avoidance tendencies with emotional sensitivity differences. Approach and avoidance sensitivity is not the same as emotional sensitivity. The Motivated Social Cognition theory of Jost and colleagues posits that conservatives are psychologically motivated by needs to reduce uncertainty, ambiguity, threat, and disgust (Jost et al., 2003). The psychological needs of conservatives ought to be associated with increased sensitivity to avoidance and inhibition signals. As avoiding uncertainty, ambiguity, threat, and disgust are connected to Gray's theory of behavioral avoidance and behavioral avoidance has a connection to inhibition in Gray's theories, we might expect to see conservatives being higher in behavioral inhibition as well.

The relationship of ideology to motivational avoidance, inhibition, and negative affect has been explored in prior chapters. This chapter makes the claim that conflict monitoring is a part of the behavioral inhibition system. With that in mind, we can develop hypotheses on the relationship of conflict monitoring to political ideology, by sketching a picture of the relationship of motivational avoidance and inhibition in ideology.

Uncertainty avoidance in conservatives is part of a general predisposition towards caution and general avoidance (Jost et al., 2003; Jost et al., 2009). Jost's theories also include the idea of conservatives being more inhibited. For example, Jost and colleagues repeatedly cite the Block and Block (2006) study that describes that "inhibited" preschoolers disproportionally grew up to be political conservatives (Jost et al., 2009; Jost, 2009). This evidence is used to demonstrate that the psychological differences between liberals and conservatives noted by their Motivated Social Cognition Theory occur early in life, before the emergence of political awareness. While motivational avoidance and inhibition are different psychological constructs, as Amodio et al. 2008 details, it could certainly be the case that conservatives have more sensitivity to both, and we have reason to believe that sensitivity to one would be positively correlated with the other.

The Negativity Bias Theory, of John Hibbing and colleagues, on the other hand, posits that conservatives have more sensitivity to negative emotions. This causes conservatives to exhibit stronger physiological reactions and focus greater attention to negative stimuli (Hibbing et al 2014). Again, emotional sensitivity is not the same construct as motivational avoidance, inhibition, or conflict monitoring, so Hibbing and colleagues have not explored how either would be related to their theory. In my view, being sensitive to negative emotions, though, has clear overlap with conflict monitoring. Behavioral avoidance—the avoiding of unpleasant things—is typically an emotionally negative experience. Feelings of fear, disgust, and sadness are all negatively valenced and associated with motivated avoidance. Most negative emotions trigger motivational avoidance, although there is one exception. Anger is emotionally negative but behaviorally causes individuals to approach, as it directs the angry individual to engage with the subject of the anger.

In another Hibbing paper, Dodd et al. 2012 develop the idea that individual-level variation in physiological and attentional responses to aversive and appetitive stimuli are correlated with political ideology. Using skin conductance measures and eyetracking software, they find that conservatives have more sensitivity to aversive stimuli while liberals have more sensitivity to appetitive (pleasing) stimuli. They demonstrate that those on the right not only respond more strongly to aversive images but also devote more attention to aversive images.

The research from Dodd, Hibbing, and colleagues does not mention conflict monitoring specifically, but their study of attention opens the idea that conflict monitoring differences may exist. They find that both liberals and conservatives pay more attention to aversive images than appetitive images, but conservatives' attention to aversive images is relatively greater and more pronounced. Dodd et al. 2012 suggest that attention should be studied alongside other psychological and physiological differences of liberals and conservatives. Thus, the Negativity Bias Theory would seem to be supportive of there being a relationship between conservatives and sensitivity to aversive stimuli. Hibbing, et al., 2014 note several studies that have found that conservatives focus more on threatening images and react more strongly to being startled (e.g. Dodd et al., 2012; Oxley et al., 2008). These reactions should properly be categorized as of an "inhibitory" nature rather than motivationally avoidant—the motivational drive to avoidance occurs chronologically *after* the startle reflex in response that Hibbing's team observed.

Overall, while both theories would be deeply interested in how conservatives would differ from liberals on conflict monitoring they have thus far declined to wade deeply into the topic. Three different papers have looked at these issues in the brains of liberals and conservatives using the Go/No-Go task. We take them up below.

#### Amodio, Jost, and the Go/No-Go Behavioral Task

Amodio et al. (2007) is the first study to look at conflict monitoring in liberals and conservatives on a Go/No-Go task (John Jost is one of the co-authors). They look at behavioral and electroencephalography data during a conventional Go/No-Go task. This study broke new ground in the literature by being one of the first to explore cognitive styles of liberals and conservatives with neuroscience techniques.

In the Go/No-Go task, participants must quickly respond to a frequently presented Go stimulus. Participants are habituated to expect the Go signal, which they receive about 80% of the time at random intervals. Each time the participant sees the Go signal, they must quickly (within 1000 milliseconds) press the Go button. However, 20% of the time, at unpredictable intervals, a No-Go signal appears. Upon seeing the unusual No-Go signal, participants must withhold the habitual pressing of the Go button. As the Go signal has become the prepotent response, participants struggle to accurately inhibit pressing the Go button. The difficulty of the struggle is demonstrated by decreased accuracy during No-Go trials. Where Go trials typically have an accuracy rate of greater than 98 percent, No-Go trials can have a significantly decreased accuracy rate of 70 percent or less. It takes careful attention and cognitive effort to resist the habitual Go response in response to the No-Go signal.

To assess performance on the task, researchers calculate an accuracy rate for Go trials and a separate accuracy rate for No-Go trials. The process of conflict monitoring is observed in how the participants handle the No-Go trials. Fewer errors on No-Go trials represents stronger, or more successful, conflict monitoring.

Electroencephalography (EEG) data can also be collected during a Go/No-Go task, which gives another measure of behavioral data corresponding to conflict monitoring. As noted above, researchers have found EEG related activity during the Go/No-Go task to be especially distinctive in the anterior cingulate cortex.

#### Figure 12. The Error-Related Negativity Component.



Note: The ERN is the sharp negative trough that occurs after 50ms during the incorrect responses (darker shade) Source: Herrmann et al. 2004

An event-related potential (ERP) is a brain response measured with electroencephalography in response to a stimuli. Observing ACC activity during a No-Go error (that is, when the subject fails to inhibit pressing the Go button) reveals an particular ERP component called error-related negativity (ERN). As observed in Figure 12, Error-related negativity is a sharp spike in negative voltage that originates in the ACC region of the brain. This distinctive brain activity, typically occurring 50 milliseconds after the subject has locked in their response, represents the subject's realization that they pressed the button when they ought not have. A stronger error-related negativity response is associated with greater conflict monitoring. Figure 13. Common ERP Components.



Source: Figure from ERPinfo.org, a non-profit website run by Steven Luck and Emily Kappenman, UC-Davis.

Another event-related potential component correlated with the Go/No-Go task is the N2. The N2 is the second negative peak observed in the ERP waveform (see Figure 13). A stronger (more negative) N2 peak is associated with increased conflict monitoring activity in the anterior cingulate cortex.

In congruence with the literature described above, Amodio and colleagues conceptualized the Go/No-Go task as a measure of conflict monitoring, which is part of a self-regulation process. They also agree with current literature that conflict monitoring is positively correlated with neurocognitive activity in the anterior cingulate cortex (ACC), in particular the two event-related potentials known as the error-related negativity component (ERN) and the second negative wave peak (N2). So, which ideological group would have relatively higher conflict monitoring as measured in the Go/No-Go task? Following Jost's Motivated Social Cognition Theory, liberals are thought to be more cognitively flexible, reflecting higher tolerance for ambiguity and complexity, and greater openness to new experiences. Conservatives, in contrast, are cognitively rigid, tending to be more structured and persistent in their decision making. Amodio and colleagues (2007) posit that liberals' greater cognitive flexibility would translate into stronger and more active conflict monitoring. Thus, they predict in their 2007 article that liberals would make fewer accuracy errors on the Go/No-Go behavioral task, and would have stronger ACC related activity including larger average amplitudes for the ERN and N2 components. Their hypotheses are confirmed in the paper.

This theory is, however, a departure from the literature on Motivated Social Cognition Theory. Recall from the literature review that Amodio, in a different paper, (Amodio et al. 2008) establishes that the Go/No-Go task (and related ACC activity) is primarily a measure of behavioral inhibition. Our summary above of the political psychology of ideology—both the Motivated Social Cognition Theory of Jost and the Negativity Bias Theory of Hibbing—suggests that conservatives would likely be more inhibited. It seems clear that political conservatives ought to have more conflict monitoring or a stronger neural response to conflict monitoring processing.

The puzzle continues. In the literature review above, you can see that the Go/No-Go task, which taps into the anterior cingulate cortex, is consistently related to a

number of attentional cognitive processes, including conflict monitoring, response inhibition, and error detection. Noticeably absent from this list is "cognitive flexibility." So where does "cognitive flexibility" come from? Amodio and his colleagues appear to conceptualize the Go/No-Go behavioral task differently between papers. In Amodio et al. 2008, when searching for a neural correlate of Gray's Behavioral Inhibition System, they report that ACC related behavior from the Go/No-Go task is correlated with selfreported BIS scores, signifying that the Go/No-Go task is a measure of behavioral inhibition. We walk away from that paper thinking that the Go/No-Go task is about behavioral inhibition. And, if you independently consulted with Jostian theory of conservatism as motivation social cognition, you would assume that conservatives are likely more inhibited and, therefore, would have a stronger performance on the Go/No-Go task and more ACC related activity during the task.

However, Amodio et al. 2007 interprets the task differently from Amodio et al. 2008. Instead of behavioral inhibition, now this disposition is called cognitive flexibility. For this paper, Amodio and colleagues state that ACC activity from the Go/No-Go task reflects a general mechanism for monitoring response conflict, and they theoretically connect this general mechanism to a more flexible cognitive style for liberals. As they believe conservatives are more cognitively rigid, they will be less sensitive to conflict monitoring compared to liberals. Thus, liberals should have greater conflict-related ACC activity on the task compared to conservatives.

#### **Replicating Amodio and Jost in the Go/No-Go Task**

To sort out this potential confusion, it would be useful to see how other researchers have looked at the Go/No-Go task and political ideology. Only one published paper has directly replicated Amodio et al. 2007. Weissflog et al. (2013) purportedly confirm the Amodio et al. 2007 findings that liberals made fewer errors and produced larger ACC-generated ERPs (ERN and NoGo N2). A key contribution for their study is that they go beyond self-identification of political ideology, employing related scales of egalitarianism and traditionalism. This is an important extension of the Amodio and Jost work because they (Amodio et al. 2007) are limited to a selfidentification scale.

Weissflog and colleagues find that increased attitudes favoring social equality (egalitarianism) had the greatest contribution to ERN and N2 effects, and behavioral accuracy on the task was more strongly associated with openness to social change (traditionalism). In general, their findings follow the same story as the Amodio et al. 2007 paper, finding political liberalism across their scales was associated with improved accuracy and stronger ACC related neuroactivity on the Go/No-Go task. Like Amodio et al. 2007, Weissflog and colleagues interpret this finding as consistent with the Motivated Social Cognition theory (Jost et al. 2003), claiming this as support for the idea that liberals are more cognitively flexible than conservatives. Like Amodio and Jost, there is no account for how inhibition may or may not be associated with political liberalism. There are some puzzling findings within the Weissflog et al. 2013 article worth considering. First, there are some issues with the sample employed. The initial pool was 256 Canadian undergraduates, but only 34 were chosen for the experiment. This very small sample had only two males, raising questions about gender as a confound. Moreover, on the self-placement scale, they lack solid representation from conservatives and oversample of extreme liberals. 13 were somewhat to extremely liberal, 13 neither, 8 were somewhat to moderately conservative. Zero participants were near the high limit of conservatism. Amodio et al. 2007 have this same problem their article and appendices do not contain clear write-ups of the variable descriptives, but the scatterplot of the relationship between ideology and ERN amplitudes in Amodio et al. 2007 reveals very few moderate conservatives and zero extreme conservatives.

Additionally, the Weissflog paper finds no relationship between accuracy on NoGo trials and a measure of egalitarianism (p > 0.41), which is inconsistent with their stated theory and the theory of the Amodio et al. 2007 paper. With egalitarianism being one of the major pillars of the Jost et al. 2003 theory of conservatism as motivated social cognition, we need more information to try to understand why the theory appears to break down on this key relationship.

Third, Weissflog and colleagues do fail to replicate one of the key relationships in Amodio et al. 2007. Recall that Amodio and colleagues find two ERP components to be related to political ideology, the ERN and the N2. These two components are supporting evidence to the idea that conflict monitoring is related to political ideology. Although Weissflog and colleagues find consistent results with the N2 in their study, they fail to replicate the finding on the ERN (r = 0.27, p = 0.13). The ERN is very important to the claims in the Amodio article, and it is puzzling that the findings do not replicate. When Jost and Amodio (2012) examine the Weissflog study in a review paper, they do not comment on this counterintuitive finding.

Although Weissflog is the only direct replication of Amodio and Jost on the Go/No-Go task, there is another study that will be of interest to us with regard to political ideology. Inzlicht et al. 2009, in two small samples (n = 28 & 29), study the ACC during a Stroop color-naming task (described above) as it relates to religious conviction. They theorize that religious conviction should be related to reduced activity in the anterior cingulate cortex (ACC) because the religious belief system helps to reduce incidents of uncertainty, conflict, and error by strengthening convictions and narrowing attention away from inconsistencies. Inzlicht and colleagues indeed find that weaker religiosity is connected to stronger ERN. As Jost and Amodio 2012 note, "given the strong association between conservatism and religiosity, this finding is broadly consistent with [our] results" (60).

Unfortunately, this study also has a few puzzling interpretation issues. First, religious belief is fundamentally different than political belief, even if the two are typically correlated. And, in fact, Inzlicht et al. find no relationship between political conservatism and religious belief in their sample. Jost and Amodio did not report religiosity in their 2007 paper, so we do not know if they were able to measure or control for it.

Second, Inzlicht and colleagues wrap their theory around the idea of religious belief being a kind of anxiolytic drug that protects believers from cognitive conflict, reducing the need for conflict detection. The theory of Jost and Amodio around cognitive flexibility does not link into the anxiolytic drug hypothesis, and we could use more theoretical work to understand what the connection could be.

Third, another issue for connecting the Inzlicht et al. 2009 work to the Amodio and Jost findings is that Inzlicht et al. find no relationship between cognitive rigidity and religious belief (their table 1, pg. 388). Jost and Amodio 2012 (and, Amodio et al. 2007) theorize that cognitive rigidity is the reason why liberals outperform conservatives in the conflict monitoring tasks. Inzlicht demonstrates in study 1 that religion drives these effects on the behavioral task, and not cognitive rigidity does not. This leaves a puzzle for our interpretation, because Inzlicht and colleagues seem to test an important part of the Amodio and Jost hypothesis, and then rejects it.

Finally, as with Weissflog et al. 2013, Inzlicht et al. 2009 actually deliver another failed replication of a central Amodio et al. 2007 finding. As before, the ERN and conservatism results fail to replicate (their table 3 on pg. 390), with Inzlicht and colleagues finding no relationship between ERN activity and political conservatism. Additionally, Inzlicht and colleagues find no correlation between accuracy on the task and political conservatism whatsoever (their table 3, pg. 390). Remember that Inzlicht et al. 2009 is not using a Go/No-Go task, but is instead using a Stroop task. These two tasks are conceptually similar in the way they are used by Inzlicht et al. and Amodio et al.

The discussion above illustrates the need for more replication of the Amodio et al. 2007 finding. The remainder of this chapter describes my study which is intended to directly replicate Amodio et al. 2007 using a larger sample of non-students while closely imitating the exact parameters of the Amodio Go/No-Go task. After I describe my findings, I will summarize how the three published studies intersect with the findings of this chapter.

# Hypotheses

This chapter focuses on a replication of Amodio et al. 2007. Although the literature review above raises more questions than answers about what kind of effects we ought to find for political ideology, I will lay out the same hypotheses as Amodio et al. 2007. The Conservatism as Motivated Social Cognition theory posits that conservatives have greater cognitive rigidity compared to liberals. This cognitive rigidity causes conservatives to be less adaptive to new and unexpected information. In a Go/No-Go task, Amodio and Jost expect that liberals will have a stronger (more negative) ERN response to making errors on No-Go trials, a stronger (more negative) N2 component on successful No-Go trials, and greater behavioral accuracy on the NoGo trials. The hypotheses are summarized in Table 9. Table 9. Go/No-Go and ACC Hypotheses.

#### Hypotheses (from Amodio et al. 2007)

- H1 Conservatives have weaker (less negative) error-related negativity (ERN), reflecting more cognitive rigidity compared to liberals.
- H2 Conservatives have a weaker (less negative) N2 component, reflecting more cognitive rigidity compared to liberals.
- H3 Conservatives will have less accuracy on NoGo trials, reflecting more cognitive rigidity compared to liberals.

Notes: ERN is the mean amplitude collected during -50ms and 150ms (response-locked) on NoGo error trials. N2 is the peak negative amplitude collected during 200ms and 400ms (response-locked) on NoGo correct trials.

## Method

#### Participants

Participants for this study are the same as in the prior chapters, but crucial details will be repeated here for completeness. Fifty-one non-student adults (age 24-52 years, M = 30.7, sd = 5.8) served as participants (32 female, 19 male). The testing session averaged less than two hours and 30 minutes, including breaks and the electroencephalography capping process. Participants completed four tasks during the session, and they had short breaks of five minutes between tasks. Participants were compensated at a rate of \$20/hour, with a minimum payment of \$40 and a maximum of \$50 for the session.

# Procedure

After arriving at the laboratory, participants were briefed on the tasks they would perform, and asked to sign consent forms. Afterwards, they were led to a dimlylit, sound-proofed room in a comfortable chair, approximately three feet from a nearby computer monitor. Experimenters were blind to participants' ideology scores and other attitudes.

After the capping process was completed, and testing was successful, participants completed four behavioral tasks in total, with numerous breaks. The total capping time was about 20-40 minutes, and the total testing time (including breaks) was about 110 minutes. The Go/No-Go behavioral task, the subject of this chapter's investigation, was the first of the four tasks. The Go/No-Go task took approximately 25 minutes to complete.

## Recording the EEG Signals

As detailed in Chapter Five, I employed software techniques to convert our 64 channels of recorded data to virtual channels that correspond to the traditional 10-20 system, which allows comparison of my work to other Go/No-Go studies. These virtual channels in the 10-20 system allow researchers using diverse systems to be able to compare their results via a common mapping of scalp electrodes. The virtual electrode system in this chapter, and the general data filtering procedure follows the specifications of Chapter Five. As with Chapter Five data, eyeblinks and other artifacts in the ERP signal were manually screened. ERP signals were filtered according to typical conventions. In this case, we used relatively strong filtering, removing frequencies below 1 Hz and above 30 Hz (96 dB, zero-phase shift). This strong filtering imitates the Amodio et al. 2007 study. *Measuring Conflict Monitoring on the Go/No-Go* 

Figure 14. An Individual Trial of Go/No-Go Task.



The procedure for collecting EEG data on the Go/No-Go task follows Amodio et al. 2007 closely. Subjects learned the Go/No-Go task with 100 practice trials, and then completed three blocks of 200 trials each, for a total of 600 trials. Figure 14 visually represents an individual trial in the Go/No-Go task. A fixation cross is first shown for 500ms, to focus and reset attention while steading the eyes and face. Then, a letter briefly flashes on the screen. The letter "M" was a Go signal, and the letter "W" was a NoGo signal. Participants then had 900ms to response. During a Go trial, when the letter "M" was displayed, participants are supposed to press a large black button on a remote controller in their hands with their pointer finger. During a NoGo trial, when the letter "W" was displayed, participants are supposed to do nothing and avoid pressing any buttons on their controller. After the response window ends, participants are shown a feedback screen for 1000ms, which informs them if they failed the trial. The feedback screen is triggered either by an incorrect response or if the participant responded after the 900ms interval (they were too slow). After the feedback screen, there is a final blank screen for 500ms before the next trial begins. After a block of 200 trials, participants received 30 seconds of rest before the next block began.

80 percent of signals were Go signals. When the high frequency of Go signals is coupled with a rhythmic, fast response, it creates a habitual, prepotent response in the subject to press the Go button in their hand. When the NoGo signal does trigger, for 20% of the trials, participants must inhibit their prepotent response and avoid pressing the Go button.

I measure three different kinds of data during the task. First, I measure an EEG component called error-related negativity (ERN), a strong negative wave that occurs after subjects make an error during a NoGo trial. An error during a NoGo trial means that the subject received the NoGo signal, but failed to correctly inhibit pressing the Go button. The ERN is the pattern that emerges when you aggregate all of the individual ERPs for each NoGo error trial together into a grand average, from -50ms to 150ms around the response. ERN occurs between 50 milliseconds before the error response through 150 milliseconds after the response. I measure the mean amplitude of the ERP

signal over that period of 200 milliseconds for analysis below. Amodio et al. 2007 used *mean* ERN amplitudes to estimate their ERP components, but Amodio et al. 2008 preferred *peak* ERN amplitudes. In this study, my results did not change by using peak amplitude measures, so I employed mean amplitude to be consistent with Amodio et al. 2007.

Second, I measured an additional EEG component called the N2, which is the second strong negative peak that occurs after subjects have a successful NoGo trial. A successful NoGo trial means the subject correctly inhibited pressing the habitual Go button after seeing the rarer NoGo signal. The N2 component is formed by aggregating ERPs from each successful NoGo trial together into a grand average waveform. Following Amodio et al. 2007, we utilize the mean negative amplitude between 200 and 400 milliseconds, during a successful NoGo trial.

Finally, I measured behavioral accuracy on NoGo trials, which is the number of correct NoGo trials divided by the total number of NoGo trials. Higher accuracy on NoGo trials represents greater conflict monitoring in the subject.

## Measuring Ideology

Five survey measures of political ideology were utilized in this sample. These subjects and the ideology measures employed are identical to prior chapters. Like Amodio et al. 2007 and other replications, the sample skews left, and does not contain any "extreme" conservatives. This sample has a deeper pool of solid conservatives than previous research teams have been able to gather. As a brief reminder, first, I utilize the conventional seven point self-placement scale from the American National Election Studies, which spans from "Very liberal" (1) to "Very conservative" (7). Second, I use the full Wilson-Patterson issue attitude battery (Wilson 1968; updated by Smith et al. 2011), which asks simple agree/disagree questions on 20 different topics in contemporary politics. The full scale is the aggregate number of conservative answers given in the 20 questions (ergo, a low score is more liberal and a high score is more conservative). Third, I use a subscale developed specifically for my data (see Chapter Four) that focuses on Wilson-Patterson items related to economic conservatism. Fourth, I use a parallel subscale designed to tap into social conservatism using Wilson-Patterson items. Finally, I employed a relatively uncommon, but useful, scale from the World Values Survey that focuses on principles of economic justice and fairness.

## Results

Before proceeding into the hypotheses, I will note a couple of results that speak to the general Go/No-Go task. In terms of accuracy in the task, NoGo trials were significantly more difficult than Go trials (accuracy of  $M_{nogo} = 0.77$ ,  $M_{go} = 0.995$ ; t(100) =12.029, p < 0.01), as expected. Additionally, as expected, being more accurate on Go trials is correlated with being more accurate on NoGo trials (r = 0.37, p < 0.01). Individual differences in task performance are expected.

In terms of response time on the task, as expected, average response time on correct Go trials was slower than response time on NoGo error trials ( $M_{go}$  = 200.3,

 $M_{nogoError} = 165.8$ ; t(100) = 6.1343, p < 0.01). NoGo errors are a failure to inhibit the button pressing, so we would expect them to be made in haste. Also, as expected, response time on correct Go trials is correlated with response time on error NoGo trials (r = 0.695, p < 0.01). Again, individual differences in task performance are expected, so differences in response time should be correlated.

In terms of the ERP components being examined, a couple of important observations of this data. First, as expected, the mean ERN and peak ERN are very strongly correlated (r = 0.91, p < 0.01). Second, as expected, the mean N2 and peak N2 are also extremely highly correlated (r = 0.93, p < 0.01). These associations reveal why my results would not change substantially by using one variation of component measurement versus another.

Table 10. Correlation Matrix of Go/No-Go and ERP Component Results.	



\* p < 0.05; ‡ p < 0.10. Correlation tests are two-tailed.

Note on ideology scales: higher values reflect more conservatism. Self-reported is the ANES 7 point scale. Issue attitudes is the full Wilson-Patterson scale. Economic and Social issues are Wilson-Patterson subscales. Economic Justice comes from the World Values Survey.

Table 10 presents correlations between the major variables of interest for the Amodio and Jost hypotheses.

# Hypothesis 1: Conservatives have weaker ERN

The first hypothesis is that political conservatism should be associated with a weaker (less negative) ERN component during errors made on NoGo trials. I fail to confirm this hypothesis. Contrary to the findings in Amodio et al. 2007, I find no statistically significant relationship between self-identified political ideology and ERN strength. Figures 14 and 15 visualizes data from Amodio et al. 2007 alongside with data from this chapter.

# Figure 15. Grand Average ERNs and Political Ideology



Amodio et al. 2007. Demonstrates stronger ERN for Liberals versus Conservatives.



Data from this chapter. No ERN differences between Liberals and Conservatives.

Figure 16. Scatterplots of Ideology and ERN Data.



Amodio et al. 2007. Demonstrates strong relationship between ideology and ERN.



Data from this chapter. No relationship between ideology and ERN.

Recall that each replication study of Go/No-Go and political ideology utilized the self-identified ideology scale. Although my failed replication of the Amodio et al. 2007
findings is surprising, my null results are consistent with findings from Weissflog et al. 2013 and Inzlicht et al. 2009. Of four research teams with this data, three cannot find this relationship in their data.

I employed four other alternative measures of political ideology, including an index of 20 political issues that have been aggregated together (Wilson-Patterson), an economic conservatism subscale from the Wilson-Patterson scale, a social conservatism subscale from the Wilson-Patterson scale, and a battery of questions concerning economic justice from the World Values Survey. None of these measures have any relationship to ERN measures whatsoever, consistent with my null finding on the selfidentification scale.

As Amodio et al. 2007 and Inzlicht et al. 2009 did not utilize alternative measures of ideology at all, the only replication paper that might contribute to this mystery is Weissflog et al. 2013. Weissflog et al. 2013 utilized Altemeyer's (1999) Right Wing Authoritarian Scale (1996) and Kluegel and Smith's egalitarianism and inegalitarianism scale (1986), which are at least distantly related to the general construct of political ideology. Weissflog et al. 2013 found that the liberal ends of each scale were correlated with stronger ERN responses. So, in this respect, my findings about alternative measures of ideology are at odds with the Weissflog findings, although it is worth considering how conceptually similar Right-Wing Authoritarianism and egalitarianism The second hypothesis is that political conservatism should be associated with a weaker (less negative) N2 component during successful NoGo trials. Unlike both Amodio et al. 2007 and Weissflog et al. 2013, I fail to find a statistical difference between self-identified liberals and conservatives on the N2 component. In fact, the correlation coefficient of self-reported ideology and the N2 component is trending towards statistical significance p = 0.055, but this effect is *reversed* from Amodio and Jost expectations—it is the self-identified conservatives who have a stronger, more negative, N2 component on the Go/No-Go task.

Table 10 shows that none of the alternative political ideology measures replicated the N2 results of Amodio and Jost, either. Unexpectedly, there is a statistically significant relationship between economic conservatism and a stronger (more negative) N2 component. This finding is unusual for a number of reasons. First, it is contrary to the Amodio and Jost predictions, as it is liberals who ought to have stronger N2 responses based on their 2007 paper, but my conservatives are so. Second, Jost's Motivated Social Cognition Theory treats *social* conservatism as a major part of being conservative, and it is odd to find the N2 completely unrelated to social conservative but instead important for economic issues, while at the same time unrelated to the economic justice scale. Overall, we should treat this finding with caution—it is inconsistent with the general picture of the null relationship, and it goes against all predictions. The third hypothesis concerns whether conservatives have less accuracy on the difficult NoGo trials compared to liberals. While Amodio et al. 2007 finds liberals to have superior performance (r = 0.30, p < 0.05), I do not replicate this finding. Although the relationship does not quite reach the 0.05 threshold, self-identified conservatives in my sample are correlated with superior accuracy during NoGo trials (r = 0.275, p = 0.055). If anything, my self-identified conservatives are more accurate than liberals on the trials. Conservatives also outperform liberals on Go trials (r = 0.326, p < 0.022); Amodio et al. 2007 did not report data about Go trial accuracy and ideology.

The pattern of findings in the self-identified conservatives do not reappear in alternative ideological measures, giving pause in how to interpret the findings above. **Discussion** 

In summary, this chapter described the current state of knowledge about whether liberals and conservatives may have differences in their basic conflict monitoring processing, in the context of a Go/No-Go task. Following a ground-breaking study from Amodio et al. 2007, I set out to replicate their study as a way to sort out confusion over how the Go/No-Go task should be interpreted. If the conflict monitoring aspect of the Go/No-Go task is akin to behavioral inhibition, we would expect conservatives to have a stronger performance and neural activity on the task. However, Amodio et al. 2007 interpreted the Go/No-Go task as a measure of cognitive flexibility, asserting (and finding) that liberals outperform conservatives on the task, as well as have stronger neural correlates during the task.

Table 11. Summary of Go/No-Go Studies.

	conjuct monitoring measurements				
Research Studies	Stronger ERN	Stronger N2	Greater Accuracy on Difficult Trials		
Amodio et al. 2007	Liberals	Liberals	Liberals		
Weissflog et al. 2013	null	Liberals	Liberals		
Inzlicht et al. 2009	null	(n/a)	null		
Wendell 2016	null	null	Conservatives		

Conflict Monitoring Measurements

Notes: Ideology relationships classified here are based on the self-identification scale, the only measure of ideology that is consistently employed by all four studies. Also note: Inzlicht et al. 2009 did not report N2 relationships between conservatism or religious belief.

My study fails to replicate any part of the Amodio et al. 2007 study. Additionally, I wrote about two other studies that closely replicate Amodio et al. 2007, each which had their own trouble replicating the findings (Weissflog et al. 2013 had several elements successfully replicate). A summary of those findings are in Table 11. Overall, these studies are rather mixed in their findings. It appears that liberal and conservative differences require additional investigation and more careful study. My work demonstrates that liberal and conservative physiological differences may exist, but not concerning conflict monitoring in the anterior cingulate cortex.

As before, the results of this chapter urge caution moving forward for the psychological differences noted by Jost, Hibbing, and their colleagues. Further discussion of the big picture of the findings of this chapter are taken up in the next chapter.

# **CHAPTER SEVEN**

# **DISCUSSION AND CONCLUSION**

The political psychology of ideology holds great promise for our understanding about the origins of political beliefs. Political psychologists working in this tradition are beginning to understand the complex web of physiological, emotional, and cognitive factors that make up our political minds. This new wave of research has successfully pushed our understanding of political attitudes deep within the black box of human cognition, and it moves us closer to understanding our political identities.

Biology and politics scholars have established a research program based on differences in root psychological characteristics of liberals and conservatives. These differences have been shown to be both emotional and cognitive, and they have been linked to differences in behavioral outcomes. Additionally, and importantly, these scholars have shown that dispositional differences between liberals and conservatives emerge even outside of political contexts or political information. In other words, the root psychological differences between liberals and conservatives are not restricted to political topics alone, but can manifest in basic, everyday behavior and thought.

Two of the most important theories about the political psychology of political ideology featured prominently in this dissertation. The Negativity Bias Theory, my term for the research program coming from John Hibbing, Kevin Smith, and the Political

Physiology Lab of Nebraska University, posits that political conservatives have a persistent bias towards negativity, which attracts their attention and triggers stronger physiological reactions. The Motivated Social Cognition Theory, developed by social psychologist John Jost and numerous colleagues over the past few years, argues that psychological factors of fear, threat, and uncertainty avoidance motivate an individual to adopt conservative political beliefs.

The three studies presented in this dissertation follow closely in the literature that proceeded them. First, I focused on dispositional traits related to emotion and cognition from outside of a political context. Second, following current work in psychology, I noted that negativity and avoidance are not the same theoretical construct, and this allowed my work to try to distinguish between the assorted dispositional differences noted by Hibbing and Jost. Third, my studies inherited hypotheses from these major theories. From the Negativity Bias Theory, I hypothesized that conservatives are more sensitive to negative emotion. From the Motivated Social Cognition Theory, I hypothesized that conservatives are more sensitive to avoidance and inhibition. Thus, the studies described in prior chapters establishes footing on these two theories, and in the theoretical space first outlined by research teams led by Hibbing and Jost.

My work contributes to this research program in two ways. First, I have integrated Gray's theories of behavioral inhibition and approach alongside these two prominent theories. My goal was to demonstrate how inhibition and activation are conceptually related to the work of Hibbing and Jost, and then test a series of dispositional traits that ought to differentiate liberals from conservatives. I had hoped to clarify and enrich our understanding of the research on dispositional differences. Second, in the process of investigating inhibition and approach, I did a direct replication of one of the most important political neuroscience studies published in the subfield of biology and politics. This replication closely mimicked the methods of Amodio et al. 2007, while also using a larger sample of non-student adults. As of August 2016, Over 400 papers have cited Amodio et al. 2007 but there have been very few replications of this important study.

Overall, the studies in this dissertation have presented surprising results that urge caution for this research program. While searching for dispositional differences between liberals and conservatives with multiple methodologies, I was surprised to learn that very few dispositional differences could be found between liberals and conservatives. The bigger picture of my work could largely be called a robust null finding. This final chapter outlines my findings and integrates the bigger picture of the work with recommendations for how future studies should proceed. In particular, I will draw attention to the failed replication of Amodio et al. 2007, and urge caution for future studies of political cognition.

In the remainder of this chapter, I review the need for more replication in social science, and well as discuss the findings in this dissertation and the implications of my work for understanding how liberals and conservatives differ in cognitive flexibility.

## **Replication Efforts in Social Science**

My dissertation joins the recent movement towards more replication in social science. Social science has never been more popularly read or more influential than it is today. Driven by increased public outreach by scientists and universities, combined with a more empirically minded style of journalism, social science research has exploded into the public policy scene. This increased demand for social science is a good thing for society, informing debates and increasing the level of knowledge through rigorous, data-driven analysis. Never before has social science been so important to good governance and society. But, also, never before has it been more important to make sure that science is methodologically rigorous and open to analysis by outside research teams.

We already have examples of how errors in research have deeply affected the policy world. For example, in 2013, a graduate student studying economics uncovered an innocent spreadsheet error, and overturned an authoritative academic study that was being utilized by heads of state and Nobel prize winners to theorize over the role of government debt in the industrialized world (Roose 2013; Pollin and Ash 2013).

We also have examples of how the drive for fame and funding can tempt scholars towards cheating and fraud. Broockman and Kalla's uncovering of Michael LaCour's fraudulent publication in *Science* shocked the entire academic world (Singal 2015). The story was picked up by numerous major media organizations, including the New York Times and FiveThirtyEight(Carey and Belluck 2015; Bialik 2015). Another case, even more galling, is Dutch social psychologist Diederik Stapel notoriously faking data for years, resulting in at least 55 publications made with fraudulent data (Bhattacharjee 2013; Neuroskeptic 2015; Univers 2012).

These cases, rare and remarkable, made a huge splash in the media. But the larger struggles of social science to foster reproducible research have also become more public in recent years. Many commentators have discussed the issues of publication bias, researcher degrees of freedom, and p-hacking. Co-authors Leif Nelson, Joe Simmons, and Uri Simonsohn and their academic blog called Data Colada (http://datacolada.org/) as well as popular academic blogger Andrew Gelman (http://andrewgelman.com/) have coined the term "researcher degrees of freedom" as short-hand for a number of the issues that can arise with researcher null hypothesis testing. Publication bias, where academic publications tend towards publishing successful experiments that confirm hypotheses while confining negative results to a file drawer, has a sharp impact on the way modern science is practiced today. Recently, social scientists have become more and more anxious about these problems. Researcher Brian Nosek and the Many Labs project (over 270 scientists) created an initiative called the Reproducibility Project, published in Science, which aimed to sponsor and conduct replication studies of 100 recent psychology studies (Yong 2015). Unfortunately, only 40 percent of those replications successfully replicated the originals, and, on average, the size of the effect in the studies that did replicate were only half as large (Yong 2015).

These events illustrate the need for more replication of important social science work. My dissertation is intended to contribute to that need. Below, I summarize my results and discuss the implications for my replication of several key studies related to the political psychology of ideology.

## Summarizing Dissertation Results

In Chapter Four, I examined the survey data collected from 466 non-student adults employed by Loyola University Chicago. The primary purpose of the chapter was to summarize the demographic characteristics and politics of the sample from which I drew participants to undergo behavioral experiments in the lab. However, this survey also gave us an opportunity to test for differences self-reported dispositions towards inhibition and activation in a large N study, which could not be done on the small samples typically brought into laboratories. This work taps into the Motivated Social Cognition hypothesis about conservatives and behavioral inhibition. Based on prior research into the psychological profiles of conservatives, drawing from the empirical work of many research teams, we expected that conservatives would report greater behavioral inhibition compared to liberals. To my surprise, and against expectations from prior work, it was the liberals in the sample who reported higher behavioral inhibition. The correlation was moderate but statistically reliable.

Chapter Five combines self-reports in surveys on emotional affect with a measurement of electroencephalography in a laboratory setting. A survey questionnaire helped me assess trait sensitivities towards many discrete emotions, and

general negative affect. I also had the opportunity to measure frontal electroencephalography in these subjects, which can depict a pattern of brain activity that is known to be connected to both negative emotion and avoidance sensitivity. My findings, again, are surprising. Across all of the trait emotional sensitivity scales employed, I found robust null results. No evidence was found for the idea that conservatives are more sensitive to negativity in this sample. Additionally, the electroencephalography measure of frontal hemispheric asymmetry revealed that liberals and conservatives do not have different patterns of brain activity. As the patterns are a neural correlate of sensitivity to both avoidance sensitivity as well as negative emotion sensitivity, these findings are puzzling and challenging for both the Negativity Bias Theory and the Motivated Social Cognition Theory predictions, as conservatives in this study are neither sensitive to negative affect nor are they dispositionally avoidant. Overall, I find that liberals and conservatives do not have differences in emotional sensitivity or in frontal hemispheric asymmetry.

Chapter Six continues the investigation using electroencephalography measures alongside a behavioral experiment called the Go/No-Go task. My goal was to investigate conflict monitoring processes in the brain, which have been connected to avoidance and inhibition in past literature. The Go/No-Go task allowed me to look for differences in accuracy and anterior cingulate cortex related neural activity between liberals and conservatives. This study was also an attempt at a direct replication of an important and ground-breaking study in political neuroscience study by Amodio et al. (2007), by using a non-student sample and more richly measuring political ideology. I did not replicate these findings, however. Liberals in my study did not have better accuracy on the Go/No-Go task, and did not have statistically different neural activity in the anterior cingulate cortex.

Next, I will address the implications of my work for our understanding of liberals and conservatives. My results have large implications for how researchers should consider and study the work of Amodio et al. 2007. Below, I specifically focus on what we can learn about the cognitive flexibility of liberals and conservatives from studies of the Go/No-Go task.

## Political Ideology and Cognitive Flexibility

Amodio et al. 2007 introduced the research world to liberal and conservative differences in behavioral conflict monitoring, as measured with the Go/No-Go task. They find that liberals are more successful at inhibiting habitual responses with the introduction of unexpected information. This increased success is also correlated with increased neural activity related to the Anterior Cingulate Cortex (ACC), a region of the brain known to be associated with conflict monitoring. Amodio and colleagues review their evidence, and find that the pattern of their data suggests that liberals are more cognitively flexible.

Three statistical relationships form the foundation of the paper. The first measure, purely behavioral, is that liberalism is associated with increased accuracy on NoGo trials. Liberals are more successful at inhibiting their habituated Go response when they see a NoGo signal. Conservatives, in contrast, are more persistent in their response patterns, causing increased errors when a change in response behavior is called for.

The next two measures are based on event-related potentials (ERPs), which are consistent patterns of brain activity observable as electric signal on the scalp in response to particular stimuli. The second measure is an ERP called error-related negativity (ERN) during failed NoGo trials. ERN is a component that arises in the individual when they fail to correctly inhibit the Go button press during a NoGo trial. Upon realizing the error, there is a distinctive pattern of brain activity that represents a sharp negative trough about 50 milliseconds after the mistaken button press. Liberals have a stronger (more negative) ERN component in response to a failed trial, reflecting greater conflict monitoring neural activity. The third measure is an ERP called the NoGo N2, which is a component that arises in the brain in response to a successful inhibition during a NoGo trial. When subjects successfully inhibit the Go button press in response to a NoGo signal, there is a strong negative trough in the electroencephalography signal, occurring roughly around 200 milliseconds. This negative trough is the second observable trough in the signal, giving rise to the name "N2." Liberals were observed to have a stronger (more negative) NoGo N2 compared to conservatives. Again, this neural difference is believed to represent a difference in basic conflict monitoring sensitivity, with liberals being more sensitive than conservatives. As conservatives have

186

less conflict monitoring related neural activity, they are deemed to be less cognitively flexible and more persistent than liberals.

As of August 2016, the Amodio et al. 2007 paper has been cited over 400 times. Biology and politics literature in both Political Science and Psychology have relied on this paper for establishing that physiological and psychological differences between liberals and conservatives can be observed directly in brain functioning, even during a task that is completely unrelated to politics. These findings have been cited by Hibbing and colleagues as an example of how liberals and conservatives are different in their core or root psychological dispositions, deeply within the biological self (e.g. Hibbing et al. 2014). These findings have also been useful and illustrative for Jost and colleagues, as liberals being more cognitively flexible fits nicely with the Conservatism as Motivated Social Cognition theory (e.g. Jost and Amodio 2012; Jost et al. 2003). In fact, revealingly, the literature that Jost created and inspired is often referred to as the Rigidity-of-the-Right hypothesis.

Although Amodio et al. 2007 continues to be widely read and widely cited, there has been only one published attempt to replicate (Weissflog et al. 2013). There is also one other published study on religious belief and conflict monitoring using a Stroop task (Inzlicht et al. 2009). This study was referenced by Jost and Amodio 2012 as being broadly consistent with Amodio et al. 2007, and they consider it broadly as an extension of their work (Jost and Amodio 2012, 60). To the best of my knowledge, this dissertation presents the second attempt to directly replicate Amodio et al. 2007.

In Chapter Six, I reviewed all known data on the relationship between conflict monitoring and political ideology and offered a new study of my own. To review briefly, I find that Amodio et al. 2007 has not been successfully replicated in full by any published study. Weissflog et al. 2013 comes closest, and it does contain findings that are supportive of the Amodio et al. 2007 theory. Some attitudinal measures in this study (traditionalism and egalitarianism) are conceptually related to political liberalism, and are shown by Weissflog and colleagues to be correlated with increased behavioral accuracy and increased neural activity reflected in the ERN and N2. This fits with Amodio et al. 2007's three central findings, but used different scales for political ideology. However, in terms of ideological self-identification, the measure employed by Amodio et al. 2007, Weissflog and colleagues did not replicate one of the three core Amodio et al. Findings, specifically that self-reported liberals have stronger (more negative) ERN during the task. Data from the Weissflog et al. study trends in a consistent direction, but failed to reach conventional levels of statistical significance (p = 0.13). Weissflog et al.'s sample was 32 undergraduate students at a Canadian university. This sample size is not unusually small for a neuroscience study, but the small size could potentially explain why they failed to replicate (but, then, we would also need to be skeptical about the relationships that they did show.)

What about the Inzlicht et al. 2009 study? Inzlicht and colleagues studied the relationship of religious belief to conflict monitoring using a Stroop task in a conceptually similar way to a Go/No-Go task. Inzlicht and colleagues presented data on

22 undergraduates that showed decreased religious belief to be related to increased accuracy on the Stroop task and increased strength of the ERN. Non-believers performed better on the behavioral task and had a stronger ERN component. Jost and Amodio 2012 cite this paper as supportive for the theory of Amodio et al. 2007 because religious belief is closely related to political conservatism. In the view of Jost and Amodio 2012, Inzlicht's work demonstrates that religious believers—who are likely political conservative—do worse on the task and have a weaker ERN response because they are less cognitively flexible than religious non-believers (who are likely to be liberal) (Jost and Amodio 2012, 60).

However, this reading of the Inzlicht findings has several problems. First, Inzlicht et al. specifically find no relationship between religious belief and political conservatism in their sample (Table 3, 390) They also specifically find no relationship between political ideology and ERN amplitude (Table 3, 390). Inzlicht and colleagues state that "the correlation between the ERN and religious conviction did not diminish after we controlled for conservatism, IQ, or any of the Big Five personality factors" (389), arguing that political conservatism cannot explain the correlation between ERN and religious conviction that they found (389). Moreover, religious believers actually had *improved* behavioral accuracy on the task, which is contrary to what Amodio et al. 2007 and Jost and Amodio 2012 would have predicted. Finally, religious believers in Inzlicht et al. had no relationship to "cognitive closure" (Table 2, 388), which presents problems for considering the religious believers to be cognitively rigid. My study of liberals, conservatives, and the Go/No-Go task attempted to closely replicate the methods and design choices of Amodio et al. 2007. As described above, and with great detail in Chapter Six, I ultimately find no support for any differences between liberals and conservatives on the task. The three important metrics connected to liberals described in Amodio et al. 2007—increased behavioral accuracy, increased ERN strength, and increased N2 strength—are not present in my data. Before considering implications of these failures to replicate, we should consider the strength and usefulness of the samples across the relevant Go/No-Go studies.

# **Comparing Samples of Go/No-Go Data**



Distribution of Self-reported Ideology in Three Go/No-Go Studies

Note: Amodio et al. 2007 and Weissflog et al. 2013 utilized a self-report ideology scale with 11 points instead of the conventional 7 point scale used widely in political science and public opinion surveys such as the American National Election Studies. I transformed their data from 11 point to 7 point by multiplying each subject's score by (7/11). The histogram bins further smooths out all three distributions to provide a generalized picture of the data from the studies that allows clear comparison.

The figure above presents the ideological distribution of the samples employed by Amodio et al. 2007, Weissflog et al. 2013, and Chapter Six of my dissertation. Information about the ideological spread of Inzlicht et al. 2009's sample is unavailable, unfortunately. Data for the outside studies in this figure were compiled by me through close viewing of the supplied scatterplots. The raw data from both studies are not available online.

A key goal of replication of the Amodio et al. 2007 Go/No-Go experiment was to increase the number of political conservatives in the sample. As can be observed in the figure above, my sample includes substantially more subjects who self-place on the conservative side of the scale. Like the prior studies, I had difficulty finding extreme conservatives (e.g. the sevens) who were willing to come into the lab. However, I was able to recruit a total of 17 subjects who identified right-of-center. Amodio et al. 2007 and Weissflog et al. 2013 each had 8 subjects who identified right-of-center.

Another key goal was to broaden the sample in age range and size. Both Amodio et al. 2007 and Weissflog et al. 2013 utilized a convenience sample of college undergraduates. Amodio et al. 2007 featured 43 students recruited from both UCLA and New York University, while Weissflog et al. 2013 used 32 students from Brock University in Canada. I used a convenience sample of 51 non-student employees at Loyola University Chicago. Consequently, both Amodio and Weissflog have age distributions in the normal undergraduate college range, while my sample averaged 30.7 years old, with standard deviation of 5.8 years and a range of ages from 24-52 years. My sample and Amodio et al. 2007 each had 63% females, while Weissflog et al. 2013 was about 94% female (2 males of 32 participants). My sample was also approximately 18% larger than Amodio et al. 2007 and 60% larger than Weissflog et al. 2013. Overall, thus, my sample has considerable strengths over the other two studies in terms of representativeness both politically and demographically.

## Next Steps

Overall, the results from these studies offer a few important conclusions. First, in a big picture sense, it does not appear that Gray's systems of behavioral inhibition and behavioral activation manifest differently between liberals and conservatives. Whether measuring disposition towards inhibition and avoidance with a survey questionnaire, or with emotional valence in hemispheric asymmetry measures, or with the Go/No-Go task and anterior cingulate cortex measures, the results paint a fairly robust null finding. This was a worthwhile area of investigation, given the emphasis in prior literature on base level physiological differences between liberals and conservatives concerning related phenomena like emotional sensitivity towards threat, fear, and disgust. Yet, across these studies, we largely get a robust null finding for conservatives being more inhibited. More work with different samples could significantly contribute to these questions, but for the moment, generalizing from these three studies, it appears clear that liberals and conservatives do not have meaningful differences in behavioral motivations to inhibit (and, approach sensitivity, too, did not seem to be related to political ideology, which was a more exploratory objective in this dissertation).

A second important conclusion is that the work on emotional sensitivity in Chapter Five fails to find evidence that conservatives are more sensitive to negative emotion or more sensitive to avoidance. Persuasive work from Hibbing and colleagues suggest conservatives pay more attention to negative stimuli and are more sensitive to the effects of negative emotion. Jost and colleagues have research that suggests conservatives could be more avoidant in cognitive style. In Chapter Five I explore these ideas by measuring sensitivity to positive and negative affect, as well as measuring frontal hemispheric asymmetry, which is a well-known neural correlate for sensitivity to emotion and avoidance. With multiple measures of political ideology, I do not find a robust correlation between negative emotion and political conservatives on the emotional scales. I also do not find that conservatives have the hemispheric asymmetry pattern that has been correlated to negative emotion and avoidance sensitivity.

Finally, a third important conclusion is that the widely cited Amodio et al. 2007 has not been successfully replicated in whole. My experiment fails to replicate, and I bring attention to two published studies that also struggle to replicate some of the major findings of the Amodio et al. 2007 article. Given the relatively scarcity of studies that note important root cognitive differences between liberals and conservatives outside of Amodio et al. 2007, we continue to have a dire need for more studies in this area. Go/No-Go task performance and related anterior cingulate cortex activity will need additional studies to more firmly establish a connection between liberalism and cognitive flexibility because critical questions remain about whether a difference currently exists.

Briefly, we can summarize some of the standard limitations with studies like mine. First, as already noted in Chapter Four, the sample is not nationally representative. This pool of participants is drawn from one geographical location at one specific time and place. Moreover, they were unusual in the sense that they worked at a university and had a much higher level of education than the general public. As nonstudent adults, they do have the advantage of having a diversity of ages and experiences that are not typically present in most convenience samples. Second, the recruitment survey sample size was reasonably large for social science research (at 466 participants) but still not large enough to effectively capture a deep poll of extreme conservatives. Peer work in this area also has great difficulty getting strong conservatives into their study, and this study is an improvement over past studies in this regard.

Another limitation is that the measures in the survey are all self-reported, and we can only draw correlational inferences. The relationship I find between self-reported behavioral inhibition and liberalism could have several explanations that the survey data alone cannot account for. Principally, the challenge for interpretation is over how to interpret the self-reports. Are conservatives less inhibited, or do they report less inhibition on surveys, due to social desirability differences or self-perception differences? Additionally, as always with correlational studies, causation cannot be determined.

Later chapters address the weaknesses of survey data by bring subjects into the lab, but these studies have standard limitations as well. The sample size of the neuroscience studies was only 51 people. While 51 subjects is significantly above average for electroencephalography studies, more statistical power would always be useful for trying to spot effect sizes that are not expected to be very large.

The project of biology and politics continues to be incredibly promising for understanding the puzzle of political ideology. As Hatemi and McDermott (2011) write, biological models of political behavior have vast potential to contribute meaningfully to studies of political behavior. When I began this work in 2012, I was looking to expand our understanding of how conservatives and liberals differ dispositionally by looking at multidimensional ideology. At the conclusion of my work, however, I want to bring attention to something else entirely. The results of my studies strongly suggest a compelling case for why we need more work to better understand the dispositional differences of liberals and conservatives. My dissertation posits that the question of root psychological differences between liberals and conservatives is still largely open for consideration.

# APPENDIX A

# SURVEY QUESTIONNAIRE



LOYOLA UNIVERSITY CHICAGO Lake Shore Campus 1032 W. Sheridan Rd Chicago, Illinois 60660

Preparing people to lead extraordinary lives

#### Survey about American Politics

Thank you for agreeing to take this survey. Our research team is studying politics and personality. Your answers will help Loyola University Chicago researchers with important questions related to the political beliefs of citizens in the United States.

Below, we ask you about some of your political beliefs as well as some questions about you. We think you may enjoy answering these questions, and your cooperation will help us. Your answers will be confidential and your personal results will never be reported in a way that could identify you. We will take several measures to insure confidentiality; responses will be held under lock and key in the professors' offices, the page with your personal information will be separated from the response page and a random code will be used to identify your responses, after the research project is finished any identifying information will be destroyed. This survey should take approximately 5-10 minutes to complete. Please find the self-addressed envelope enclosed to send back your completed survey.

There are no foreseeable risks or discomforts involved in participating in this research beyond those experienced in everyday life. There are no direct benefits from this research to you from participation, but your participation will be useful in helping us develop a better understanding of how people think about politics and make political decisions, which is important to maintaining a healthy democracy. We understand that your participation is voluntary and we do appreciate you taking the time to help us. If you do not want to be in this study, you do not have to participate. Even if you decide to participate, you are free not to answer any question or to withdraw from participation at any time without penalty. If you have any questions concerning this research please contact Professor Richard E. Matland at Loyola University Chicago at (773) 508-7127. If you have further questions about your rights as a research participant, you may contact the Loyola University Office of Research Services at 773 508 2689. By completing the survey you are agreeing to participate in this portion of our research.

Thank you again for your help with this research. We hope you enjoy the survey!

Before we begin, what is your unique ID#? \_\_\_\_\_\_\_\_\_(You can find this in the letter or email that we sent to you)

1. Sometimes, people consider themselves to be liberals or conservatives when it comes to politics. On a scale of 1-7, with one being very liberal, 4 being centrist, and 7 being very conservative, where would you place yourself? (*circle the number*)



2. Below, you will see two opposed statements along a line of boxes. On a scale of 1 to 10, which statement tends to represent your viewpoint better? (*circle the number*)



Prayer in schools	Support	Oppose	War in Afghanistan	Support	Oppose
Pornography being sold to adults	Support	Oppose	Welfare spending	Support	Oppose
Strong penalties for illegal immigration	Support	Oppose	More tax cuts	Support	Oppose
Women's equality	Support	Oppose	Increases in gun control	Support	Oppose
Death penalty	Support	Oppose	Increases in military spending	Support	Oppose
Patriot Act	Support	Oppose	Police searches without warrants	Support	Oppose
Biblical truth	Support	Oppose	Increases in pollution control	Support	Oppose
Gay marriage	Support	Oppose	Small government	Support	Oppose
Legal abortions	Support	Oppose	Foreign aid	Support	Oppose
Patriotism	Support	Oppose	Free trade	Support	Oppose

**3.** Now, please indicate with a circle whether you support or oppose each topic below. Choose the answer that is closest to your beliefs. It will sometimes be difficult for you to choose because you can see both sides of the issue. When this occurs, try to pick the option that you think you lean towards more.

For the questions below please put an X next to the statement in each pair that you lean towards the most.

## 4a. In your opinion, society works best when ...

- \_\_\_\_ People live according to traditional values
- People adjust their values to fit changing circumstances

### 4b. In your opinion, society works best when ...

- \_\_\_ Expectations of how people should act are allowed to evolve over time
- Expectations of how people should act are based on an unchanging code

## 4c. In your opinion, society works best when ...

- Our leaders stick to their beliefs regardless
- Our leaders change positions when situations change

#### 4d. In your opinion, society works best when ...

- People assume those in faraway places are mostly like us
- People realize those in faraway places are mostly different from us

#### 4e. In your opinion, society works best when ...

We realize we need to take care of our own people first We realize people everywhere deserve our help

#### 4f. In your opinion, society works best when ...

Those who break minor rules are forgiven
Those who break minor rules are punished

## 4g. In your opinion, society works best when ...

\_\_\_\_ More fortunate members contribute more \_\_\_\_ Every member contributes equally

#### 4h. In your opinion, society works best when ...

People are rewarded according to merit People are rewarded according to need

## 4i. In your opinion, society works best when ...

- People join together to help others
- People take responsibility for their own welfare

#### 4j. In your opinion, society works best when ...

People are proud they belong to the best society there is People realize there are many ideal societies and no single best society

## 4k. In your opinion, society works best when ...

- Our leaders are mostly questioned
- Our leaders are mostly obeyed

#### 4l. In your opinion, society works best when ...

- Our leaders make the decisions
- Our leaders are forced to listen to others

#### 4m. In your opinion, society works best when ...

People recognize that flaws of human nature can be positively changed People recognize that flaws of human nature are unchangeable

### 4n. In your opinion, society works best when ...

Our leaders compromise with their opponents in order to get things done Our leaders adhere to their principles no matter what

If I think something unpleasant is going to happen I usually get pretty worked up.	SA	Α	D	SD	It would excite me to win a contest.	SA	Α	D	SD
I worry about making mistakes.	SA	Α	D	SD	When I see an opportunity for something I like, I get excited right away.	SA	Α	D	SD
Criticism or scolding hurts me quite a bit.	SA	A	D	SD	When I want something, I usually go all-out to get it.	SA	A	D	SD
I feel pretty worried or upset when I think or know somebody is angry at me.	SA	Α	D	SD	I go out of my way to get things I want.	SA	Α	D	SD
Even if something bad is about to happen to me, I rarely experience fear or nervousness.	SA	A	D	SD	If I see a chance to get something I want, I move on it right away.	SA	A	D	SD
I feel worried when I think I have done poorly at something.	SA	Α	D	SD	When I go after something I use a "no holds barred" approach.	SA	Α	D	SD
I have very few fears compared to my friends.	SA	Α	D	SD	I will often do things for no other reason than that they might be fun.	SA	Α	D	SD
When I get something I want, I feel excited and energized.	SA	A	D	SD	I crave excitement and new sensations.	SA	A	D	SD
When I'm doing well at something, I love to keep at it.	SA	Α	D	SD	I'm always willing to try something new if I think it will be fun.	SA	Α	D	SD
When good things happen to me, it affects me strongly.	SA	A	D	SD	I often act on the spur of the moment.	SA	A	D	SD

5. Now, we are going to ask you several questions about you and your personality. Please indicate whether you strongly agree (SA), agree (A), disagree (D) or strongly disagree (SD) with each statement.(*circle*)

6. Generally speaking, do you usually think of yourself as a Democrat, a Republican, or a	n
<b>Independent?</b> (put an X in the best fit)	

I am a Democrat		I am a strong Democrat I am not a strong Democrat
I am an independent	<b>→</b>	I am independent but lean towards Democrats I am independent from either party I am independent but lean towards Republicans
I am a Republican	<b>→</b>	I am a not a strong Republican I am a strong Republican
I am not sure	→	I am not sure

We are almost done with the questions. Here are a few questions to provide us with general demographic information to help us compare answers.

7. What is your gender? (put an X in the best fit) Male \_\_\_\_ Female

8. I consider myself:(put an X in the best fit)

- Strongly right-handed
- Somewhat right-handed
- Ambidextrous (Neither right-handed nor left-handed) Somewhat left-handed
- \_\_\_\_ Strongly left-handed

9. In what year were you born?

10. What is your current marital status? (put an X in the best fit)

- Married
- Widowed
- \_ Divorced
- Separated
- Never married
- Living with partner
- 11. What is the highest level of education you have completed? (put an X in the best fit)
  - No formal education

  - Elementary school (1<sup>st</sup> grade through 5<sup>th</sup> grade) Middle school (6<sup>th</sup> grade through 8<sup>th</sup> grade) Some high school, no diploma (9<sup>th</sup> grade through 11<sup>th</sup> grade)

High school diploma

- Some college, no degree
- Associate degree
- Bachelor's degree
- Master's degree
- Professional or Doctorate degree

12. Are you of Hispanic or Latino ethnicity, such as Mexican, Chicano, Puerto Rican, Cuban, or other background from Central America, South America, or the Caribbean? (*put an X in the best* fit)

\_\_\_\_No \_\_\_\_Yes

13. Which of the following best describes your race? (put an X in the best fit)

\_\_\_\_Black or African American

Asian American

Pacific Islander or Hawaiian

Native American or Alaskan Native

White or European American

\_\_\_\_\_ Two or more races

\_\_\_\_ Other: \_\_\_\_\_

Thanks for helping us with this survey! Your help is greatly appreciated. We just have one more question for you on the following page.

14. Do you wish to be considered for our research project involving brain imaging and society? We will offer you \$40 to \$50 for 2 or 2.5 hours of your time answering questions in our lab at Loyola University Chicago on the Lakeshore Campus. We will schedule the appointment at your convenience in the next month. The experiment is non-invasive and will not harm you in any way. If this sounds interesting please say yes and we will contact you with more details. Your participation will continue to be confidential. Your results will provide us very important data and your help would be greatly appreciated.

\_\_\_\_ No, I am not interesting in participating.

\_\_\_\_ Yes, I am interested in participating, please send me more information.

Name:			
E-mail address:			
Phone number(s):			
When is the best time to call?			

Thanks for your participation in this survey! Please mail back your completed survey using the included envelope.

# WORKS CITED

- Abramowitz, A. I., & Saunders, K. L. (1998). Ideological realignment in the US electorate. The Journal of Politics, 60(03), 634–652. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0022381600052075
- Abramowitz, A. I. (2010). The disappearing center: Engaged citizens, polarization, and American democracy. Yale University Press.
- Alford, J. R., Funk, C. L., & Hibbing, J. R. (2005). Are political orientations genetically transmitted. American Political Science Review, 99(2), 153–167. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=307695
- Alford, J. R., Funk, C. L., & Hibbing, J. R. (2008). Twin studies, molecular genetics, politics, and tolerance: A response to Beckwith and Morris. Perspectives on Politics, 6(04), 793–797. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract S1537592708081929
- Alford, J. R., Funk, C. L., & Hibbing, J. R. (2008). Beyond liberals and conservatives to political genotypes and phenotypes. Perspectives on Politics, 6(2), 321–28. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=1881740
- Alford, J. R., & Hibbing, J. R. (2004). The origin of politics: An evolutionary theory of political behavior. Perspectives on Politics, 2(04), 707–723. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S1537592704040460
- Alford, J. R., & Hibbing, J. R. (2008). The new empirical biopolitics. Annu. Rev. Polit. Sci., 11, 183–203. Retrieved from http://www.annualreviews.org.flagship.luc.edu/doi/pdf/10.1146/annurev.polisci.1 1.060806.161216
- Allen, J. J. B., Coan, J. A., & Nazarian, M. (2004). Issues and assumptions on the road from raw signals to metrics of frontal EEG asymmetry in emotion. Biological Psychology, 67(1), 183–218.

Allen, J. J. B., & Reznik, S. J. (2015). Frontal EEG asymmetry as a promising marker of depression vulnerability: Summary and methodological considerations. Current Opinion in Psychology. http://doi.org/10.1016/j.copsyc.2014.12.017

Altemeyer, B. (1996). The authoritarian specter. Cambridge Univ Press.

- Althaus, S. L. (2006). False Starts, Dead Ends, and New Opportunities in Public Opinion Research. Critical Review, 18(1-3), 75–104. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217270114/abstra ct?accountid=12163
- Amadae, S. M., & Bueno de Mesquita, B. (1999). THE ROCHESTER SCHOOL: The Origins of Positive Political Theory. Annual Review of Political Science, 2(1), 269–295. http://doi.org/10.1146/annurev.polisci.2.1.269
- Amodio, D. M., & Frith, C. D. (2006). Meeting of minds: the medial frontal cortex and social cognition. Nature Reviews. Neuroscience, 7(4), 268–77. http://doi.org/10.1038/nrn1884
- Amodio, D. M., Jost, J. T., Master, S. L., & Yee, C. M. (2007). Neurocognitive correlates of liberalism and conservatism. Nature Neuroscience, 10(10), 1246–1247. http://doi.org/10.1038/nn1979
- Amodio, D. M., Master, S. L., Yee, C. M., & Taylor, S. E. (2008). Neurocognitive components of the behavioral inhibition and activation systems: Implications for theories of self-regulation. Jan 2008. Psychophysiology, .45(1), pp. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/2007-18980-002
- Ansolabehere, S., Rodden, J., & Snyder, J. M. (2008). The Strength of Issues: Using Multiple Measures to Gauge Preference Stability, Ideological Constraint, and Issue Voting, 102. http://doi.org/10.2307/27644512
- Axelrod, R., & Hamilton, William D. (2006). The Evolution of Cooperation in Biological Systems. Basic books. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=KFf2HXzVO58C&oi =fnd&pg=PR7&dq=Axelrod+and+Hamilton+evolution&ots=9hVE8uTUFj&sig=6dfjHl Un11mTGLFYHW0k458JWvs
- Bartels L.M., & Brady H.E. (2003). Economic Behavior in Political Context. The American Economic Review, 93(2), 156–161. http://doi.org/10.1257/000282803321946976

- Beck, P. A., & Jennings, M. K. (1979). Political Periods and Political Participation. The American Political Science Review, 73(3), 737–750. http://doi.org/10.2307/1955401
- Becker, E. (1985). Escape from Evil (Reissue). Free Press. Retrieved from http://www.amazon.com/dp/0029024501
- Bennett, S. E. (2006). Democratic competence, before converse and after. Critical Review, 18(1-3), 105–141. http://doi.org/10.1080/08913810608443652
- Bhattacharjee, Y. (2013). Diederik Stapel's Audacious Academic Fraud. Retrieved from http://www.nytimes.com/2013/04/28/magazine/diederik-stapels-audaciousacademic-fraud.html?pagewanted=all
- Bialik, C. (2015). As A Major Retraction Shows, We're All Vulnerable To Faked Data | FiveThirtyEight. Retrieved from http://fivethirtyeight.com/datalab/as-a-majorretraction-shows-were-all-vulnerable-to-faked-data/
- Block, J., & Block, J. H. (2006). Nursery school personality and political orientation two decades later. Journal of Research in Personality, 40(5), 734–749. http://doi.org/10.1016/j.jrp.2005.09.005
- Bonanno, G. A., & Jost, J. T. (2006). Conservative shift among high-exposure survivors of the September 11th terrorist attacks. Basic and Applied Social Psychology, 28(4), 311–323. Retrieved from http://www.tandfonline.com.flagship.luc.edu/doi/abs/10.1207/s15324834basp280 4\_4
- Botvinick, M., Braver, T., Barch, D., Carter, C., & Cohen, J. (2001). Conflict monitoring and cognitive control. Psychological Review, 108(3), 624–652. http://doi.org/10.1037/0033-295X.108.3.624
- Botvinick, M. M., Cohen, J. D., & Carter, C. S. (2004). Conflict monitoring and anterior cingulate cortex: An update. Trends in Cognitive Sciences. http://doi.org/10.1016/j.tics.2004.10.003
- Boyd, R., GINTIS, H., Bowles, S., & Richerson, P. J. (2003). The evolution of altruistic punishment. Proceedings of the National Academy of Sciences, 100(6), 3531–3535. http://doi.org/10.1073/pnas.0630443100
- Brader, T. (2005). Striking a responsive chord: How political ads motivate and persuade voters by appealing to emotions. American Journal of Political Science, 49(2), 388–

405. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.0092-5853.2005.00130.x/full

- Brader, T. (2006). Campaigning for hearts and minds: How emotional appeals in political ads work. University of Chicago Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=CKQJA2RgS8sC&oi =fnd&pg=PR9&dq=Brader+Hearts+and+Minds&ots=kCyxjDvJED&sig=zsUpGwM4b7 80FuyfECzzDB1rnwU
- Bridge, D. J., Chian, J. Y., & Paller, K. A. (2010). Emotional context at learning systematically biases memory for facial information. Memory & Cognition, 38(2), 125–133. http://doi.org/10.3758/MC.38.2.125
- Bullock, J. G. (2011). Elite influence on public opinion in an informed electorate. American Political Science Review, 105(3), 496–515. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=8368179
- Bush, G., Luu, P., & Posner, M. (2000). Cognitive and emotional influences in anterior cingulate cortex. Trends in Cognitive Sciences, 4(6), 215–222. http://doi.org/10.1016/S1364-6613(00)01483-2
- Cacioppo, J. T., & Visser, P. S. (2003). Political psychology and social neuroscience: Strange bedfellows or comrades in arms? Political Psychology, 24(4), 647–656. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1046/j.1467-9221.2003.00345.x/full

Campbell, A., Converse, P. E., Miller, W. E., & Stokes, D. E. (1980). The American Voter. University of Chicago Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=JeYUrs\_GOcMC&oi =fnd&pg=PA18&dq=Converse+Campbell+miller+stokes&ots=wEfCBHSnqf&sig=xvY VMoYhUweBMkonDUTZ5bFF6c8

Campbell, A., Converse, P. E., Miller, W. E., & Stokes, D. E. (1960). The American voter.

Carey, B., & Belluck, P. (2015). Doubts About Study of Gay Canvassers Rattle the Field -The New York Times. Retrieved from http://www.nytimes.com/2015/05/26/science/maligned-study-on-gay-marriage-isshaking-trust.html
- Carmines, E. G., Ensley, M. J., & Wagner, M. W. (2012). Political Ideology in American Politics: One, Two, or None? The Forum, 10(3).
- Carney, D. R., Jost, J. T., Gosling, S. D., & Potter, J. (2008). The secret lives of liberals and conservatives: Personality profiles, interaction styles, and the things they leave behind. Political Psychology, 29(6), 807–840. http://doi.org/10.1111/j.1467-9221.2008.00668.x
- Carraro, L., Castelli, L., & Macchiella, C. (2011). The automatic conservative: ideologybased attentional asymmetries in the processing of valenced information. PloS One, 6(11), e26456. http://doi.org/10.1371/journal.pone.0026456
- Carver, C. S., & Harmon-Jones, E. (2009). Anger is an approach-related affect: evidence and implications. Psychological Bulletin, 135(2), 183. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/2009-02580-003
- Carver, C. S., & White, T. L. (1994). Behavioral inhibition, behavioral activation, and affective responses to impending reward and punishment: The BIS/BAS Scales. Journal of Personality and Social Psychology, 67(2), 319–333. http://doi.org/10.1037/0022-3514.67.2.319
- Castano, E., Leidner, B., Bonacossa, A., Nikkah, J., Perrulli, R., Spencer, B., & Humphrey, N. (2011). Ideology, fear of death, and death anxiety. Political Psychology, 32(4), 601–621. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1467-9221.2011.00822.x/full
- Charney, E. (2008). Genes and ideologies. Perspectives on Politics, 6(02), 299–319. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S1537592708080626
- Charney, E., & English, W. (2012). Candidate genes and political behavior. American Political Science Review, 106(01), 1–34. http://doi.org/10.1017/S0003055411000554
- Claassen, R. L., & Highton, B. (2009). Policy polarization among party elites and the significance of political awareness in the mass public. Political Research Quarterly, 62(3), 538–551. Retrieved from http://prq.sagepub.com.flagship.luc.edu/content/62/3/538.short

- Coan, J. A., & Allen, J. J. B. (2003). The state and trait nature of frontal EEG asymmetry in emotion. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/2003-02750-016
- Coan, J. A., Allen, J. J. B., & McKnight, P. E. (2006). A capability model of individual differences in frontal EEG asymmetry. Biological Psychology, 72(2), 198–207. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S030105110500 1651
- Coan, J. A., & Allen, J. J. B. J. (2004). Frontal EEG asymmetry as a moderator and mediator of emotion. Biological Psychology, 67(1), 7–50. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S030105110400 0316
- Coan, J. A., & Allen, J. J. B. J. B. (2003). Frontal EEG asymmetry and the behavioral activation and inhibition systems. Psychophysiology, 40(1), 106–114. http://doi.org/10.1111/1469-8986.00011
- Cole, S., Balcetis, E., & Dunning, D. (2012). Affective Signals of Threat Increase Perceived Proximity. Psychological Science. http://doi.org/10.1177/0956797612446953
- Conover, P. J., & Feldman, S. (1981). The Origins and Meaning of Liberal / Conservative Self-Identifications. American Journal of Political Science, 25(4), 617–645. http://doi.org/10.1007/s11109-010-9145-6
- Converse, P. E. (2006). The Nature of Belief Systems in Mass Publics (1964). Critical Review: A Journal of Politics and Society, 18(1-3), 1–74. http://doi.org/10.1080/08913810608443650
- Converse, P. E. (2006). Democratic theory and electoral reality. Critical Review, 18(September 2012), 297–329. http://doi.org/10.1080/08913810608443662
- Converse, P. E. (1964). The Nature of Belief Systems in Mass Publics." In David Apter, ed., Ideology and Discontent. New York: Free Press.
- Crawford, J. T., & Pilanski, J. M. (2013). Political Intolerance, Right and Left. Political Psychology, no-no. http://doi.org/10.1111/j.1467-9221.2012.00926.x
- Critchley, H. D., Mathias, C. J., & Dolan, R. J. (2001). Neural activity in the human brain relating to uncertainty and arousal during anticipation. Neuron, 29(2), 537–545. http://doi.org/10.1016/S0896-6273(01)00225-2

- Davidson, R. J. (1998). Anterior electrophysiological asymmetries, emotion, and depression: Conceptual and methodological conundrums. Psychophysiology, 35(05), 607–614. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0048577298000134
- Davidson, R. J. (1998). Affective style and affective disorders: Perspectives from affective neuroscience. Cognition & Emotion, 12(3), 307–330. Retrieved from http://www.tandfonline.com.flagship.luc.edu/doi/abs/10.1080/026999398379628
- Davidson, R. J. (1993). Cerebral asymmetry and emotion: Conceptual and methodological conundrums. Cognition & Emotion. http://doi.org/10.1080/02699939308409180
- Dawes, R. M., & Thaler, R. H. (1988). Anomalies: cooperation. The Journal of Economic Perspectives, 2(3), 187–197. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/1942822
- Dawes, R. M., & Thaler, R. H. (1988). Anomalies. Journal of Economic Perspectives, 2(3), 187–197. Retrieved from http://content.ebscohost.com.flagship.luc.edu/ContentServer.asp?T=P&P=AN&K=4 432074&S=R&D=bth&EbscoContent=dGJyMNLr40SeprI4wtvhOLCmr0qep69SsK64T LCWxWXS&ContentCustomer=dGJyMPGusky2p65JuePfgeyx44Dt6fIA
- de St Aubin, E. (1996). Personal ideology polarity: Its emotional foundation and its manifestation in individual value systems, religiosity, political orientation, and assumptions concerning human nature. Journal of Personality and Social Psychology, 71(1), 152.
- DeBruine, L. M., Jones, B. C., Tybur, J. M., Lieberman, D., & Griskevicius, V. (2010).
   Women's preferences for masculinity in male faces are predicted by pathogen disgust, but not by moral or sexual disgust. Evolution and Human Behavior, 31(1), 69–74. http://doi.org/10.1016/j.evolhumbehav.2009.09.003
- DeCanio, S. (2006). Mass Opinion and American Political Development. Critical Review, 18(1-3), 143–155. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217262554/abstra ct?accountid=12163
- Delli Carpini, M. X., & Ketter, S. (1993). Measuring political knowledge: Putting first things first. American Journal of Political Science, 37(4), 1179–1206. http://doi.org/10.2307/2111549

- Demaree, H. A., Everhart, D. E., Youngstrom, E. A., & Harrison, D. W. (2005). Brain Lateralization of Emotional Processing: Historical Roots and a Future Incorporating "Dominance." Behavioral and Cognitive Neuroscience Reviews, 4(1), 3–20. http://doi.org/10.1177/1534582305276837
- Dennis, T. A., & Solomon, B. (2010). Frontal EEG and emotion regulation: Electrocortical activity in response to emotional film clips is associated with reduced mood induction and attention interference effects. Biological Psychology, 85(3), 456–464. http://doi.org/16/j.biopsycho.2010.09.008
- Depue, R. A., & Collins, P. F. (1999). Neurobiology of the structure of personality: Dopamine, facilitation of incentive motivation, and extraversion. Behavioral and Brain Sciences, 22(3), 491–517. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=31178
- Depue, R. A., & Iacono, W. G. (1989). Neurobehavioral aspects of affective disorders. Annual Review of Psychology, 40(1), 457–492. Retrieved from http://www.annualreviews.org.flagship.luc.edu/doi/abs/10.1146/annurev.ps.40.02 0189.002325
- Devinsky, O., Morrell, M. J., & Vogt, B. a. (1995). Contributions of anterior cingulate cortex to behaviour. Brain : A Journal of Neurology, 118 (Pt 1, 279–306. http://doi.org/10.1093/brain/118.1.279
- Dodd, M. D., Balzer, A., Jacobs, C. M., Gruszczynski, M. W., Smith, K. B., & Hibbing, J. R. (2012). The political left rolls with the good and the political right confronts the bad: connecting physiology and cognition to preferences. Philosophical Transactions of the Royal Society B: Biological Sciences, 367(1589), 640–649. http://doi.org/10.1098/rstb.2011.0268
- Downs, A. (1957). An economic theory of democracy. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=Sg0sEpeAxPoC&oi= fnd&pg=PA344&dq=Anthony+Downs+Economic&ots=zuzfdjnenN&sig=Wb1XlucEiFl U7wf5uoXLseX3SFM
- Druckman, J. N., & McDermott, R. (2008). Emotion and the Framing of Risky Choice. Political Behavior, 30(3), 297–321. http://doi.org/10.1007/s11109-008-9056-y
- Eidelman, S., Crandall, C. S., Goodman, J. A., & Blanchar, J. C. (2012). Low-Effort Thought Promotes Political Conservatism. Personality and Social Psychology Bulletin, 38(6), 808–820. http://doi.org/10.1177/0146167212439213

- Ein-Dor, T., Mikulincer, M., & Shaver, P. R. (2011, September). Effective Reaction to Danger: Attachment Insecurities Predict Behavioral Reactions to an Experimentally Induced Threat Above and Beyond General Personality Traits. Social Psychological and Personality Science. http://doi.org/10.1177/1948550610397843
- Ein-Dor, T., & Tal, O. (2012). Scared saviors: Evidence that people high in attachment anxiety are more effective in alerting others to threat. European Journal of Social Psychology, 42(6), 667–671. http://doi.org/10.1002/ejsp.1895
- Ellis, C., & Stimson, J. A. (2012). Ideology in America. Cambridge University Press. http://doi.org/10.1017/CBO9781139094009
- Ellis, C., & Stimson, J. A. (2009). Symbolic ideology in the American electorate. Electoral Studies, 28(3), 388–402. http://doi.org/10.1016/j.electstud.2009.05.010
- Evans, G., Heath, A., & Lalljee, M. (1996). Measuring Left-Right and Libertarian-Authoritarian Values in the British Electorate. The British Journal of Sociology, 47(1), 93–112. http://doi.org/10.2307/591118
- Federico, C. M., Hunt, C. V., & Ergun, D. (2009). Political expertise, social worldviews, and ideology: Translating "competitive jungles" and "dangerous worlds" into ideological reality. Social Justice Research, 22(2), 259–279. Retrieved from http://www.springerlink.com.flagship.luc.edu/index/91403R60881342L3.pdf
- Fehr, E., & Fischbacher, U. (2004). Third-party punishment and social norms. Evolution and Human Behavior, 25(2), 63–87. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S109051380400 0054
- Fehr, E., & Gachter, S. (2000). Fairness and Retaliation: The Economics of Reciprocity. SSRN Electronic Journal. http://doi.org/10.2139/ssrn.229149
- Feldman, S. (2003). Values, ideology, and the structure of political attitudes. Oxford Handbook of Political Psychology, 477–508.
- Field, T., Fox, N. A., Pickens, J., & Nawrocki, T. (1995). Relative right frontal EEG activation in 3-to 6-month-old infants of" depressed" mothers. Developmental Psychology, 31(3), 358. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/dev/31/3/358/
- Fiorina, M. P., Abrams, S. A., & Pope, J. C. (2008). Polarization in the American public: Misconceptions and misreadings. The Journal of Politics, 70(02), 556–560.

Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S002238160808050X

- Fiorina, M. P., & Abrams, S. J. (2008). Political polarization in the American public. Annu. Rev. Polit. Sci., 11, 563–588. Retrieved from http://www.annualreviews.org.flagship.luc.edu/doi/abs/10.1146/annurev.polisci.1 1.053106.153836
- Fiorina, M. P., Abrams, S. J., & Pope, J. (2005). Culture war? Pearson Education. Retrieved from http://www.uvm.edu/~dguber/POLS125/articles/fiorina2.pdf
- Fishkin, J. S. (2006). Beyond Polling Alone: The Quest for an Informed Public. Critical Review, 18(1-3), 157–165. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217270204/abstra ct?accountid=12163
- Fowler, J. H., Baker, L. A., & Dawes, C. T. (2008). Genetic variation in political participation. American Political Science Review, 102(02), 233–248. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0003055408080209
- Fowler, J. H., & Dawes, C. T. (2008). Two genes predict voter turnout. Journal of Politics, 70(3), 579–94. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=1927204
- Fowler, J. H., & Schreiber, D. (2008). Biology, Politics, and the Emerging Science of Human Nature. Science, 322(5903). http://doi.org/10.1126/science.1158188
- Fowles, D. C. (1988). Psychophysiology and psychopathology: A motivational approach. Psychophysiology.
- Fox, N. A., Schmidt, L. A., Calkins, S. D., Rubin, K. H., & Coplan, R. J. (1996). The role of frontal activation in the regulation and dysregulation of social behavior during the preschool years. Development and Psychopathology, 8(01), 89–102. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0954579400006982
- Fraley, R. C., Griffin, B. N., Belsky, J., & Roisman, G. I. (2012). Developmental Antecedents of Political Ideology A Longitudinal Investigation From Birth to Age 18 Years. Psychological Science, 23(11), 1425–1431. http://doi.org/10.1177/0956797612440102

- Friedman, J. (2006). Democratic Competence in Normative and Positive Theory: Neglected Implications of "the Nature of Belief Systems in Mass Publics." Critical Review, 18(1-3), R1–R43. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217259846/abstra ct?accountid=12163
- Friese, M., Fishman, S., Beatson, R., Sauerwein, K., & Rip, B. (2009). Whose Fault Is it Anyway? Political Orientation, Attributions of Responsibility, and Support for the War in Iraq. Social Justice Research, 22(2-3), 280–297. http://doi.org/10.1007/s11211-009-0095-2
- Funk, C., Smith, K. B., Alford, J., Hibbing, M., Hatemi, P., & Hibbing, J. R. (2010). Toward a modern view of political man: Genetic and environmental sources of political orientations and participation. Retrieved from http://papers.ssrn.com.flagship.luc.edu/sol3/papers.cfm?abstract\_id=1643954
- Gable, S. L., Reis, H. T., & Elliot, A. J. (2000). Behavioral activation and inhibition in everyday life. Journal of Personality and Social Psychology, 78(6), 1135.
- Gastil, J., Braman, D., Kahan, D., & Slovic, P. (2011). The Cultural Orientation of Mass Political Opinion. PS: Political Science & Politics, 44(04), 711–714. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S1049096511001326
- Gerring, J. (1997). Ideology: A definitional analysis. Political Research Quarterly, 50(4), 957–994. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/448995
- Gintis, H. (2000). Strong reciprocity and human sociality. Journal of Theoretical Biology, 206(2), 169–179. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S002251930092 1118
- GINTIS, H. (2003). Solving the Puzzle of Prosociality. Rationality and Society, 15(2), 155– 187. http://doi.org/10.1177/1043463103015002001
- GINTIS, H. (2003). The Hitchhiker's Guide to Altruism: Gene-culture Coevolution, and the Internalization of Norms. Journal of Theoretical Biology, 220(4), 407–418. http://doi.org/10.1006/jtbi.2003.3104
- GINTIS, H., Henrich, J., Bowles, S., Boyd, R., & Fehr, E. (2008). Strong Reciprocity and the Roots of Human Morality. Social Justice Research, 21(2), 241–253. http://doi.org/10.1007/s11211-008-0067-y

- Graber, D. A. (2006). Government by the People, for the People-Twenty-First Century Style. Critical Review, 18(1-3), 167–178. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217259963/abstra ct?accountid=12163
- Graham, J., Haidt, J., & Nosek, B. A. (2009). Liberals and conservatives rely on different sets of moral foundations. Journal of Personality and Social Psychology, 96(5), 1029. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/2009-05192-002
- Gray, J. A. (1981). A critique of Eysenck's model of personality. In A Model for Personality (pp. 246–276). http://doi.org/10.1007/978-3-642-67783-0\_8
- Gray, J. A. (1972). The psychophysiological nature of Introversion-Extraversion: a modification of Eysenck's theory. In The biological bases of individual behaviour (pp. 182–205).
- Gray, J. A. (1990). Brain Systems that Mediate both Emotion and Cognition. Cognition & Emotion, 4(3), 269–288. http://doi.org/10.1080/02699939008410799
- Gray, J. A. (1982). The Neuropsychology of Anxiety an Inquiry into the Functions of the Septo-Hippocampal System. Behavioral and Brain Sciences, 5, 469–534. http://doi.org/10.1017/S0140525X00013170
- Gray, J. A. (1988). The Psychology of Fear and Stress, (Vol. 5). Cambridge university press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&Ir=&id=nww5AAAAIAAJ&oi =fnd&pg=PR6&dq=The+psychology+of+fear+and+stress&ots=rmAu3rSeuZ&sig=nQf 8Y2KoS5MnjYpkeIxIKMeU5KI
- Gray, J. A., & Gray, J. A. (2008). Brain Systems that Mediate both Emotion and Cognition
   Brain Systems that Mediate both Emotion and Cognition. Cognition & Emotion,
   4(August 2015), 269–288. http://doi.org/10.1080/02699939008410799
- Gray, J. A., & McNaughton, N. (2000). The Neuropsychology of Anxiety, Second Edition.British Journal of Psychology (Vol. 69).http://doi.org/10.1093/acprof:oso/9780198522713.001.0001
- Green, D. P., & Shapiro, I. (1994). Pathologies of rational choice theory: A critique of applications in political science. Cambridge Univ Press. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=6296592

- Green, J. (2007). When Voters and Parties Agree: Valence Issues and Party Competition. Political Studies, 55(3), 629–655. http://doi.org/10.1111/j.1467-9248.2007.00671.x
- Green, S. B., Lissitz, R. W., & Mulaik, S. A. (1977). Limitations of Coefficient Alpha as an Index of Test Unidimensionality1. Educational and Psychological Measurement, 37(4), 827–838. http://doi.org/10.1177/001316447703700403
- Greenberg, J., Pyszczynski, T., & Solomon, S. (1986). The causes and consequences of a need for self-esteem: A terror management theory. Public Self and Private Self, 189, 212.
- Haidt, J., & Graham, J. (2007). When morality opposes justice: Conservatives have moral intuitions that liberals may not recognize. Social Justice Research, 20(1), 98–116.
   Retrieved from http://www.springerlink.com.flagship.luc.edu/index/T11828205JT42001.pdf
- Haidt, J., Graham, J., & Joseph, C. (2009). Above and below left–right: Ideological narratives and moral foundations. Psychological Inquiry, 20(2-3), 110–119.
  Retrieved from http://www.tandfonline.com.flagship.luc.edu/doi/abs/10.1080/104784009030285 73
- Hajcak, G., & Foti, D. (2008). Errors are aversive: Defensive motivation and the errorrelated negativity: Research report. Psychological Science, 19(2), 103–108. http://doi.org/10.1111/j.1467-9280.2008.02053.x
- Hajcak, G., McDonald, N., & Simons, R. F. (2003). Anxiety and error-related brain activity. Biological Psychology, 64(1-2), 77–90. http://doi.org/10.1016/S0301-0511(03)00103-0
- Halperin, E., Porat, R., Tamir, M., & Gross, J. J. (2012). Can Emotion Regulation Change Political Attitudes in Intractable Conflicts? From the Laboratory to the Field. Psychological Science. http://doi.org/10.1177/0956797612452572
- Han, H. (2011). The Disappearing Center: Engaged Citizens, Polarization, and American Democracy. Public Opinion Quarterly. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=LVHF3roJPU0C&oi= fnd&pg=PR9&dq=The+disappearing+center:+Engaged+citizens,+polarization,+and+ American+democracy&ots=6U1Z1Uyx0z&sig=YA5ZQWNnloEEjhLVfBJTIP0-LHM
- Hardin, R. (2006). Ignorant Democracy. Critical Review, 18(1-3), 179–195. Retrieved from

http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217255468/abstra ct?accountid=12163

- Harmon-Jones, E., & Sigelman, J. (2001). State anger and prefrontal brain activity: evidence that insult-related relative left-prefrontal activation is associated with experienced anger and aggression. Journal of Personality and Social Psychology, 80(5), 797. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/psp/80/5/797/
- Harmon-Jones, E. (2003). Anger and the behavioral approach system. Personality and Individual Differences, 35(5), 995–1005. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S019188690200 3136
- Harmon-Jones, E., & Allen, J. J. B. J. (1998). Anger and frontal brain activity: EEG asymmetry consistent with approach motivation despite negative affective valence. Journal of Personality and Social Psychology, 74(5), 1310. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/psp/74/5/1310/
- Harmon-Jones, E., & Allen, J. J. B. J. (1997). Behavioral activation sensitivity and resting frontal EEG asymmetry: covariation of putative indicators related to risk for mood disorders. Journal of Abnormal Psychology, 106(1), 159. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/1997-02518-016
- Hatemi, P. K., & McDermott, R. (2011). Man is by nature a political animal: evolution, biology, and politics. University of Chicago Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=2Nem4knFLk8C&oi =fnd&pg=PT3&dq=Hatemi+and+McDermott&ots=04RNK\_B-VD&sig=WbF4Uak1FPbKljR46LYFzxHYAzU
- Hatemi, P. K., Alford, J. R., Hibbing, J. R., Martin, N. G., & Eaves, L. J. (2009). Is there a "party" in your genes? Political Research Quarterly, 62(3), 584–600. Retrieved from http://prq.sagepub.com.flagship.luc.edu/content/62/3/584.short
- Hatemi, P. K., Eaves, L., & McDermott, R. (2012). It's the end of ideology as we know it. Journal of Theoretical Politics, 24(3). http://doi.org/10.1177/0951629812437749
- Hatemi, P. K., Hibbing, J. R., Medland, S. E., Keller, M. C., Alford, J. R., Smith, K. B., ... Eaves, L. J. (2010). Not by twins alone: Using the extended family design to investigate genetic influence on political beliefs. American Journal of Political Science, 54(3), 798–814. Retrieved from

http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1540-5907.2010.00461.x/full

- Hatemi, P. K., & McDermott, R. (2011). Evolution as a theory for political behavior. Man Is by Nature a Political Animal: Evolution, Biology, and Politics. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&Ir=&id=2Nem4knFLk8C&oi =fnd&pg=PT33&dq=boundary+between+the+natural+sciences+on+one+side+and+ the+humanities+on+and+humanistic+social+sciences+on+the+other+is+not+an+epi stemological+fault+line,+but+a+broad+domain+of+poorly+understood+material+p henomena+that+invites+cooperative+exploration+from+b&ots=04SKO4D0WB&sig =V83LyqmSnuiQUvtj5ZVX-YMT5wQ
- Hatemi, P. K., & McDermott, R. (2012). The genetics of politics: discovery, challenges, and progress. Trends in Genetics, 28(10), 525–533. http://doi.org/10.1016/j.tig.2012.07.004
- Hatemi, P. K., McDermott, R., Eaves, L. J., Kendler, K. S., & Neale, M. C. (2013). Fear as a Disposition and an Emotional State: A Genetic and Environmental Approach to Out-Group Political Preferences. American Journal of Political Science. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/ajps.12016/full
- Hayduk, L. A., Ratner, P. A., Johnson, J. L., & Bottorff, J. L. (1995). Attitudes, ideology, and the factor model. Political Psychology, 479–507. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/3792223
- Heath, A., Evans, G., & Martin, J. (1994). The Measurement of Core Beliefs and Values: The Development of Balanced Socialist/Laissez Faire and Libertarian/Authoritarian Scales. British Journal of Political Science, 24(01), 115–132. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0007123400006815
- Heller, W., Nitschke, J. B., Etienne, M. A., & Miller, G. A. (1997). Patterns of regional brain activity differentiate types of anxiety. Journal of Abnormal Psychology, 106(3), 376. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/abn/106/3/376/
- Helzer, E., & Pizarro, D. A. (2011). Dirty liberals! Reminders of cleanliness promote conservative political and moral attitudes. Psychological Science, 22, 517–522.
- Henrich, J., Boyd, R., Bowles, S., Camerer, C., Fehr, E., Gintis, H., ... Ensminger, J. (2005). "
   Economic man" in cross-cultural perspective: Behavioral experiments in 15 small-scale societies. Behavioral and Brain Sciences, 28(6), 795–814. Retrieved from

http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fulltextid=364679

- Henrich, J., Boyd, R., Bowles, S., Camerer, C., Fehr, E., Gintis, H., & McElreath, R. (2001). In search of homo economicus: behavioral experiments in 15 small-scale societies. American Economic Review, 73–78. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2677736
- Hernandez, I., & Preston, J. L. (2013). Disfluency disrupts the confirmation bias. Journal of Experimental Social Psychology, 49(1), 178–182. http://doi.org/10.1016/j.jesp.2012.08.010
- Herrmann, M. J., Römmler, J., Ehlis, A. C., Heidrich, A., & Fallgatter, A. J. (2004). Source localization (LORETA) of the error-related-negativity (ERN/Ne) and positivity (Pe). Cognitive Brain Research, 20(2), 294–299. http://doi.org/10.1016/j.cogbrainres.2004.02.013
- Hetherington, M. J. (2001). Resurgent mass partisanship: The role of elite polarization. American Political Science Review, 95(3), 619–632. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=93848
- Hibbing, J. R., Smith, K. B., & Alford, J. R. (2014). Predisposed: Liberals, conservatives, and the biology of political differences. Routledge.
- Hibbing, J. R., Smith, K. B., & Alford, J. R. (2014). Differences in negativity bias underlie variations in political ideology. The Behavioral and Brain Sciences, 37(3), 297–307. http://doi.org/10.1017/S0140525X13001192
- Hirsh, J. B., & Inzlicht, M. (2008). The devil you know: Neuroticism predicts neural response to uncertainty. Psychological Science, 19(10), 962–967. http://doi.org/10.1111/j.1467-9280.2008.02183.x
- Holroyd, C. B., & Coles, M. G. H. (2002). The Neural Basis of Human Error Processing: Reinforcement Learning, Dopamine, and the Error-Related Negativity. Psychological Review, 109(4), 679–709. http://doi.org/10.1037//0033-295X.109.4.679
- Holroyd, C. B., & Yeung, N. (2012). Motivation of extended behaviors by anterior cingulate cortex. Trends in Cognitive Sciences, 16(2), 122–128. http://doi.org/10.1016/j.tics.2011.12.008

- Huddy, L., Mason, L., & Aarøe, L. (2015). Expressive Partisanship: Campaign Involvement, Political Emotion, and Partisan Identity. American Political Science Review, 109(1), 1–17. http://doi.org/10.1017/S0003055414000604
- Huntington, S. P. (1957). Conservatism as an Ideology. The American Political Science Review, 51(2), 454–473. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/1952202
- Hussey, L. S. (2012). Polarized Politics and Citizen Disengagement The Role of Belief Systems. American Politics Research, 40(1), 85–115. http://doi.org/10.1177/1532673X11416441
- Inzlicht, M., McGregor, I., Hirsh, J. B., & Nash, K. (2009). Neural markers of religious conviction. Psychological Science, 20(3), 385–392. http://doi.org/10.1111/j.1467-9280.2009.02305.x
- Iyengar, S., Sood, G., & Lelkes, Y. (2012). Affect, Not Ideology A Social Identity Perspective on Polarization. Public Opinion Quarterly. Retrieved from http://poq.oxfordjournals.org.flagship.luc.edu/content/early/2012/09/15/poq.nfs0 38.short
- Iyer, R., Koleva, S., Graham, J., Ditto, P., & Haidt, J. (2012). Understanding libertarian morality: The psychological dispositions of self-identified libertarians. PloS One, 7(8), e42366. Retrieved from http://dx.plos.org/10.1371/journal.pone.0042366
- Jackson, P. L., Meltzoff, A. N., & Decety, J. (2005). How do we perceive the pain of others? A window into the neural processes involved in empathy. NeuroImage, 24(3), 771–779. http://doi.org/10.1016/j.neuroimage.2004.09.006
- Jacoby, W. G. (2009). Ideology and vote choice in the 2004 election. Electoral Studies, 28(4), 584–594. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S026137940900 0614
- Jacoby, W. G. (1995). The structure of ideological thinking in the American electorate. American Journal of Political Science, 314–335. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2111615
- Jennings, M. K. (1992). Ideological Thinking Among Mass Publics and Political Elites. Public Opinion Quarterly, 56(4), 419–441. http://doi.org/10.1086/269335

- Jennings, M. K., & Niemi, R. G. (1978). The Persistence of Political Orientations: An Over-Time Analysis of Two Generations. British Journal of Political Science, 8(03), 333– 363. http://doi.org/10.1017/S000712340000140X
- Joel, S., Burton, C. M., & Plaks, J. E. (2013). Conservatives Anticipate and Experience Stronger Emotional Reactions to Negative Outcomes than Liberals. Journal of Personality, n/a–n/a. http://doi.org/10.1111/jopy.12031
- Johnston, C. D. (2011). The Motivated Formation of Economic Preferences. Retrieved from https://dspace.sunyconnect.suny.edu/handle/1951/56029
- Jost, J. T., Nosek, B. A., & Gosling, S. D. (2008). Ideology: Its resurgence in social, personality, and political psychology. Perspectives on Psychological Science, 3(2), 126–136. Retrieved from http://pps.sagepub.com.flagship.luc.edu/content/3/2/126.short
- Jost, J. T. (2006). The end of the end of ideology. American Psychologist, 61(7), 651.
- Jost, J. T. (2009). "Elective Affinities": On the Psychological Bases of Left–Right Differences. Psychological Inquiry, 20(2-3), 129–141. http://doi.org/10.1080/10478400903028599
- Jost, J. T. (2007). Coda--after "The end of the end of ideology". The American Psychologist, 62(9), 1077–1080. http://doi.org/10.1037/0003-066X.62.9.1077
- Jost, J. T. (2011). That's Incredible! Science, 333(6047), 1222–1223. http://doi.org/10.1126/science.1209161
- Jost, J. T., Federico, C. M., & Napier, J. L. (2009). Political ideology: its structure, functions, and elective affinities. Annual Review of Psychology, 60(1), 307–337. http://doi.org/10.1146/annurev.psych.60.110707.163600
- Jost, J. T., Glaser, J., Kruglanski, A. W., & Sulloway, F. J. (2003). Political conservatism as motivated social cognition. Psychological Bulletin, 129(3), 339–375. http://doi.org/10.1037/0033-2909.129.3.339
- Jost, J., & Amodio, D. (2012). Political ideology as motivated social cognition: Behavioral and neuroscientific evidence. Motiv Emot, 36, 55–64. Retrieved from http://www.springerlink.com.flagship.luc.edu/index/N1V616836K016X58.pdf
- Kahan, D. (2012). Ideology, Motivated Reasoning, and Cognitive Reflection: An Experimental Study. Retrieved from http://papers.ssrn.com/abstract=2182588

- Kahn, R., Misra, K., & Singh, V. (n.d.). Ideology and Brand Consumption. Retrieved from http://faculty.london.edu/kmisra/assets/documents/Conservatism\_MainText.pdf
- Kahneman, D., & Thaler, R. (1991). Economic analysis and the psychology of utility: applications to compensation policy. The American Economic Review, 341–346. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2006882
- Kam, C. D., Wilking, J. R., & Zechmeister, E. J. (2007). Beyond the "narrow data base": Another convenience sample for experimental research. Political Behavior, 29(4), 415–440. http://doi.org/10.1007/s11109-007-9037-6
- Kanai, R., Feilden, T., Firth, C., & Rees, G. (2011). Political Orientations Are Correlated with Brain Structure in Young Adults. Current Biology, 21(8), 677–680. http://doi.org/10.1016/j.cub.2011.03.017
- Kanske, P., & Kotz, S. a. (2011). Emotion triggers executive attention: Anterior cingulate cortex and amygdala responses to emotional words in a conflict task. Human Brain Mapping, 32(2), 198–208. http://doi.org/10.1002/hbm.21012
- Kerlinger, F. N. (1984). Liberalism and Conservatism: The nature and structure of social attitudes. Lawrence Erlbaum Assoc Incorporated.
- Kerns, J. G. (2004). Anterior Cingulate Conflict Monitoring and Adjustments in Control. Science, 303(5660), 1023–1026. http://doi.org/10.1126/science.1089910
- Kesebir, P., Phillips, E., Anson, J., Pyszczynski, T., & Motyl, M. (n.d.). Ideological Consistency Across the Political Spectrum: Liberals are More Consistent But Conservatives Become More Consistent When Coping with Existential Threat. Rochester, NY. Retrieved from http://papers.ssrn.com/abstract=2215306
- Khan, R., Misra, K., & Singh, V. (2013). Ideology and Brand Consumption. Psychological Science. http://doi.org/10.1177/0956797612457379
- Kinder, D. R. (1998). Opinion and action in the realm of politics. Retrieved from http://psycnet.apa.org.flagship.luc.edu/psycinfo/1998-07091-034
- Kinder, D. R. (2006). Belief Systems Today. Critical Review, 18(1-3), 197–216. http://doi.org/10.1080/08913810608443657
- Klugel, J., Smith, E. R., Kluegel, J. R., & Smith, E. R. (1986). Beliefs about inequality: Americans' views of what is and what ought to be. New York, NY: Aline De Grader. Transaction Publishers.

- Knight, K. (1999). Liberalism and conservatism. In Measures of political attitudes (pp. 59–158). http://doi.org/10.1016/0191-8869(85)90101-1
- Knight, K. (2006). Transformations of the Concept of Ideology in the Twentieth Century. American Political Science Review, 100(4), 619–626. http://doi.org/10.1017/S0003055406062502
- Koleva, S. P., & Rip, B. (2009). Attachment Style and Political Ideology: A Review of Contradictory Findings. Social Justice Research, 22(2-3), 241–258. http://doi.org/10.1007/s11211-009-0099-y
- Kritzer, H. M. (1978). Ideology and American Political Elites. Public Opinion Quarterly, 42(4), 484–502. http://doi.org/10.1086/268475
- Kruger, D. J. (2003). Evolution and altruism: Combining psychological mediators with naturally selected tendencies. Evolution and Human Behavior, 24(2), 118–125. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S109051380200 1563
- Kunda, Z. (1990). The case for motivated reasoning. Psychological Bulletin, 108(3), 480– 498. http://doi.org/10.1037/0033-2909.108.3.480
- Kurzban, R., & Houser, D. (2005). Experiments investigating cooperative types in humans: A complement to evolutionary theory and simulations. Proceedings of the National Academy of Sciences of the United States of America, 102(5), 1803–1807.
   Retrieved from http://www.pnas.org.flagship.luc.edu/content/102/5/1803.short
- Lahti, D. C., & Weinstein, B. S. (2005). The better angels of our nature: group stability and the evolution of moral tension. Evolution and Human Behavior, 26(1), 47–63. http://doi.org/10.1016/j.evolhumbehav.2004.09.004
- LaMarre, H. L., Landreville, K. D., & Beam, M. A. (2009). The Irony of Satire Political Ideology and the Motivation to See What You Want to See in The Colbert Report. The International Journal of Press/Politics, 14(2), 212–231. http://doi.org/10.1177/1940161208330904
- Lavin, C., Melis, C., Mikulan, E., Gelormini, C., Huepe, D., & Ibañez, A. (2013). The anterior cingulate cortex: an integrative hub for human socially-driven interactions. Frontiers in Neuroscience, 7(May), 64. http://doi.org/10.3389/fnins.2013.00064

- Lavine, H., Lodge, M., & Freitas, K. (2005). Threat, authoritarianism, and selective exposure to information. Political Psychology, 26(2), 219–244. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1467-9221.2005.00416.x/abstract
- Lavine, H., Lodge, M., Polichak, J., & Taber, C. (2002). Explicating the black box through experimentation: Studies of authoritarianism and threat. Political Analysis, 10(4), 343–361. Retrieved from http://pan.oxfordjournals.org.flagship.luc.edu/content/10/4/343.short
- Layman, G. C., & Carsey, T. M. (2002). Party Polarization and "Conflict Extension" in the American Electorate. American Journal of Political Science, 46(4), 786–000.
- Lazarsfeld, P. F., Berelson, B., & Gaudet, H. (1944). The people's choice. New York, 332. Retrieved from http://ecsocman.hse.ru/text/19161674/
- Lazarus, R. S. (1982). Thoughts on the relations between emotion and cognition. American Psychologist, 37(9), 1019. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/amp/37/9/1019/
- Leone, L., & Chirumbolo, A. (2008). Conservatism as motivated avoidance of affect: Need for affect scales predict conservatism measures. Journal of Research in Personality, 42(3), 755–762. http://doi.org/10.1016/j.jrp.2007.08.001
- Levelt: fraud detected in 55 publications | Univers. (2012). Retrieved from https://universonline.nl/2012/11/28/levelt-report-fraud-detected-in-55publications
- Levendusky, M. (2009). The partisan sort: how liberals became Democrats and conservatives became Republicans. University of Chicago Press.
- Levendusky, M. S. (2010). Clearer cues, more consistent voters: A benefit of elite polarization. Political Behavior, 32(1), 111–131. Retrieved from http://www.springerlink.com.flagship.luc.edu/index/6h76637512000649.pdf
- Leventhal, H., & Scherer, K. (1987). The relationship of emotion to cognition: A functional approach to a semantic controversy. Cognition and Emotion, 1(1), 3–28. Retrieved from http://www.tandfonline.com.flagship.luc.edu/doi/abs/10.1080/026999387084083 61

- Lewis-Beck, Michael S., Norpoth, Helmut, and Jacoby, W. (2008). American Voter Revisited. In Section V. University of Michigan Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=5t3\_S8zoYnUC&oi= fnd&pg=PR9&dq=Lewis-Beck+2008+American+Voter&ots=f4BCFCONst&sig=efLFY9-5roEFEhDaPUhmOfS6VQ8
- Lieberman, D. L., Tybur, J. M., & Latner, J. D. (2012). Disgust Sensitivity, Obesity Stigma, and Gender: Contamination Psychology Predicts Weight Bias for Women, Not Men. Obesity, 20(9), 1803–1814. http://doi.org/10.1038/oby.2011.247
- Lieberman, M. D., Schreiber, D., & Ochsner, K. N. (2003). Is political cognition like riding a bicycle? How cognitive neuroscience can inform research on political thinking. Political Psychology, 24(4), 681–704. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1046/j.1467-9221.2003.00347.x/full
- Lindeman, M., Svedholm, A. M., Riekki, T., Raij, T., & Hari, R. (2012). Is it Just a Brick Wall or a Sign from the Universe? An fMRI Study of Supernatural Believers and Skeptics. Social Cognitive and Affective Neuroscience. http://doi.org/10.1093/scan/nss096
- Lupia, A. (2006). How Elitism Undermines the Study of Voter Competence. Critical Review, 18(1-3), 217–232. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217255369/abstra ct?accountid=12163
- Mallery, P. (1999). SPSS for Windows Step by Step: A simple guide and reference. Needham Heights, MA: Allyn & Bacon.
- Marcus, G. E., MacKuen, M., & Neuman, W. R. (2011). Parsimony and Complexity: Developing and Testing Theories of Affective Intelligence. Political Psychology, 32(2), 323–336. http://doi.org/10.1111/j.1467-9221.2010.00806.x
- Marcus, G. E., Neuman, W. R., & MacKuen, M. (2000). Affective Intelligence and Political Judgment. University of Chicago Press. Retrieved from http://books.google.com/books?id=XkyjBNvIMKQC
- Marcus, G. E., Wood, S. L., & Theiss-Morse, E. (1998). Linking neuroscience to political intolerance and political judgment. Politics and the Life Sciences, 165–178. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/4236433

- Martin, A. D., & Quinn, K. M. (2002). Dynamic ideal point estimation via Markov chain Monte Carlo for the US Supreme Court, 1953–1999. Political Analysis, 10(2), 134– 153. Retrieved from http://pan.oxfordjournals.org.flagship.luc.edu/content/10/2/134.short
- Martin, K. C. (2014). What are the most common American political insults? | OxfordWords blog. Retrieved May 23, 2016, from http://blog.oxforddictionaries.com/2014/11/political-insults/
- Masters, R. D. (1983). The Biological Nature of the State. World Politics, 35(02), 161– 193. http://doi.org/10.2307/2010269
- McAdams, D. P., Albaugh, M., Farber, E., Daniels, J., Logan, R. L., & Olson, B. (2008). No Title. Journal of Personality and Social Psychology, 95(4), 978–990. http://doi.org/10.1037/a0012650
- McAdams, D. P., Hanek, K. J., & Dadabo, J. G. (2012). Themes of Self-Regulation and Self-Exploration in the Life Stories of Religious American Conservatives and Liberals. Political Psychology, no–no. http://doi.org/10.1111/j.1467-9221.2012.00933.x
- McAdams, D. P., & Pals, J. L. (2006). A new Big Five: Fundamental principles for an integrative science of personality. American Psychologist, 61(3), 204–217. http://doi.org/10.1037/0003-066X.61.3.204
- McDermott, R. (2004). The feeling of rationality: The meaning of neuroscientific advances for political science. Perspectives on Politics, 2(4), 691–706. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=266159
- McDermott, R. (2009). Mutual Interests The Case for Increasing Dialogue between Political Science and Neuroscience. Political Research Quarterly, 62(3), 571–583. http://doi.org/10.1177/1065912909336273
- McGuire, W. J. (1986). The vicissitudes of attitudes and similar representational constructs in twentieth century psychology. European Journal of Social Psychology, 16(2), 89–130. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1002/ejsp.2420160202/abst ract

- Merelman, R. M. (1969). The Development of Political Ideology: A Framework for the Analysis of Political Socialization. The American Political Science Review, 63(3), 750–767. http://doi.org/10.2307/1954426
- Measuring the Conservatism of the Roberts Court. (2010, July). The New York Times. Retrieved from http://www.nytimes.com/interactive/2010/07/25/us/20100725roberts-graphic.html
- Miller, G. J. (1997). The impact of economics on contemporary political science. Journal of Economic Literature, 35(3), 1173–1204. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2729975
- Minar, D. M. (1961). Ideology and political behavior. Midwest Journal of Political Science, 5(4), 317–331. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2108991
- Nail, P. R., & McGregor, I. (2009). Conservative shift among liberals and conservatives following 9/11/01. Social Justice Research, 22(2), 231–240. Retrieved from http://www.springerlink.com.flagship.luc.edu/index/15030320077N1724.pdf
- Nail, P. R., McGregor, I., Drinkwater, A. E., Steele, G. M., & Thompson, A. W. (2009). Threat causes liberals to think like conservatives. Journal of Experimental Social Psychology, 45(4), 901–907. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S002210310900 0948
- Neuroskeptic. (2015). How Diederik Stapel Became A Science Fraud Neuroskeptic. Retrieved from http://blogs.discovermagazine.com/neuroskeptic/2015/01/20/how-diederikstapel-became-fraud/
- Nie, N. H., Verba, S., & Petrocik, J. R. (1976). The Changing American Voter. Cambridge, Mass.: Harvard University Press.
- Nie, N. H., & Andersen, K. (1974). Mass Belief Systems Revisited: Political Change and Attitude Structure. The Journal of Politics, 36(03), 540. http://doi.org/10.2307/2129245
- Norman, D. A., & Shallice, T. (1986). Consciousness and Self-Regulation. In R. J. Davidson, G. E. Schwartz, & D. Shapiro (Eds.), Consciousness and Self-Regulation (pp. 1–18). Boston, MA: Springer US. http://doi.org/10.1007/978-1-4757-0629-1

- Olatunji, B. O., Adams, T., Ciesielski, B., David, B., Sarawgi, S., & Broman-Fulks, J. (2012, June). The Three Domains of Disgust Scale: Factor Structure, Psychometric Properties, and Conceptual Limitations. Assessment. http://doi.org/10.1177/1073191111432881
- Owe, E., Vignoles, V. L., Becker, M., Brown, R., Smith, P. B., Lee, S. W. S., ... Jalal, B. (2013). Contextualism as an Important Facet of Individualism-Collectivism Personhood Beliefs Across 37 National Groups. Journal of Cross-Cultural Psychology, 44(1), 24–45. http://doi.org/10.1177/0022022111430255
- Oxley, D. R., Smith, K. B., Alford, J. R., Hibbing, M. V., Miller, J. L., Scalora, M., ... Hibbing, J. R. (2008). Political Attitudes Vary with Physiological Traits. Science, 321(5896), 1667–1670. http://doi.org/10.1126/science.1157627
- Page, B. I., & Shapiro, R. Y. (1992). The rational public: Fifty years of trends in Americans' policy preferences. University of Chicago Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=3R9XFfX4bfcC&oi=f nd&pg=PR5&dq=Page+and+Shapiro+1992&ots=auGL6ABFvj&sig=t2ppyq2ory2W0V XwJr4jcMuruMI
- Peffley, M. A., & Hurwitz, J. (1985). A hierarchical model of attitude constraint. American Journal of Political Science, 871–890. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2111185
- Pessoa, L. (2008). On the relationship between emotion and cognition. Nature Reviews Neuroscience, 9(2), 148–158. Retrieved from http://www.nature.com.flagship.luc.edu/nrn/journal/v9/n2/abs/nrn2317.html
- Pieters, R., & Baumgartner, H. (2002). Who talks to whom? Intra-and interdisciplinary communication of economics journals. Journal of Economic Literature, 40(2), 483– 509. Retrieved from http://www.ingentaconnect.com.flagship.luc.edu/content/aea/jel/2002/0000040 /0000002/art00004
- Pollin, B. R., & Ash, M. (2013). Debt and Growth : A Response to Reinhart and Rogoff. New York Times, 2012–2014. Retrieved from http://www.nytimes.com/2013/04/30/opinion/debt-and-growth-a-response-toreinhart-and-rogoff.html?\_r=1
- Poole, K. T. (2005). Spatial models of parliamentary voting. Cambridge University Press. Retrieved from

http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=OmeQNHvcULoC& oi=fnd&pg=PR11&ots=1HeCVid9Ue&sig=Tw3ZkuDm4xJTr97GBstCX3-HBP8

- Poole, K. T., & Rosenthal, H. (2001). D-NOMINATE after 10 Tears. A Comparative Update to Congress: A Political-Economic History of Roll Call Voting. Legislative Studies Quarterly, 26, 5–29. Retrieved from http://www.citeulike.org/group/582/article/369671
- Popkin, S. L. (2006). The factual basis of "belief systems": A reassessment. Critical Review, 18(1-3), 233–254. http://doi.org/10.1080/08913810608443659
- Popkin, S. L. (2006). The Factual Basis of "belief Systems": A Reassessment. Critical Review, 18(1-3), 233–254. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217263425/abstra ct?accountid=12163
- Popp, E., & Rudolph, T. J. (2011). A Tale of Two Ideologies: Explaining Public Support for Economic Interventions, 73. http://doi.org/10.1017/S0022381611000478
- Quay, H. C. (1988). The behavioral reward and inhibition system in childhood behavior disorder.
- Redding, R. E. (2012). Likes Attract The Sociopolitical Groupthink of (Social) Psychologists. Perspectives on Psychological Science, 7(5), 512–515. http://doi.org/10.1177/1745691612455206
- RePass, D. E. (2008). Searching for Voters along the Liberal-Conservative Continuum: The Infrequent Ideologue and the Missing Middle (Vol. 6). Retrieved from http://www.degruyter.com.flagship.luc.edu/view/j/for.2008.6.2\_20120105083453 /for.2008.6.2/for.2008.6.2.1220/for.2008.6.2.1220.xml
- Revelle, W. (2016). psych: Procedures for Psychological, Psychometric, and Personality Research. Evanston, Illinois. Retrieved from http://cran.rproject.org/package=psych
- Ridderinkhof, K. R., Ullsperger, M., Crone, E. A., & Nieuwenhuis, S. (2004). The role of the medial frontal cortex in cognitive control. Science, 306(5695), 443–447. http://doi.org/10.1126/science.1100301
- Rigney, D., & Barnes, D. (1980). Patterns of Interdisciplinary Citation in the Social Sciences. Social Science Quarterly, 61(1), 114–27. Retrieved from

http://www.eric.ed.gov.flagship.luc.edu/ERICWebPortal/recordDetail?accno=EJ228 978

- Robson, A. J., & Kaplan, H. S. (2003). The evolution of human life expectancy and intelligence in hunter-gatherer economies. American Economic Review, 150–169.
   Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/3132166
- Roose, K. (2013). Meet the 28-Year-Old Grad Student Who Just Shook the Global Austerity Movement. New York Magazine. Retrieved from http://nymag.com/daily/intelligencer/2013/04/grad-student-who-shook-globalausterity-movement.html
- Rozin, P., & Royzman, E. B. (2001). Negativity Bias, Negativity Dominance, and Contagion. Personality and Social Psychology Review, 5(4), 296–320. http://doi.org/10.1207/S15327957PSPR0504\_2
- Schaich Borg, J., Lieberman, D., & Kiehl, K. A. (2008). Infection, Incest, and Iniquity: Investigating the Neural Correlates of Disgust and Morality. Journal of Cognitive Neuroscience, 20(9), 1529–1546. http://doi.org/10.1162/jocn.2008.20109
- Schlenker, B. R., Chambers, J. R., & Le, B. M. (2012). Conservatives are happier than liberals, but why? Political ideology, personality, and life satisfaction. Journal of Research in Personality, 46(2), 127–146. http://doi.org/10.1016/j.jrp.2011.12.009
- Schmidt, L. A., & Fox, N. A. (1994). Patterns of cortical electrophysiology and autonomic activity in adults' shyness and sociability. Biological Psychology, 38(2), 183–198. http://doi.org/10.1016/0301-0511(94)90038-8
- Schmidt, L. A., Fox, N. A., Schulkin, J., & Gold, P. W. (1999). Behavioral and psychophysiological correlates of self-presentation in temperamentally shy children. Developmental Psychobiology, 35(2), 119–135. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1002/(SICI)1098-2302(199909)35:2%3C119::AID-DEV5%3E3.0.CO;2-G/abstract
- Schmitt, N. (1996). Uses and abuses of coefficient alpha. Psychological Assessment, 8(4), 350–353. http://doi.org/10.1037/1040-3590.8.4.350
- Schreiber, D. (2007). Political cognition as social cognition: Are we all political sophisticates. The Affect Effect: Dynamics of Emotion in Political Thinking and Behavior, 48–70. Retrieved from http://fog.its.uiowa.edu/~c030111/decisionmaking/grad2005/schreiber2.pdf

- Schreiber, D. (2011). From SCAN to neuropolitics. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=2Nem4knFLk8C&oi =fnd&pg=PT418&dq=Schreiber+2005+politics&ots=04SKO4B3PC&sig=C65pmQPOZ k91if0BBD3W74eJxTg
- Schreiber, D. M. (2006). Evaluating politics | A search for the neural substrates of political thought. UNIVERSITY OF CALIFORNIA, LOS ANGELES.
- Schwartz, S. H. (1994). Are there universal aspects in the structure and contents of human values? Journal of Social Issues, 50(4), 19–45. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1540-4560.1994.tb01196.x/abstract
- Sears, D. O. (1986). College Sophomores in the Laboratory: Influences of a Narrow Data Base on Social Psychology's View of Human Nature. Journal of Personality and Social Psychology, 51(3), 515–530. http://doi.org/10.1037//0022-3514.51.3.515
- Sears, D. O., & Valentino, N. A. (1997). Politics Matters: Political Events as Catalysts for Preadult Socialization. The American Political Science Review, 91(1), 45–65. http://doi.org/10.2307/2952258
- Settle, J. E., Dawes, C. T., Christakis, N. A., & Fowler, J. H. (2010). Friendships moderate an association between a dopamine gene variant and political ideology. Journal of Politics, 72(4), 1189–1198. Retrieved from http://journals.cambridge.org.flagship.luc.edu/production/action/cjoGetFulltext?fu lltextid=7909322
- Settle, J. E., Dawes, C. T., & Fowler, J. H. (2009). The Heritability of Partisan Attachment. Political Research Quarterly, 62(3), 601–613. http://doi.org/10.1177/1065912908327607
- Sharot, T., Kanai, R., Marston, D., Korn, C. W., Rees, G., & Dolan, R. J. (2012). Selectively altering belief formation in the human brain. Proceedings of the National Academy of Sciences. http://doi.org/10.1073/pnas.1205828109
- Sherman, G. D., Haidt, J., & Clore, G. L. (2012). The Faintest Speck of Dirt Disgust Enhances the Detection of Impurity. Psychological Science. http://doi.org/10.1177/0956797612445318
- Shook, N. J., & Clay, R. (2011). Valence Asymmetry in Attitude Formation A Correlate of Political Ideology. Social Psychological and Personality Science, 2(6). http://doi.org/10.1177/1948550611405219

- Sibley, C. G., & Duckitt, J. (2010). The Personality Bases of Ideology: A One-Year Longitudinal Study. The Journal of Social Psychology, 150(5), 540–559. http://doi.org/10.1080/00224540903365364
- Sidanius, J., & Pratto, F. (1999). Social dominance: An intergroup theory of social hierarchy and oppression. Cambridge University Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=Gkiy2jAi-3wC&oi=fnd&pg=PR9&dq=Sidanius,+J.,+&ots=IdQL7cLVil&sig=Vby6UUrMsgPR86eY vEuQP3QmOw8
- Sigelman, L., & Goldfarb, R. (2008). The Diffusion of Concepts from Economics to Political Science: Disciplinary Dependency Meets Disciplinary Imperialism. Rochester, NY. Retrieved from http://papers.ssrn.com/abstract=1115973
- Sigelman, L., & Goldfarb, R. (2012). The influence of economics on political science: by what pathway? Journal of Economic Methodology, 19(1), 1–19. http://doi.org/10.1080/1350178X.2012.661067
- Singal, J. (2015). The Case of the Amazing Gay-Marriage Data: How a Graduate Student Reluctantly Uncovered a Huge Scientific Fraud. Retrieved from http://nymag.com/scienceofus/2015/05/how-a-grad-student-uncovered-a-hugefraud.html
- Singer, T., Seymour, B., O'Doherty, J., Dolan, R. J., Kaube, H., & Frith, C. D. (2004). Empathy for pain involves the affective but not sensory components of pain. Science (New York, N.Y.), 303(5661), 1157–62. http://doi.org/10.1126/science.1093535
- Skitka, L. J., & Mullen, E. (2002). Psychological determinants of public opinion. The Social Psychology of Politics, 107–134. Retrieved from http://www.springerlink.com.flagship.luc.edu/index/W876K5156Q875Q74.pdf
- Smith, K. B. (2006). Representational Altruism: The Wary Cooperator as Authoritative Decision Maker. American Journal of Political Science, 50(4), 1013–1022. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1540-5907.2006.00230.x/full
- Smith, K. B., Larimer, C. W., Littvay, L., & Hibbing, J. R. (2007). Evolutionary Theory and Political Leadership: Why Certain People Do Not Trust Decision Makers. Journal of Politics, 69(2), 285–299. http://doi.org/10.1111/j.1468-2508.2007.00532.x

- Smith, K. B., Oxley, D. R., Hibbing, M. V., Alford, J. R., & Hibbing, J. R. (2011). Linking genetics and political attitudes: Reconceptualizing political ideology. Political Psychology, 32(3), 369–397. http://doi.org/10.1111/j.1467-9221.2010.00821.x
- Smith, K., Alford, J. R., Hatemi, P. K., Eaves, L. J., Funk, C., & Hibbing, J. R. (2012). Biology, Ideology, and Epistemology: How Do We Know Political Attitudes Are Inherited and Why Should We Care? American Journal of Political Science. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1540-5907.2011.00560.x/full
- Socarides, R. (2013, February). Ken Mehlman's Gay-Marriage Mission. Retrieved from http://www.newyorker.com/online/blogs/newsdesk/2013/02/ken-mehlmansrepublican-gay-marriage-mission.html
- Somin, I. (2006). Knowledge About Ignorance: New Directions in the Study of Political Information. Critical Review, 18(1-3), 255–278. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217261219/abstra ct?accountid=12163
- Somit, A. (1972). Biopolitics. British Journal of Political Science, 2(2), 209–238. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/193357
- Somit, A., & Peterson, S. A. (1987). Political Socialization's Primacy Principle: A Biosocial Critique. International Political Science Review, 8(3), 205–213. http://doi.org/10.1177/019251218700800302
- Somit, A., & Peterson, S. A. (1998). Review article: biopolitics after three decades–a balance sheet. British Journal of Political Science, 28(03), 559–571. Retrieved from http://journals.cambridge.org.flagship.luc.edu/abstract\_S0007123498000246
- Somit, A., & Peterson, S. A. (1999). Rational choice and biopolitics: A (Darwinian) tale of two theories. PS: Political Science and Politics, 32(1), 39–44. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/420748
- Stewart, J. L., Coan, J. A., Towers, D. N., & Allen, J. J. B. (2011). Frontal EEG asymmetry during emotional challenge differentiates individuals with and without lifetime major depressive disorder. Journal of Affective Disorders, 129(1), 167–174. Retrieved from http://www.sciencedirect.com.flagship.luc.edu/science/article/pii/S016503271000 5653

- Stewart, J. L., Coan, J. A., Towers, D. N., & Allen, J. J. B. (2014). Resting and task-elicited prefrontal EEG alpha asymmetry in depression: Support for the capability model. Psychophysiology, 51(5), 446–455. http://doi.org/10.1111/psyp.12191
- Stimson, J. A. (2004). Tides of consent: How public opinion shapes American politics. Cambridge University Press. Retrieved from http://books.google.com.flagship.luc.edu/books?hl=en&lr=&id=gPkrGEw37tgC&oi= fnd&pg=PR21&dq=Stimson+2004+Tides+of+COnsent&ots=lzV2rsuV\_U&sig=GXYLZk ksNjTnBOmkUlkt-1lmwNU
- Stimson, J. A. (1975). Belief systems: Constraint, complexity, and the 1972 election. American Journal of Political Science, 393–417. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2110536
- Stokes, D. E. (1963). Spatial models of party competition. The American Political Science Review, 57(2), 368–377. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/1952828
- Stone, W. F., & Schaffner, P. E. (1988). The psychology of politics . Springer-Verlag Publishing.
- Sullivan, J. L., Piereson, J. E., & Marcus, G. E. (1978). Ideological Constraint in the Mass Public: A Methodological Critique and Some New Findings. American Journal of Political Science, 22(2), 233–249. http://doi.org/10.2307/2110615
- Sutton, S. K., & Davidson, R. J. (1997). Prefrontal brain asymmetry: A biological substrate of the behavioral approach and inhibition systems. Psychological Science, 8(3), 204–210. Retrieved from http://pss.sagepub.com.flagship.luc.edu/content/8/3/204.short
- Talhelm, T., Haidt, J., Oishi, S., Zhang, X., Miao, F., & Chen, S. (2012). Liberals Think More Analytically (More "Weird") than Conservatives. SSRN eLibrary. Retrieved from http://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2111700
- Tedin, K. L. (1987). Political ideology and the vote. Research in Micropolitics, 2(1), 63–94.
- Tetlock, P. E. (1983). Cognitive style and political ideology. Journal of Personality and Social Psychology, 45(1), 118–126. http://doi.org/10.1037/0022-3514.45.1.118

- Thórisdóttir, H., & Jost, J. T. (2011). Motivated Closed-Mindedness Mediates the Effect of Threat on Political Conservatism. Political Psychology, 32(5), 785–811. http://doi.org/10.1111/j.1467-9221.2011.00840.x
- Tomarken, A. J., Davidson, R. J., & Henriques, J. B. (1990). Resting frontal brain asymmetry predicts affective responses to films. Journal of Personality and Social Psychology; Journal of Personality and Social Psychology, 59(4), 791.
- Treier, S. (2010). Where does the president stand? Measuring presidential ideology. Political Analysis, 18(1), 124–136. Retrieved from http://pan.oxfordjournals.org.flagship.luc.edu/content/18/1/124.short
- Treier, S., & Hillygus, D. S. (2009). The Nature of Political Ideology in the Contemporary Electorate. Public Opinion Quarterly, 73(4), 679–703. http://doi.org/10.1093/poq/nfp067
- Treier, S., & Hillygus, S. (2005). The structure and meaning of political ideology. University of Georgia and Harvard University, 1–38. Retrieved from http://www.tc.umn.edu/~satreier/apsa05TreierHillygus.pdf
- Tritt, S. M., Inzlicht, M., & Peterson, J. B. (n.d.). Preliminary support for a generalized arousal model of political conservatism.
- Trivers, R. L. (1971). The evolution of reciprocal altruism. Quarterly Review of Biology, 35–57. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/2822435
- Tybur, J., Lieberman, D., & Griskevicius, V. (2009). Microbes, mating, and morality: individual differences in three functional domains of disgust. Journal of Personality and Social Psychology, 97(1), 103–122. http://doi.org/10.1037/a0015474
- Tybur, J. M., Bryan, A. D., Lieberman, D., Caldwell Hooper, A. E., & Merriman, L. A. (2011). Sex differences and sex similarities in disgust sensitivity. Personality and Individual Differences, 51(3), 343–348. http://doi.org/10.1016/j.paid.2011.04.003
- Van Veen, V., & Carter, C. S. (2002). The anterior cingulate as a conflict monitor: FMRI and ERP studies. Physiology and Behavior, 77(4-5), 477–482. http://doi.org/10.1016/S0031-9384(02)00930-7
- Verhulst, B., Eaves, L. J., & Hatemi, P. K. (2012). Correlation not Causation: The Relationship between Personality Traits and Political Ideologies. American Journal of Political Science, 56(1), 34–51. http://doi.org/10.1111/j.1540-5907.2011.00568.x

Wawro, G. J. (2006). The Rationalizing Public? Critical Review, 18(1-3), 279–296. Retrieved from http://search.proquest.com.flagship.luc.edu/abiglobal/docview/217273251/abstra ct?accountid=12163

- Weber, C., & Federico, C. M. (2007). Interpersonal attachment and patterns of ideological belief. Political Psychology, 28(4), 389–416. Retrieved from http://onlinelibrary.wiley.com.flagship.luc.edu/doi/10.1111/j.1467-9221.2007.00579.x/full
- Weissflog, M. J., Noordt, S. van, & Choma, B. L. (2010). Sociopolitical ideology and electrocortical responses. In Psychophysiology (Vol. 61, p. 47). Retrieved from https://www.brocku.ca/psychology/people/Weissflog SPR poster Sept23 2010.pdf
- Weissflog, M., Choma, B. L., Dywan, J., van Noordt, S. J. R., & Segalowitz, S. J. (2013). The political (and physiological) divide: Political orientation, performance monitoring, and the anterior cingulate response. Social Neuroscience, 8(5), 434–47. http://doi.org/10.1080/17470919.2013.833549
- Westen, D., Blagov, P. S., Harenski, K., Kilts, C., & Hamann, S. (2006). Neural bases of motivated reasoning: An fMRI study of emotional constraints on partisan political judgment in the 2004 US presidential election. Journal of Cognitive Neuroscience, 18(11), 1947–1958. Retrieved from http://www.mitpressjournals.org.flagship.luc.edu/doi/abs/10.1162/jocn.2006.18.1 1.1947
- Wheeler, R. E., Davidson, R. J., & Tomarken, A. J. (1993). Frontal brain asymmetry and emotional reactivity: a biological substrate of affective style. Psychophysiology, 30(1), 82–89. http://doi.org/10.1111/j.1469-8986.1993.tb03207.x
- Wilcox, F. O. (1942). The Monroe Doctrine and World War II. The American Political Science Review, 36(3), 433–453. Retrieved from http://www.jstor.org.flagship.luc.edu/stable/10.2307/1949622
- Wilson, E. O. (1979). Sociobiology: The new synthesis. ABC. Retrieved from http://www.tcrecord.org.flagship.luc.edu/library/Abstract.asp?ContentId=1254
- Wilson, E. O. (2000). Sociobiology: The New Synthesis, Twenty-Fifth Anniversary Edition (Twenty-Fif). Belknap Press of Harvard University Press. Retrieved from http://www.amazon.com/dp/0674002350

- Wilson, E. O. (2004). On Human Nature: Revised Edition (Revised Ed). Harvard University Press. Retrieved from http://www.amazon.com/dp/0674016386
- Wilson, G. D., & Patterson, J. R. (1968). A New Measure of Conservatism\*. British Journal of Social and Clinical Psychology, 7(4), 264–269. http://doi.org/10.1111/j.2044-8260.1968.tb00568.x
- Wingrove, J., & Bond, A. J. (1998). Angry reactions to failure on a cooperative computer game. Aggressive Behavior, 24(1), 27–36. Retrieved from http://ocw.metu.edu.tr/pluginfile.php/2358/mod\_resource/content/1/Optional\_A ngry reactions to failure on a cooperative computer game.pdf
- Xie, B., & Jaeger, P. T. (2007). Older Adults and Political Participation on the Internet: A Cross-cultural Comparison of the USA and China. Journal of Cross-Cultural Gerontology, 23(1), 1–15. http://doi.org/10.1007/s10823-007-9050-6
- Yeung, N., Botvinick, M., & Cohen, J. (2004). The Neural Basis of Error Detection: Conflict Monitoring and The Error Related Nagativity. Psychological Review, 11(4), 931–959. http://doi.org/10.1037/0033-295X.111.4.931
- Zajonc, R. (1980). Feeling and thinking: Preferences need no inferences. American Psychologist, 35(2), 151. Retrieved from http://psycnet.apa.org.flagship.luc.edu/journals/amp/35/2/151/
- Zajonc, R. (1984). On the primacy of affect. Retrieved from https://login.flagship.luc.edu/login?url=http://psycnet.apa.org/journals/amp/39/2/ 117/
- Zaller, J. R. (1992). The nature and origins of mass opinion. Cambridge university press.
- Zettler, I., & Hilbig, B. E. (2010). Attitudes of the selfless: Explaining political orientation with altruism. Personality and Individual Differences, 48(3), 338–342. http://doi.org/10.1016/j.paid.2009.11.002
- Yong, E. (2015, August 27). How Reliable Are Psychology Studies? Retrieved August 15, 2016, from http://www.theatlantic.com/science/archive/2015/08/psychology-studies-reliability-reproducability-nosek/402466/

## VITA

Dane Wendell, Ph.D. in Political Science from Loyola University Chicago, is currently a visiting professor at Rhodes College in Memphis, Tennessee. Dr. Wendell studies political cognition, neuropolitics, and the emotion of disgust. His dissertation is a study of emotional receptivity in the political brain using measures of electroencephalography (electrical signals read from the scalp). Dr. Wendell's recent projects include investigations of public opinion, evolutionary biology, and the role of disgust sensitivity for political attitude formation. He has taught courses on American politics, political behavior, evolutionary politics, and political psychology to undergraduates.